

Carbon Capture, Usage and Storage

An update on the business model for Transport and Storage explanatory note and indicative heads of terms

Disclaimer:

The details, as set out in this document, in whatever form they are expressed, are indicative only and do not constitute an offer by government and do not create a basis for any form of expectation or reliance. Parties are expected to get their own financial and legal advice. The business model referred to is not final and is subject to further development by the government, and approval by Ministers, in consultation with relevant regulators and the devolved administrations, as well as the development and Parliamentary approval of any necessary legislation, and completion of necessary contractual documentation. Government reserves the right to review and amend all provisions, for any reason and in particular to ensure that any proposals provide value for money (VfM) and are consistent with the current subsidy control regime.



© Crown copyright 2023

This publication is licensed under the terms of the Open Government Licence v3.0 except where otherwise stated. To view this licence, visit <u>nationalarchives.gov.uk/doc/open-government-licence/version/3</u> or write to the Information Policy Team, The National Archives, Kew, London TW9 4DU, or email: <u>psi@nationalarchives.gsi.gov.uk</u>.

Where we have identified any third-party copyright information you will need to obtain permission from the copyright holders concerned.

Any enquiries regarding this publication should be sent to us at: enquiries@beis.gov.uk

Contents

Update to the business model for Transport and Storage - indicative heads of terms and explanatory note_____

Overview	4
General	4
Key updates	4
Economic Regulatory Regime (ERR) Licence	5
Government Support Package	7
Revenue Support Agreement	10
Heads of terms	11
Annex A: Economic Regulatory Regime (ERR) Licence	11
Annex B: Government Support Package	140
Annex C: Revenue Support Agreement	230

4

Update to the business model for Transport and Storage - indicative heads of terms and explanatory note

This document, and Annexes A-C to it, set out our updated proposals for the business model for transport and storage (TRI model), including the Licence, Government Support Package and the Revenue Support Agreement. The latest heads of terms were published in June 2023¹.

Overview

General

The indicative heads of terms dated December 2023 (published as Annexes A-C to this document) set out the current proposals on the matters to be addressed as part of the CCUS TRI model.

They remain subject to ongoing development in conjunction with the development of all other relevant components of the CCUS programme, in particular the CCS Network Code.

Key updates

We set out below some key updates of note across the TRI model documentation, since the previous version of the indicative heads of terms were published in June 2023. This should be read in conjunction with Annexes A-C to this document, where the heads of terms are set out as follows:

- Annex A: Economic Regulatory Regime (ERR) Licence
- Annex B: Government Support Package
- Annex C: Revenue Support Agreement

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/116473 5/ccus-ts-business-model-update.pdf

Economic Regulatory Regime (ERR) Licence

There have been some key developments to the Licence since the previous publication in June 2023.

Key updates to the special conditions of the Licence include:

- refinement of the COD Readiness mechanism which links to T&SCo's right to receive Pre-operations Difference Payments under the RSA where there is an Event of First User Delay
- clarifications to the application of Delay WACC, in particular in circumstances where there is both a T&SCo delay and a delay not attributable to T&SCo
- introduction of a Force Majeure reopener to allow T&SCo to request relief
- introduction of a Relief Event mechanism to grant T&SCo relief by way of extension(s) of time to achieve the Scheduled Handover Date, Scheduled COD and/or the Longstop Date on specified grounds
- introduction of a mechanism for specified extension of time in conjunction with provision of a remedial plan, in circumstances where T&SCo may fail to achieve System Acceptance by the Longstop Date
- refinement of T&SCo's obligations to achieve specific levels of Availability (Availability Target and Availability Floor thresholds), and introduction of a relief mechanism that provides T&SCo relief from the consequences of failing to reach such thresholds on specific grounds
- refinement of the obligations on T&SCo to provide specified remediation plans where Availability falls below the required thresholds or corrective measures are required at a store, including cost recovery/sharing mechanisms where additional costs are required to remediate
- introduction of an obligation on T&SCo to procure independent audits of its storage capacity assumptions and/or potential new storage complexes

Updates to the standard conditions of the Licence include:

- clarifications around the interface between the Licence and the CCS Network Code
- updates with regards to the disposal of assets and restrictions on charges
- updates to provisions on financial ringfencing
- clarification and further parameters around a Qualifying Acquisition in the context of T&SCo's notification obligations
- the inclusion of a fit and proper person requirement
- the introduction of a mechanism for the future assessment of T&SCo for an investment grade issuer credit rating and, where appropriate, for T&SCo to obtain and maintain that credit rating

- the introduction of obligations on T&SCo to abide by the Regulator's data best practice guidance
- inclusion of a requirement on T&SCo to have two non-executive directors
- development of a framework for both the onshore and offshore decommissioning obligations on T&SCo

Other updates to the Licence to note are:

- clarification around the process to reach key milestones e.g. Handover and System Acceptance/Commercial Operations Date
- refinement of cost assessment process and calculations, noting that work is ongoing in this area
- updates to the process around reopeners and refinement of the definition of Change in Scope to explicitly refer to expansions and/or cancellations of part of the Approved T&S Network

Further progress has also been made to the Schedules to the Licence, in particular, the template for the Approved Project Development Plan (APDP) has been developed.

Government Support Package

1. Supplemental Compensation Agreement

The key update to the SCA heads of terms is the further development of the insurance schedule following input from technical and insurance advisers – in particular the inclusion of the definition of Specified Significant Irregularities. The SCA continues to develop and remains subject to further refinement by technical and insurance advisers.

The term of the SCA has been updated to reflect that following an SoS discontinuation under the Discontinuation Agreement, there may be a need for the SCA during the decommissioning and post closure monitoring period unless and until the SoS institute a statutory transfer scheme.

In addition, we have clarified the following with regards to the SCA:

- the Unavailability test has been updated to clarify that commercial insurance will be considered Unavailable if it is not available in respect of the Project and comparable Licensable Activities (as defined in the Energy Bill) in the UK
- the Excess Loss test has been updated to clarify where SoS will meet amounts in excess of the applicable policy limits under commercial insurances
- inclusion of an obligation on T&SCo to procure that its commercial insurances in respect of the Insured Risks and Insured Losses contain a loss payee provision
- refinement of the process around Renewals, Market Testing and assurance
- provision that T&SCo may request a variation or amendment to the Insurance Schedule, which is subject to SoS consent.

It is also worth noting that work continues with regards to the methodology for calculation of the SCA Fee. Key principles are noted in the SCA heads of terms.

A lead insurer agreement has been included to address the interface between government and commercial insurers. The approach to these arrangements remain subject to development and insurance market feedback.

2. Discontinuation Agreement

There have been a number of updates to the Discontinuation Agreement around the triggers and process for discontinuation as follows:

- the process that SoS will follow in issuing a Discontinuation Notice and the updates that it will provide to T&SCo
- the thresholds for the SCA Discontinuation Threshold and the RSA Discontinuation Threshold have also been included, along with further detail on how the triggers will operate and increase where SoS elects not to discontinue despite the trigger being reached
- a second discontinuation right has also been included for where the SCA is continuing following an initial discontinuation.

It has been clarified that the Discontinuation Agreement will expire on the later to occur of the expiry of the SCA and the expiry of the RSA.

The key update to the Discontinuation Agreement is the inclusion of the Agreed Financing Principles and Approved Hedging Policy which is designed to determine DA Approved Hedging for the purposes of calculation of the Total Compensation Amount. The details of the Total Compensation Amount for where SoS elects to discontinue have been updated to align with these principles. Within the Total Compensation Amount, the position on decommissioning shortfall has also been clarified to provide that SoS will compensate for decommissioning shortfall where the RSA Discontinuation Threshold or the Unavailability test has been triggered.

Other updates to the Discontinuation Agreement include:

- the inclusion of a tax gross-up provision
- updates to the SoS right of set-off against equity compensation
- clarifications to the definition of Wilful Misconduct and Gross Negligence, which can result in the reduction or forfeiture of equity compensation
- updates to the timing and mechanics of the Total Compensation Amount
- inclusion of further detail on the Discontinuation Plan, including a template plan

3. Liaison Agreement

It has been clarified that the Liaison Agreement will continue until the expiry of the Discontinuation Agreement – it is intended to continue until the GSP is no longer in effect.

A key update to the Liaison Agreement is to the Variation mechanism. This gives SoS a consent right for Change in Scope that meet specified thresholds e.g. where there is an extension to the asset life of the Approved T&S Network that results in an extension of the period of application of Allowed Revenue.

Another key update is to the process around a Transfer Scheme instituted by SoS under the prospective Energy Bill (which remains subject to Royal Assent). Further detail is included around the basis of valuation for the Transfer Scheme, depending on whether T&SCo undertaking is transferred as a going concern or not.

Updates have also been made to:

- clarification and further parameters around a Qualifying Acquisition in the context of T&SCo's notification obligations
- the formative list of Conditions Precedent
- the process around the Remedy Event and Failure Event regime

Revenue Support Agreement

It is now indicated that the RSA will continue until expiry of the first to occur of a specified date, Discontinuation or revocation of the Licence. The intention is for the specified date to be defined by reference to the asset life of the Approved T&S Network (as at Licence Award) currently assumed to be no longer than 25 years from the Commercial Operations Date, noting HMG reserves its position in relation to RSA coverage of any future expansion.

The key updates to the RSA are to the process around the provision of evidence and calculation of the Difference Payments and reconciliation. RSA payments will be made monthly to T&SCo with a quarterly reconciliation and a reconciliation on expiry.

The support that will be provided prior to operations has also been clarified in the event that there is an Event of First User Delay.

Heads of terms

Annex A: Economic Regulatory Regime (ERR) Licence

CCUS: T&SCo Licence

Indicative Heads of Terms

Note: these indicative heads of terms are subject to the "Disclaimer" section at the front of the explanatory note document to which they are annexed.

INTRODUCTION

1.1 Background

- (a) The indicative heads of terms in Parts I to III (*Licence Heads of Terms*) of this document set out the basis for the provision of a licence (the "Licence"), expected to be granted by the Secretary of State or the economic regulator (the "Regulator") to a company licensed to provide transport and storage services ("T&SCo") under HM Government's ("HMG") CCUS programme (the "Programme").
- (b) The contents of the heads of terms in Parts I to III (*Licence Heads of Terms*) of this document are indicative only and do not constitute an offer by HMG or the Regulator and do not create a basis for any form of expectation or reliance. Any economic regulatory regime ("ERR") that is developed in the future will be subject to Parliamentary approval and any Licence that is developed in the future will be subject to approval by the Regulator and the development and completion of necessary documentation.
- (c) This introduction does not form part of the indicative heads of terms in Parts I to III (*Licence Heads of Terms*) of this document and is intended only to provide an overview of the rationale and assumptions for the terms and conditions of the Licence.

1.2 Rationale

- (a) An ERR is required to provide a framework for users of the transport and storage ("T&S") network to pay T&SCo fees which will be regulated by the Regulator, and to set an allowed revenue for T&SCo and minimum standard of service to be provided by T&SCo for the T&S users.
- (b) Under the ERR, there will be mechanisms for the allocation of certain risks between the T&S users and incentives on T&SCo to provide the T&S services.

1.3 **Regulatory periods**

- (a) The ERR will cover a number of regulatory periods over the term of the Licence. The first regulatory period will commence on the award of the Licence and conclude on the first 31 March to occur on or following the third anniversary of the Commercial Operations Date (the "First Regulatory Period"). This means that the First Regulatory Period will cover construction period activities, commissioning period activities as well as the initial operating period. The second regulatory period will commence after the conclusion of the First Regulatory Period and there will be subsequent regulatory periods at agreed intervals.
- (b) HMG will determine the ERR for the First Regulatory Period, with the Regulator having a more limited role than it will in the second and subsequent regulatory periods. The

Regulator will, however still have certain responsibilities during this First Regulatory Period and these are set out in more detail below.

1.4 **Regulator's duties and enforcement powers**

- (a) The Regulator's objectives for carrying out its functions will be set out in legislation. The relevant enabling legislation is currently progressing through Parliament¹. The final wording of the Regulator's objectives will be subject to the outcome of the Parliamentary approval process.
- (b) The Licence will be supplemented by guidance from the Regulator regarding key areas under the Licence where the Regulator is required to exercise a degree of discretion, including, in particular, the determination of the WACC in the second and subsequent regulatory periods. While it is intended that the guidance can be prepared in draft in advance of the award of the Licence, the guidance can only be issued in final form once the Regulator has been given its necessary powers, duties and functions under the relevant legislation and the Licence is in place.
- (c) The statutory framework, which will provide for the grant of the Licence, will give the Regulator the powers required to enforce the conditions of the Licence, following the approach in other regulated sectors.

1.5 Description

- (a) This document sets out high level indicative terms for each of the following areas covered by the terms and conditions of the Licence:
 - (i) introduction;
 - (ii) heads of terms for the licence:
 - (A) Part I: the Terms of the Licence;
 - (B) Part II: the Standard Conditions;
 - (C) Part III: the Special Conditions. These Special Conditions will, primarily, deal with the ERR – that is, the building blocks and mechanics for determining the Allowed Revenue of T&SCo. These will be amended for each individual T&SCo to incorporate the variables relevant to each individual T&SCo;
 - (D) Schedule 1: Specified Area;
 - (E) Schedule 2: Revocation;
 - (F) Schedule 3: Financial Settlement Schedule;
 - (G) Schedule 4: Technical Details Schedule;
 - (H) Schedule 5: Derogations;
 - (I) Schedule 6: Template Form of Approved Project Development Plan; and

See the Energy Bill here: https://bills.parliament.uk/bills/3311

- (J) Schedule 7: Qualifying Acquisition Information.
- (b) The scope of the conditions set out in the indicative heads of terms in Parts I to III (*Licence Heads of Terms*) of this document remains subject to review, in particular in the context of the development of the wider Programme.

1.6 Initial assumptions

- (a) This document is based on the following assumptions in respect of the T&S Network:
 - (i) ownership of the T&S Network: the "onshore" (i.e. Onshore Transportation Systems) and "offshore" (i.e. Offshore Transportation and Storage Systems) elements of the T&S Network will be owned by T&SCo. In particular, T&SCo will hold any Crown Estate (including Crown Estate Scotland) lease and licence/permit issued by the NSTA (or licensing authority of a devolved administration);
 - development and operation of the T&S Network: T&SCo will be responsible for the development, construction, operation and maintenance of the T&S Network, including obtaining all necessary permits and approvals;
 - expansion of the T&S Network: in the future T&SCo may need to expand the T&S Network under the terms of the ERR and will provide access to new T&S users in accordance with the terms of its licence, the CCS Network Code and all relevant laws and regulations;
 - (iv) decommissioning: T&SCo will be responsible for the decommissioning and monitoring of:
 - (A) the Offshore Transportation and Storage System in accordance with the Petroleum Act 1998 as applied under the Energy Act 2008 (as may be amended or supplemented), and with regards to post-closure and monitoring obligations, T&SCo will be responsible as set out in its storage permit and licence granted by the NSTA; and
 - (B) the Onshore Transportation System in accordance with the relevant decommissioning conditions of any development/planning consent;
 - T&S Charges: users of the T&S Network will pay T&SCo Charges for provision of T&S services which will be regulated under the ERR;
 - (vi) CO₂ ownership: title and risk in CO₂ delivered to the T&S Network will pass to T&SCo at the Delivery Point;
 - (vii) government support package ("GSP"): T&SCo will benefit from a GSP provided by HMG to cover certain high impact, but low probability, risks beyond those which are manageable by operation of the ERR and the RSA, which the investors and/or supply chain, including insurers, of T&SCo cannot take, or cannot price at an efficient level which is good value for money for UK taxpayers, consumers or Users; and
 - (viii) revenue support agreement ("RSA"): in the development of the revenue model for the T&S Network, HMG has identified that there may be certain demand-related revenue risks to T&SCo, where T&SCo's actual revenues may fall short of T&SCo's Allowed Revenue under the ERR. Such demand-related revenue risks will be addressed through a number of mitigation measures in the ERR and where the ERR is not sufficient, T&SCo will have access to the

RSA as a mechanism which will enable the recovery by T&SCo of its Allowed Revenue in the event of a shortfall.

1.7 **Defined terms**

Capitalised terms not otherwise defined in this introduction shall have the meaning given to them in Parts II and III (*Licence Heads of Terms*).

LICENCE HEADS OF TERMS

PART I: TERMS OF THE LICENCE

- This licence, granted under section [7] of [*the Energy Bill*] (the "Act"), authorises [*the relevant T&SCo*], a company registered in England and Wales under company registration number [●] (the "Licensee") whose registered office is situated at [●], to participate in carrying out the activities set out in section 2(2) of the Act in the area specified in Schedule 1, during the period specified in paragraph 3 below, subject to:
 - (a) the standard conditions set out in Part II;
 - (b) the special conditions set out in Part III, (together the "conditions"); and
 - (c) such Schedules to the licence, if any, as may be referenced in the conditions or the terms of the licence.
- 2. This licence is subject to transfer, modification or amendment in accordance with the provisions of the Act, or the conditions.
- 3. This licence shall come into force on [●] and unless revoked in accordance with Schedule 2 or any conditions providing for revocation, shall continue until determined by not less than 25 years' notice in writing given by the economic regulator (the **"Regulator"**) to the Licensee where such 25 years' notice cannot be served before the date on which the Licensee commences commercial operation of the T&S network.

Note: the duration of the ERR will cover the construction, operational, closure, and postclosure (including decommissioning) phases of the T&S network. The licence shall come into force on the date of award of the licence and, unless revoked in accordance with certain limited revocation events, shall continue until terminated by a specified period of notice given by the Regulator to the Licensee.

4. Without prejudice to sections 11 and 23(1) of the Interpretation Act 1978, Parts I to III of this licence and its Schedules shall be interpreted and construed in the same manner as an Act of Parliament passed after the commencement of the Interpretation Act 1978.

Note: other more mechanical and interpretive provisions, such as provisions dealing with service of documents, may also be included here in due course.

- 5. References in this licence to a provision of any enactment, where after the date of this licence:
 - (a) the enactment has been replaced or supplemented by another enactment; and
 - (b) such enactment incorporates a corresponding provision in relation to fundamentally the same subject matter,

shall be construed, so far as the context permits, as including a reference to the corresponding provision of that other enactment.

PART II: STANDARD CONDITIONS

Section A: Interpretation, Application and Payments

1.	Definitions and interpretation	In the standard conditions unless the context otherwise requires:
		"Act" means [the Energy Bill 2022-23];
		"Affiliate" means, in relation to the Licensee, any Parent Undertaking of the Licensee, any Subsidiary Undertaking of the Licensee, or any Subsidiary Undertaking of a Parent Undertaking of the Licensee;
		"Allowed Revenue" means the allowed revenue calculated, from time to time, in accordance with this licence;
		"Appropriate Proportion" means the proportion of the Regulator Costs attributable to the Licensee in accordance with the principles determined by the Regulator for the purposes of this condition generally (after consulting the Licensee and others likely to be affected by the application of those principles) and notified to the Licensee;
		"Approved Offshore Decommissioning Programme" means a decommissioning programme prepared in relation to the Offshore Transportation and Storage System under the Petroleum Act 1998 and the Energy Act 2008 (in each case as amended or supplemented) in the form approved by the Secretary of State, as may be amended from time to time with the Secretary of State's approval;
		"Approved Project Development Plan" or "APDP" means the project development plan as approved in the financial settlement and included in Schedule 6 to the licence;
		"Approved T&S Network" means the parts of the T&S Network comprised in Phase 1 and Phase 2, as described in [sections 3.1 and 3.2] of the APDP, as updated pursuant to any Change in Scope;
		"Associate" means:
		(a) an Affiliate or Related Undertaking of the Licensee;
		(b) an Ultimate Controller of the Licensee;
		(c) a Participating Owner of the Licensee; or
		(d) a Common Control Company;

"Business Day" means a day (other than a Saturday or Sunday) on which banks are open for domestic business in the City of London;
" Capex " means post-Licence Award capital expenditure costs incurred in respect of the T&S Network, which costs may relate to construction and/or commissioning-related activities;
"Carbon Dioxide Storage Regulations" means regulations in respect of the storage of carbon dioxide under the Energy Act 2008;
"CCS Licensee" means any holder of a licence issued under [section 7 of the Act];
"CCS Network Code" has the meaning given to it in condition B6 (CCS Network Code);
"Change in Scope" means following Licence Award a change to the scope of the Project set out in the APDP as a result of:
 (a) the conclusion of On-going Devex in respect of works referenced in the APDP, such that settlement can be reached on expansion or enhancement of the Approved T&S Network in accordance with the APDP;
 (b) the introduction of a proposed new expansion for prospective new Users and/or expansion of an existing Storage Complex or an addition of a new Storage Complex, in each case (i) as envisaged in the APDP; or (ii) which requires an allocation of new On-going Devex; or
 (c) a cancellation of part of the Approved T&S Network, including a cancellation of a Phase 1 activity or Phase 2 activity and/or a Phase 3 activity;
"Charging Year" means a period of 12 months commencing on 1 April and ending on 31 March, save that:
(a) the first Charging Year will commence [on Licence Award] and end on the immediately following 31 March; and
 (b) the final Charging Year will commence on 1 April immediately preceding the last day of the term of this licence and end on that day;
"Code Agreement" means the agreement signed or acceded to by the Licensee, CCS Licensees and Users to make the CCS Network Code binding on them;

	"Commercial Operations Date" or "COD" means the date when System Acceptance is achieved, as confirmed by the Regulator under condition G3.4(a);
	"Commissioning Activities" has the meaning given to it in the APDP;
	"Commissioning Period" has the meaning given to it in condition G1.1;
	"Common Control Company" means any company, any of whose Ultimate Controllers (applying the definition set out in this condition A1 but substituting that company for the Licensee) is also an Ultimate Controller of the Licensee;
	"Construction Period" has the meaning given to it in condition F1.1;
	"CPIH" means Consumer Prices Index, including owner occupiers' housing costs (series L522) published by the ONS;
	" CPIH Indexed " means that the relevant amount is subject to indexation by CPIH at the start of each Charging Year after Licence Award;
	"Data Best Practice Guidance" means the guidance document issued by the Regulator in accordance with conditions B23.4 to B23.8;
	"Delivery Point" means a point at the boundary between a User's facilities and the T&S Network, more particularly described in the User's connection agreement, at which a User transfers title in carbon dioxide to the Licensee;
	"Disposal" means:
	(a) in relation to disposal of a T&S Asset situated in England and Wales, a transaction that includes any sale, gift, exchange, assignment, lease, licence, loan, mortgage, charge, or grant of any other encumbrance or the permitting of any encumbrance to subsist or any other disposition;
	(b) in relation to disposal of a T&S Asset situated in Scotland, a transaction that includes the grant of any disposition, conveyance, contract of excambion, any lease, assignation, licence, the grant of any right of possession, loan, standard security, floating charge to a third party, or the grant of any servitude right, wayleave or any other transaction or event which is capable under any enactment or rule

of low of offective the title to a verification
of law of affecting the title to a registered interest in land;
"Energy System Data" has the meaning given to that term in the Data Best Practice Guidance;
Note : Ofgem's Data Best Practice Guidance currently refers to "Energy System Data". It is proposed that the guidance will be updated to make it clear that data relating to the T&S Network is included within the scope of Energy System Data.
"First Regulatory Period" means the period from Licence Award until (and including) the first 31 March to occur on or following the third anniversary of the Commercial Operations Date;
"Good Industry Practice" means in relation to any undertaking and any circumstances, the exercise of that degree of skill, care, diligence, prudence, foresight and operating practice which would reasonably and ordinarily be expected from a skilled and experienced operator engaged in the same or similar type of undertaking as that of the Licensee under the same or similar circumstances;
"Government Support Package" or "GSP" means the government support package;
"Handover Works" has the meaning given to it in the APDP;
"Investment Grade Issuer Credit Rating" means an investment grade issuer credit rating of the Licensee from S&P, Moody's, Fitch or DBRS Morningstar, requested by the Regulator in accordance with condition B17.3;
"Investment Grade Issuer Credit Rating Guidance" means the guidance document issued by the Regulator as referred to in condition B17.4;
"Legal Requirement" means:
(a) any licences, permits, approvals and consents issued by any body of competent jurisdiction (including, for the avoidance of doubt, the consents of any devolved administration) in relation to the construction, commissioning, operation or decommissioning (and post- decommissioning monitoring) of the T&S Network; and
 (b) any act of Parliament, subordinate legislation, ordinance, code, decision, directive, direction, requirement, order, decree, regulation, determination or published procedure, policy or

		guidance of any body of competent jurisdiction (including courts and tribunals) which is legally binding on the Licensee;
		nce Award" means the date on which this e comes into force;
	"Lice licence	nsee" is the licensee named in the terms of this e;
	"Loca	Il Content" means:
	(a)	in respect of services, those services provided by a company carrying on business in the UK; and,
	(b)	in respect of goods, those goods which are being made, changed or improved in the UK (using the same definition as goods eligible for a UK country of origin certificate);
		t Restricted Communal Element" has the ing given by reference to the following concepts:
	(a)	"Most Restricted", which means the part of the T&S Network (or part of the T&S Network) through which the lowest mass of carbon dioxide can instantaneously flow, determined by the lower of any physical, technical and/or regulatory limits; and
	(b)	"Communal Element", which refers to any part of the T&S Network (or part of the T&S Network) that all Users' carbon dioxide must flow through, excluding any contingent infrastructure;
	the p onsho	the MRCE definition is under review to reflect parallel nature of aspects of sections of the re and offshore networks, e.g multiple stores or ole gathering pipelines.
	agree gas in name,	h Sea Transition Deal " means the initiative d between the UK government and the oil and dustry, as documented in the document of that , dated March 2021, as may be updated or emented;
		A " means the North Sea Transition Authority, is the trading name of the Oil and Gas Authority;
	capac Comm	gated Network Capacity " means the total ity available, determined by the Most Restricted nunal Element of the T&S Network measured in conne (Mt) of CO ₂ per day;

	"Offshore Decommissioning Authority" means the Secretary of State, represented by the Offshore Petroleum Regulator for Environment and Decommissioning;
	"Offshore Decommissioning Fund" means a fund or multiple funds established under regulations made pursuant to Chapter 2 of Part 2 of the Act;
	"Offshore Decommissioning Fund Account" means the account held with $[\bullet]$ in the Licensee's name, with account number $[\bullet]$, for the purpose of holding the amounts standing to the credit of the Offshore Decommissioning Fund from time to time;
	"Offshore Decommissioning Fund Accrual Profile" means a straight line accrual profile;
	"Offshore Decommissioning Fund Contribution Amount" means the monthly contribution amount the Licensee is obliged to pay into the Offshore Decommissioning Fund Account in accordance with condition D3.7;
	"Offshore Decommissioning Fund Investment Strategy" has the meaning given to it in condition D3.3(b);
	"Offshore Decommissioning Requirements" means the Legal Requirements relating to:
	 (a) the decommissioning of offshore carbon capture and storage installations, pipelines and wells;
	(b) post-decommissioning monitoring;
	 (c) obligations to pay a contribution for the post- licence termination monitoring and other costs of the government; and
	(d) obligations to pay post-closure financial security costs,
	as set out in the Petroleum Act 1998, the Energy Act 2008, the Act, regulations made under the Act, the Storage of Carbon Dioxide (Termination of Licences) Regulations 2011, the Storage of Carbon Dioxide (Licensing etc) Regulations 2010, the Storage Licence, the Storage Permit and any guidance issued in relation to these statutory provisions;
	Note : definition of Offshore Decommissioning Requirements may be further developed.
	"Offshore Pipeline Infrastructure" means that part of the T&S Network which is offshore, including

the pipelines and related infrastructure from the pig trap at the Onshore Transportation System up to but not including the pig trap at the inlet to the Storage Complex (on the basis that such pig trap is part of the Storage Complex), but excluding the Storage Complex and including the Terminal;
"Offshore Transportation and Storage System" means the Offshore Pipeline Infrastructure and any Storage Complex;
"On-going Capex" means Capex incurred in respect of Phase 2 which is not SRAV Capex;
" On-going Devex " means the development costs for front end engineering design (FEED) and other pre- FEED development spend which will be incurred in respect of the continued development and expansion of the T&S Network as set out in the APDP, which are incurred (i) post-Licence Award; and (ii) prior to the date of the final investment decision of the relevant expansion project;
"Onshore Decommissioning Authority" means:
(a) the local planning authority; or
(b) multiple local planning authorities, where the Onshore Transportation System is located in a geographical area that straddles more than one local planning authority area,
in charge of enforcing the Onshore Decommissioning Requirements that apply to the Onshore Transportation System;
"Onshore Decommissioning Fund" means a fund or multiple funds established for providing security for the discharge of liabilities in respect of decommissioning costs (and costs of related activities) in relation to the Onshore Transportation System, established in accordance with condition C3;
"Onshore Decommissioning Fund Account" means the account held with $[\bullet]$ in the Licensee's name, with account number $[\bullet]$, for the purpose of holding the amounts standing to the credit of the Onshore Decommissioning Fund from time to time;
"Onshore Decommissioning Fund Accrual Profile" means a straight line accrual profile;
"Onshore Decommissioning Fund Contribution Amount" means the monthly contribution amount the Licensee is obliged to pay into the Onshore Decommissioning Fund Account in accordance with condition C3.8;

"Onshore Decommissioning Fund Investment Strategy" has the meaning given to it in condition C3.3(b);
"Onshore Decommissioning Plan" has the meaning given to it in condition C3.3;
"Onshore Decommissioning Requirements" means the Legal Requirements that require onshore infrastructure to be decommissioned (and any activities related to decommissioning to be carried out), including but not limited to any requirements imposed or administered by the Onshore Decommissioning Authority;
"Onshore Transportation System" means that part of the T&S Network which is located onshore, including the pipelines and related infrastructure from the User(s) boundary fence to the pipeline entry pig trap at the entry to the Terminal, but excluding any part of the T&S Network which constitutes the Offshore Transportation and Storage System;
Note: the definitions of Onshore Transportation System, Offshore Transportation and Storage System, Offshore Pipeline Infrastructure and Terminal may require further adjustment.
"Opex" means the operating, maintenance and management costs associated with the operation of the T&S Network;
" Parent Undertaking " means a parent undertaking within the meaning of section 1162 of the Companies Act 2006;
"Participating Interest" has the meaning given in section 421A of the Financial Services and Markets Act 2000;
"Participating Owner" for the purposes of the definition of Associate, a person is subject to a Participating Interest by another person (a "Participating Owner") if:
(a) that other person holds a Participating Interest in the person; or
 (b) the person is subject to a Participating Interest by a person who is themselves subject to a Participating Interest by that other person;
"Periodic Review" means a periodic price control;

"Permitted Purpose" means the purpose of all or any of the following:
 (a) the T&S Business or any other business or activity within the limits of conditions B11.4 and B11.6 in condition B11 (<i>Restriction on</i> <i>Activity and Financial Ring-fencing</i>);
 (b) without prejudice to the generality of paragraph (a), any payment or transaction lawfully made or undertaken by the Licensee for a purpose within purpose within paragraphs (b)(i) to (vii) of condition B19.1 (<i>Indebtedness</i>);
"Phase 1" means the Handover Works and Commissioning Activities to be carried out in the Construction Period and Commissioning Period respectively, with associated SRAV Capex relating to the Approved T&S Network included in the APDP, as further described in [section 3.1] of the APDP;
"Phase 2" means post-Commercial Operations Date activities with associated On-going Capex relating to the Approved T&S Network included in the APDP, as further described in [section 3.2] of the APDP;
" Project " means the financing, design, construction, commissioning, acceptance, operation, maintenance, decommissioning, post-closure monitoring and ownership of the T&S Network;
"RAV" means the regulatory asset value;
" Receivable " means a contractual right to receive any sum or sums or any other financial asset from another person;
" Regulator " means the Gas and Electricity Markets Authority in its role as the economic regulator under Part 1 of the Act;
" Regulator Costs " has the meaning given to it in condition A2.3;
"Related Undertaking" in relation to the Licensee, means any undertaking in which the Licensee has a participating interest within the meaning of section 421A of the Financial Services and Markets Act 2000;
"Relevant Objectives" means:
 (a) the safe, economic, efficient and effective operation of the T&S Network to which the licence relates (including the coordinated, safe, efficient and economic operation of the T&S

networks of one or more other CCS Licensees where relevant); and
(b) the economic, efficient and effective discharge of the Licensee's obligations under the licence;
"Relinquishment of Operational Control" includes entering into any agreement or arrangement under which decisions relating to the operational management and control of the asset are not (or cease to be) at the sole direction of the licensee, and "relinquish" and any related expressions in this context are to be read accordingly;
"RES/RAS Process Guidance" means the guidance document issued by the Regulator as referred to in condition B17.4;
"RES/RAS Process" means either a "Rating Evaluation Service" conducted by S&P, a "Rating Assessment Service" conducted by Moody's or Fitch, or an equivalent credit rating assessment conducted by DBRS Morningstar to give an assessment of the potential credit rating of the Licensee;
"Revenue Calculations" means the applicable values and calculations of the SRAV, RAV and/or Allowed Revenue (as applicable) as set out or referred to in these special conditions;
"Secretary of State" or "SoS" means the Secretary of State for the Department for Energy Security and Net Zero;
"Significant Code Review" means a review of one or more matters which:
(a) relate to the CCS Network Code; and
(b) the Regulator considers likely to be of particular significance in relation to the Regulator's principal objectives and/or general duties (under section 1 of the Act),
concerning which the Regulator has issued a notice to the CCS Network Code parties stating that the review will constitute a Significant Code Review, the start date of the Significant Code Review, and the matters that will fall within the scope of the review;
"Significant Managerial Responsibility or Influence" means where a person plays a role in:
 (a) the making of decisions about how the whole or a substantial part of a Licensee's activities are to be managed or organised, or

 (b) the actual managing or organising of the whole or a substantial part of those activities;
"SRAV" means a shadow regulatory asset value, including the shadow regulatory asset value referenced in condition F4.1 and any additional shadow regulatory asset value set up pursuant to condition H7 (<i>Transferred SRAV and Additional SRAV(s)</i>);
"SRAV Capex" means Capex which is attributable to an SRAV;
Note: this includes Capex in respect of the Handover Works and Commissioning Activities and Capex assessed as part of a Change in Scope for inclusion on an additional SRAV.
"Storage Complex" has the meaning given to it in the Carbon Dioxide Storage Regulations;
Note: checks being run on definition in context of use.
"Storage Licence" means the licence held by the Licensee in relation to the Storage Complex under the Carbon Dioxide Storage Regulations;
"Storage Permit" means the storage permit held by the Licensee in relation to the Storage Complex in accordance with the Energy Act 2008;
"Subsidiary Undertaking" means a subsidiary undertaking within the meaning of section 1162 of the Companies Act 2006;
"Supplemental Compensation Agreement" means the supplemental compensation agreement;
"System Acceptance" means confirmation by the Regulator that the Commissioning Activities have been completed;
"System Operation" means those activities and functions of the Licensee which relate to the operation of the T&S Network, being the allocation of capacity to Users, the management and system planning of that capacity, and constraint management;
"T&S Assets" means all the infrastructure and equipment that makes up the T&S Network;
"T&S Business" means the development and operation of a network for the transportation and storage of carbon dioxide, including compliance with the Licensee's obligations under the licence;

-

"T&S Network" means the Licensee's Transportation System and Offshore Trans and Storage System, and includes any part of	sportation
"T&SCo of Last Resort Direction" direction issued by the Regulator pursuant to B21.2;	
"Terminal" means the onshore terminal fa all ancillaries between the pig trap at the Transportation System and the pipeline pig to Offshore Pipeline Infrastructure, which is do form part of the Offshore Transportation and System and not form part of the Transportation System;	Onshore rap at the eemed to d Storage
"Transfer Scheme" means a transfer instituted by the Secretary of State in response T&S Network under section [50] of the Act;	
"UK ETS" means the UK Emissions Trading established under the Greenhouse Gas Trading Scheme Order 2020 (as ame supplemented by other legislation) and any issued in relation to it;	Emissions ended or
"Ultimate Controller" means:	
 (a) a Parent Undertaking of the Licensee not itself a Subsidiary Undertaking o company; and 	
(b) any person who (whether alone of person or persons connected with the position to control, or to exercise of influence over, the policy of the License Parent Undertaking of the Licensee by	m) is in a significant see or any
(i) rights under contractual arran to which they are a party or they are a beneficiary; or	-
(ii) rights of ownership (includin attached to or deriving from sec rights under a trust) which ar them or of which they are a be	curities or e held by
but excluding any director or emplo corporate body in their capacity as su	
for the purposes of sub-paragraph (b), a connected with another person if they are pa arrangement regarding the exercise of any su as are described in that paragraph;	rty to any
"Urgent Modification Proposal" means a for a modification to the CCS Network Code	

		deals w	or agrees or considers is urgent because it it an urgent issue relating to the safe and/or operation of the T&S Network;
		other ho	means a person other than the Licensee or any older of a licence issued under [section 7 of the no is for the time being bound by the CCS < Code;
		of the T connect Network	Arrangements'' means arrangements in the Licensee and Users/ future Users for use T&S Network, including arrangements for the ion of Users' / future Users' facilities to the T&S k, and the delivery of carbon dioxide by Users ivery Point; and
		"WACC	" means weighted average cost of capital.
		dioxide primaril to carbo	the context otherwise requires, "carbon e" (or " CO_2 ") means a stream consisting y of carbon dioxide molecules and references on dioxide being delivered by a User to the T&S k assume that the stream being delivered is dioxide.
		approva given oi	rection, consent, derogation, determination, al, designation, decision, or other instrument r made by the Regulator under this licence will n or made in writing.
			the general interpretation provisions will be ed in the full-form licence conditions.
2.	Payments by the Licensee to the Regulator	2.1	The Licensee shall pay to the Regulator such amounts as are determined by or under this condition.
		2.2	In respect of each Charging Year at the beginning of which the Licensee holds this licence, the Licensee shall pay to the Regulator an amount which is the Appropriate Proportion of the Regulator Costs during the year in question.
			Note: as contemplated by this condition, the Regulator will develop principles for the equitable apportionment of its costs.
		2.3	"Regulator Costs" are the costs:
			 (a) estimated by the Regulator as likely to be or which have been the costs of the Regulator in performing its functions under the Act as the economic regulator for transport and storage networks (and excluding any costs of the Regulator associated with the

			 performance by the Regulator of any of its other functions); and (b) calculated in accordance with the principles determined by the Regulator for the purpose of this condition generally (after consulting the Licensee and others likely to be affected by the application of such principles) and notified to the Licensee.
3.	T&S Network Area	3.1	This licence is granted for the "[<i>name of the relevant T&S Network to be inserted</i>]" within the Specified Area (stated in Schedule 1 (<i>Specified Area</i>) of this licence) and more particularly defined in condition A3.2.
		3.2	In this condition, "[<i>name of the relevant T&S</i> <i>Network to be inserted</i>]" means the T&S Network to be built, operated and maintained by the Licensee of which are:
			(a) the geographical location of the Onshore Transportation System is identified in Annex A to section A, as expanded or varied in accordance with the terms of this licence; and
			(b) the assets of the Offshore Transportation and Storage System are identified in Annex A to section A, as expanded or varied in accordance with the terms of this licence.

Annex A to section A

1. Geographical location of the Onshore Transportation System

[*General description of geographical location to be inserted*] as shown on the map in Exhibit A.

2. Assets of the Offshore Transportation and Storage System

[The assets being:

- (a) the Offshore Pipeline Infrastructure, the approximate route of which is shown on the map in Exhibit B, and which is subject to [Crown Estate/Crown Estate Scotland lease [●]]; and
- (b) the Storage Complex, being [insert description] as shown on the map in Exhibit B, and which is subject to carbon dioxide appraisal and storage lease number [●] and subject to [Crown Estate/Crown Estate Scotland lease [●]].]

Section B: General

1.	Conduct of T&S Business	1.1	The Licensee shall conduct its T&S Business in the manner best calculated to secure that neither:	
			(a) the Licensee or any Affiliate or Related Undertaking or Ultimate Controller of the Licensee, nor	
			(b) any User or future User,	
			obtains any unfair commercial advantage including, in particular, any such advantage from a preferential or discriminatory arrangement, being, in the case of such an advantage accruing to the Licensee, one in connection with a business other than the T&S Business.	
2.	Requirement to enter into User Arrangements in conformity with CCS Network Code	2.1	Unless otherwise agreed by the Regulator, the Licensee shall only enter into User Arrangements which are in conformity with any relevant provisions of the CCS Network Code.	
			Note: as the CCS Network Code will include template Connection Agreements and Construction Agreements, this condition will also capture the requirement for those contracts to be on standard terms.	
		2.2	The Licensee shall comply with any obligation in the CCS Network Code to disclose information relating to the operation of the T&S Network to which this licence relates.	
3.	Access to T&S Network	3.1	The Licensee must offer access to its T&S Network in accordance with the CCS Network Code and any relevant law or regulation.	
		3.2	The Licensee must make available to Users capacity in the T&S Network based on the Obligated Network Capacity as set out in condition B3.3.	
		3.3	The level of Obligated Network Capacity that the Licensee must make available and the timetable for testing the Obligated Network Capacity is set out in [section 3.4] of the APDP, unless revised in accordance with the conditions of this licence.	
			Note: the Obligated Network Capacity will be set out in [section 3.4] of the APDP.	
4.	System development obligations	4.1	The Licensee must develop, maintain and operate the T&S Network in an economic, efficient, effective and co- ordinated manner, having regard to, among other things:	
			(a) the duty to offer access in accordance with condition B3 (<i>Access to the T&S Network</i>);	

			(b) the Licensee's obligations under the CCS Network Code; and
			(c) the Licensee's obligations and commitments under the Approved Project Development Plan.
			Note: further consideration is being given to any other specific obligations that the Licensee may have in respect of the development, maintenance and operation of the T&S Network.
5.	System Development Statement	5.1	On an annual basis, by the start of each Charging Year, the Licensee shall prepare and publish a System Development Statement which sets out a forward-looking summary of the development activities that the Licensee is or will be carrying out in discharging its obligations under condition B4 (<i>System development obligations</i>).
		5.2	The System Development Statement shall also include information about the Licensee's decommissioning obligations, and proposals for the discharge of those obligations, in relation to the Onshore Transportation System and the Offshore Transportation and Storage System, as the case may be.
			Note : the System Development Statement will be a publicly available document and it is intended that it will be similar in nature to the documents published in relation to the gas and electricity networks, such as National Grid's Electricity Ten Year Statement (ETYS) and National Grid's Network Options Assessment (NOA). However, given the smaller scale and emerging nature of T&S networks, it is expected that the System Development Statement will be significantly narrower in scope than the ETYS and NOA combined. It is also expected that in preparing the System Development Statement, the Licensee will have regard to any CCUS strategy and policy statement published by the Secretary of State pursuant to the Act.
6.	CCS Network Code	6.1	The CCS Network Code is the document which:
			 (a) sets out terms that apply to the use of the T&S network of every CCS Licensee (unless otherwise provided in the CCS Network Code); and
			(b) is designed to facilitate the achievement of the Relevant Objectives.
		6.2	The Licensee shall:
			 (a) become legally bound by the provisions of the CCS Network Code by signing or acceding to the Code Agreement;

	(i)	administer the CCS Network Code in accordance with the Relevant Objectives;
	(ii)	maintain and operate procedures for the modification of the CCS Network Code so as to better facilitate the achievement of the Relevant Objectives; and
	(iii)	establish, develop and operate (or otherwise procure the operation of (including without limitation on a sub- contracted basis)) arrangements for the administration of the CCS Network Code (including its governance and modification), as well as the maintenance of a portal or website for the publication of the CCS Network Code and any information required to be published or otherwise disseminated pursuant to the provisions of the CCS Network Code; and
	admi share	e: it is envisaged that the costs of inistration of the CCS Network Code will be ed between the CCS Licensees. The details of will be further considered at a later stage.
6.3	Code sha considerat	ion of Urgent Modification Proposals, including or the Regulator to make Urgent Modification
6.4	of Urgent I shall also notices reo doing som	n to providing for the expedited consideration Modification Proposals, the CCS Network Code include operational procedures (such as quiring Users to do something or refrain from ething) for dealing with any urgent issues that rsely impact the safe and/or efficient operation 6 Network.
6.5	The CCS provide the	Network Code modification procedures shall at:
	Net res and CC Res	pposals for the modification of the CCS twork Code may be made by the Regulator in spect of a Significant Code Review or by other party to the CCS Network Code that is a S Licensee pursuant to a direction of the gulator which is made further to a Significant de Review;
	Net Sig the	oposals for the modification of the CCS twork Code falling within the scope of a inificant Code Review may not be made while Regulator is undertaking the Significant Code view except:

				(i) (ii)	where the Regulator determines that the modification proposal may be made, having taken into account (among other things) the urgency of the subject matter of the proposal; or at the direction of, or by, the Regulator; and
			(c)	Regula with modifi the op the ex	g completed a Significant Code Review, the ator may require the Licensee to comply a direction of the Regulator to make a ication to the CCS Network Code which, in inion of the Regulator, will, as compared to tisting provisions of the CCS Network Code, facilitate the achievement of the Relevant tives.
			Revie Netwo the de	w proc ork Cod	onditions relating to the Significant Code ess, which is to be set out in the CCS e, will be further developed in tandem with nent of the relevant provisions in the CCS e.
7.	Regulatory Accounts	7.1	Chargi	ing Yea ee (w	must prepare regulatory accounts for each r, for each of the following activities of the here the Licensee is conducting that
			(a)	the O	nshore Transportation System;
			(b)	the Of	fshore Pipeline Infrastructure;
			(c)	any S	torage Complex;
			(d)	Syste	n Operation;
			(e)	to wh accord	activities to which this licence relates and ich the Regulator has given its consent in dance with condition B11 (<i>Restriction on</i> <i>ry and financial ring-fencing</i>); and
			(f)	as re busine	hole business to which this licence relates, presented by the consolidation of the esses and activities referred to within sub- raphs (a) to (e), where applicable.
			be co	onsister ted in t	ll-form drafting of this licence condition will nt with Ofgem's updated approach, as he Electricity Distribution standard licence
		7.2	the r accord	regulato lance w	o far as the Regulator otherwise consents, bry accounts should be prepared in with the requirements of this condition B7 accounts).

		7.3	the Lie accour regula	the Regulator otherwise directs, after consulting censee, the Licensee shall publish its regulatory nts with the exception of any part of such tory accounts agreed by the Regulator in writing confidential.
		7.4	Unless shall:	the Regulator otherwise consents, the Licensee
			(a)	procure, in relation to its regulatory accounts an audit by an appropriate auditor and a report by that auditor, addressed to the Regulator, stating whether in the auditor's opinion those accounts fairly present the financial position, financial performance and cash flows of or reasonably attributable to the different parts of the Licensee's business in accordance with the requirements of this condition; and
			(b)	deliver to the Regulator those accounts and the auditor's reports as soon as is reasonably practicable, and in any event prior to their publication and not later than 31 July following the end of the Charging Year to which the regulatory accounts relate.
8.	Disposal of assets and restrictions on	8.1	The Li Regula	censee must not, except with the consent of the ator:
	charges	(a)	take any action that is or would be a Disposal of, or Relinquishment of Operational Control over, any T&S Asset (in whole or in part);	
			(b)	grant any mortgage, charge, or other form of security over any Receivables, contractual rights and/or any T&S Asset.
		8.2	Where	the Licensee is:
			(a)	seeking the Regulator's consent to an action otherwise prohibited under condition B8.1; or
			(b)	intending to proceed with an action that would be prohibited under condition B8.1 except that it falls within the exception set out in condition B8.3(a),
			month action Regula details	censee shall give the Regulator not less than two ns' prior written notice of its intention to take that n, together with such further information as the ator may request relating to such action (including s of the relevant Legal Requirement, where the see is seeking to rely on the exception in condition)).

	(a)	take any action referred to in condition B8.1(a) or B8.1(b) without the consent of the Regulator if any Legal Requirement requires it to do so;
	(b)	take any action referred to in condition B8.1(a) without the consent of the Regulator if:
		 (i) the Regulator has issued directions for the purposes of this condition B8 (<i>Disposal of assets and restrictions on charges</i>) generally containing a general consent (whether or not subject to conditions) to (A) transactions of a specified description; or (B) the Disposal of, or Relinquishment of Operational Control over, a T&S Asset of a specified description; and
		 (ii) the transaction or the T&S Asset subject to the proposed Disposal or Relinquishment of Operational Control are of a description to which such directions apply and the Disposal or Relinquishment of Operational Control is in accordance with any conditions to which such directions are subject;
	(c)	take any action referred to in condition B8.1(b) above where:
		 the indebtedness of the Licensee which is secured represents the novation or rollover of existing indebtedness; and
		 (ii) the proceeds of the indebtedness of the Licensee which is secured are used to fully repay the existing indebtedness referred to in condition B8.3(b)(i).
8.4	Dispos compr notwit Operat or B8. least of Relinq Asset; days Dispos	relevant T&S Asset that is subject to a proposed al or Relinquishment of Operational Control ises a significant part of the T&S Network, hstanding that the Disposal or Relinquishment of cional Control is permitted under condition B8.3(a) 3(b), the Licensee shall notify the Regulator at 50 days in advance of the proposed Disposal or uishment of Operational Control over the T&S and if the Regulator directs the Licensee within 30 of such notification, not to proceed with the al or Relinquishment of Operational Control over &S Asset, the Licensee shall comply with such on.
8.5	the T& Reling procee to exis	e purposes of condition B8.4, "a significant part of &S Network" means such that, if a Disposal or uishment of Operational Control of that part ded, then this would reduce the capacity available ting and future Users and/or cause constraints in S Network.

			Note: this condition will be reviewed in the context of any proposed financing structures.
		8.6	The restriction in condition B8.1 shall not apply to:
			(a) Disposals at arm's length and on normal commercial terms of surplus or obsolete assets which are replaced by items of equivalent or better quality in accordance with the Licensee's usual maintenance programme; or
			(b) Disposal of any asset having a market value which, when aggregated with the market value of all other assets disposed of by it under this condition in any Charging Year, does not exceed £50,000 (CPIH Indexed), provided that any such disposal (either alone or when aggregated with any related disposal) does not result in a material adverse effect on the T&S Network.
9.	Provision of information to the Regulator	9.1	Subject to condition B9.5, the Licensee shall, in addition to any requirements under the Act, furnish to the Regulator, in such manner and at such times as the Regulator may require, such information and shall procure and furnish to it such reports, as the Regulator may require or as may be necessary for the purpose of performing:
			(a) any functions transferred to or conferred on it by or under the Act; and
			(b) the regulatory functions conferred on the Regulator by other statute or enactment.
		9.2	The Licensee shall procure from each company or other person which the Licensee knows or reasonably should know is at any time an Ultimate Controller of the Licensee a legally enforceable undertaking in favour of the Licensee in a form specified by the Regulator that the Ultimate Controller (the "information covenantor") will give to the Licensee, and will procure that any person (including, without limitation, a corporate body) which is a Subsidiary Undertaking of, or is controlled by, the information covenantor (other than the Licensee and the Licensee's subsidiaries) will give to the Licensee, all such information as may be necessary to enable the Licensee to comply fully with the obligation imposed on it in condition B9.1. Such undertaking shall be obtained within 7 days of such corporate body or other person in question becoming an Ultimate Controller of the Licensee remains the holder of this licence and the information covenantor remains an Ultimate Controller of the Licensee.
		9.3	The Licensee shall deliver to the Regulator evidence (including a copy of such undertaking) that the Licensee

			has complied with the obligation to procure undertakings pursuant to condition B9.2 above, and shall comply with any direction from the Regulator to enforce any undertaking so procured.
		9.4	The Licensee shall not enter into any agreement or arrangement with any Ultimate Controller of the Licensee where the Ultimate Controller is a corporate body or any of the subsidiaries of such a corporate Ultimate Controller (other than the subsidiaries of the Licensee) at a time when:
			 (a) an undertaking complying with condition B9.2 above is not in place in relation to that Ultimate Controller;
			(b) there is an unremedied breach of such undertaking; or
			(c) the Licensee is in breach of the terms of any direction issued by the Regulator under condition B9.3.
		9.5	This condition B9 (<i>Provision of information to the Regulator</i>) shall not require the Licensee to produce any documents or give any information which it could not be compelled to produce or give in evidence in civil proceedings before a court.
		9.6	The power of the Regulator to call for information under this condition B9 (<i>Provision of information to the</i> <i>Regulator</i>) is in addition to the power of the Regulator to call for information under or pursuant to any other condition and under any statutory powers. There shall be a presumption that the provision of information in accordance with any other condition is sufficient for the purposes of that condition, but that presumption shall be rebutted if the Regulator states in writing that in its opinion such further information is, or is likely to be, necessary to enable it to exercise functions under the condition in question.
			Note: specific areas of information provision (e.g. in relation to the results of On-going Devex) remain under consideration.
10.	Prohibition of cross- subsidies	10.1	The Licensee shall procure that the T&S Business shall not give any cross-subsidy to, or receive any cross- subsidy from, any other business of the Licensee or of an Affiliate, Related Undertaking or Ultimate Controller of the Licensee.
		10.2	The Licensee may apply to the Regulator to disapply condition B10.1.

11.	Restriction activity financial fencing	on and ring-	11.1	The Licensee shall not, without the prior written conser of the Regulator, conduct any business or carry on an activity other than the T&S Business.	
	. cheng		11.2	of th	censee shall not, without the prior written consent e Regulator, hold or acquire shares or other ments of any kind except:
				(a)	shares or other investments in a body corporate the sole activity of which is to carry on the activities of the T&S Business;
				(b)	shares or other investments in a body corporate which is a Subsidiary Undertaking of the Licensee and incorporated by it solely for the purpose of raising finance for the T&S Business; or
				(c)	investments acquired in the usual and ordinary course of the Licensee's treasury management operations, subject to the Licensee maintaining in force, in relation to those operations, a system of internal controls which complies with best corporate governance practice as required (or, in the absence of any such requirement, recommended) by the UK listing authority (or a successor body) from time to time for listed companies in the United Kingdom.
			11.3	-	ct to the provisions of condition B11.2, nothing in ondition shall prevent:
				(a)	any Affiliate in which the Licensee does not hold shares or other investments from conducting any business or carrying on any activity;
				(b)	the Licensee from holding shares as, or performing the supervisory or management functions of, an investor in respect of any body corporate in which it holds an interest consistent with the provisions of this licence;
				(c)	the Licensee from performing the supervisory or management functions of a Parent Undertaking in respect of any Subsidiary Undertaking; or
				(d)	the Licensee from carrying on any business or conducting any activity to which the Regulator has given its consent.
			11.4	Affiliat the L "rele busine	ng in this condition shall prevent the Licensee or an te or Related Undertaking of the Licensee in which icensee holds shares or other investments (a vant associate ") from conducting de minimis ess as defined in this paragraph so long as the tions specified in this paragraph are complied with.

(a)	For the purpose of this paragraph, "de minimis business" means any business or activity carried on by the Licensee or a relevant associate or relevant associates other than: (i) the T&S Business; and (ii) any other business activity to which the Regulator has given its consent in writing in accordance with condition B11.3(d).
(b)	The Licensee or a relevant associate may carry on de minimis business provided that neither of the following limitations is exceeded, namely:
	(i) the aggregate turnover of all the de minimis business carried on by the Licensee and the equity share of the aggregate turnover of all the de minimis business carried on by all its relevant associates does not in any period of 12 months commencing on 1 April of any Charging Year exceed 2.5 per cent of the aggregate turnover of the Licensee as shown by its most recent audited historical cost accounting statements; and
	(ii) the aggregate amount (determined in accordance with sub-paragraph (d) below) of all investments made by the Licensee in de minimis business, carried on by the Licensee and all relevant associates, does not at any time after the date at which this condition takes effect in this licence exceed 2.5 per cent of the sum of the share capital in issue, the share premium and the consolidated reserves (including retained earnings) of the Licensee as shown by its most recent audited historical cost accounting statements.
(c)	For the purpose of sub-paragraph (b) above, "investment" means any form of financial support or assistance given by or on behalf of the Licensee for the de minimis business whether on a temporary or permanent basis and including (without limiting the generality of the foregoing) any commitment to provide any such support or assistance in the future.
(d)	At any relevant time, the amount of an investment shall be the sum of:
	 (i) the value at which such investment was included in the audited historical cost balance sheet of the Licensee as at its latest accounting reference date to have occurred prior to the date on which this condition came into effect (or, where the investment was not so included, zero);

) the aggregate gross amount of all expenditure (whether of a capital or revenue nature) howsoever incurred by the Licensee in respect of such investment in all completed accounting reference periods since such accounting reference date; and i) all commitments and liabilities (whether actual or contingent) of the Licensee relating to such investment outstanding at the end of the most recently completed accounting reference period,
			all na res ac ac	income (whether of a capital or revenue ture) howsoever received by the Licensee in spect of such investment in all completed counting reference periods since the counting reference date referred to in bparagraph (d)(i).
		11.5	relation to of the equassociate,	urpose of condition B11.4, "equity share", in o any shareholding, means the nominal value uity shares held by the Licensee in a relevant as a percentage of the nominal value of the sued equity share capital of that relevant
		11.6	investing Offshore [this condition shall prevent the Licensee from any Onshore Decommissioning Funds and any Decommissioning Funds in accordance with the s of this licence.
12.	Business separation	12.1	under this the Regula T&S Busin	prejudice to the Licensee's other obligations blicence, and unless otherwise consented to by ator, the Licensee must at all times conduct the bess separately to any other businesses carried by Affiliates of the Licensee.
		12.2	in order to	rejudice to the generality of condition B12.1, comply with that paragraph the Licensee must ar ensure that at a minimum:
			со	y businesses other than the T&S Business are nducted entirely by corporate entities which e legally separate from that of the Licensee;
			the on wh	e Licensee's accounts are maintained, and to e extent required by law audited and reported , separately from those of any corporate entity nich conducts any business other than the T&S siness;
			ma an	rsons engaged in, or in respect of, the anagement or operation of the Licensee (up to d including the members of the senior anagement team reporting to the Licensee's

		board of directors) are not simultaneously engaged either full or part time in respect of any business other than the T&S Business or any corporate entity which conducts any business other than the T&S Business without the prior consent of the Regulator;
	(d)	arrangements are in place which are effective in restricting access by persons engaged in or in respect of the management or operation of any business other than the T&S Business to any part of any premises which is occupied by persons engaged in, or in respect of, the management and operation of the T&S Business, unless with the prior consent of the Regulator; and
	(e)	the systems for the recording, processing or storage of information and data used by persons engaged in or in respect of, the management or operation of the T&S Business cannot be accessed by persons engaged in, or engaged in respect of, the management or operation of any business other than the T&S Business.
12.	the the with	Licensee shall ensure, following consultation with Regulator, that a compliance officer is appointed for purpose of facilitating compliance by the Licensee the business separation duties pursuant to this dition.
12.		duties and tasks of the compliance officer shall ude:
	(a)	providing advice and information to the Licensee (including directors of the Licensee) and for the purpose of ensuring the Licensee's compliance with the specified business separation duties;
	(b)	monitoring the effectiveness of the practices, procedures and systems adopted by the Licensee to ensure its compliance with the specified business separation duties; and
	(c)	providing information about compliance with the duties for the purpose of the compliance statement to be prepared by the Licensee under condition B12.5.
12.	and the of t cove prod	Licensee shall, within [90 days] of Licence Award after that no later than every 12 months, provide to Regulator a report setting out details of the activities the compliance officer during the relevant period ered by the compliance report, as well as the cedures in place to ensure that the Licensee is aplying with this condition.
12.		compliance report referred to in condition B12.5 Il be accompanied by a compliance certificate, in a

			form approved by the Regulator, certifying the accuracy of the compliance report, and which has been approved by a resolution of the Licensee's board of directors and signed by a director of the Licensee.
13.	Availability of resources	13.1	The Licensee shall at all times act in a manner calculated to secure that it has available to itself such resources, including (without limitation) management and financial resources, personnel, fixed and moveable assets, rights, licences, consents and facilities, on such terms and with such rights, as shall ensure that it is at all times able:
			(a) to properly, economically, efficiently and effectively carry on the T&S Business; and
			(b) to comply in all respects with its obligations under this licence, such obligations under the [the Act] as apply to those activities authorised by this licence and the CCS Network Code.
		13.2	The Licensee must by 31 July each year give the Regulator a certificate that has been approved by a resolution of the Licensee's board of directors and signed by a director of the Licensee pursuant to that resolution, stating that:
			(a) after making enquiries and having taken into account in particular (but without limitation) any dividend or other distribution that might reasonably be expected to be declared or paid by the Licensee, the Licensee's directors have a reasonable expectation that the Licensee will have sufficient financial resources and financial facilities available to itself to enable the Licensee to carry on the T&S Business for a period of 12 months from the date of the certificate;
			(b) after making enquiries the Licensee's directors have a reasonable expectation that the Licensee will have sufficient operational resources including management, personnel, fixed and moveable assets, rights, licences, consents, and facilities available to itself to enable the Licensee to carry on the T&S Business for a period of 12 months from the date of the certificate; and
			(c) after making enquiries the Licensee's directors consider that, at the time of their approval of this certificate, the Licensee is in compliance in all material respects with all of the obligations imposed on it by the licence.
		13.3	The Licensee shall only declare or pay dividends or otherwise make distributions in accordance with a distribution policy which has been approved by the board of the Licensee and which complies with the principle that the dividends or distributions declared or paid will

			not impa Business	air the ability of the Licensee to finance the T&S 5.
				perating model for T&SCos remains subject to ncluding as part of the cost assessment process.
14.	Undertaking from Ultimate Controller	14.1	person v Licensee the Licen that Ulti from an (includir a Subsi covenan will refra to cause under th to in con the com Ultimate as the L	Insee shall procure from each company or other which is at any time an Ultimate Controller of the e a legally enforceable undertaking in favour of insee in the form specified by the Regulator that mate Controller (the "covenantor") will refrain by action, and will procure that any person ing, without limitation, a corporate body) which is idiary Undertaking of or controlled by, the tor (other than the Licensee and its subsidiaries) ain from any action, which would then be likely the Licensee to breach any of its obligations he Act or this licence. The undertaking referred indition B14.1 shall be obtained within 7 days of pany or other person in question becoming an e Controller and shall remain in force for as long Licensee remains the holder of this licence and enantor remains an Ultimate Controller of the
		14.2	The Lice	nsee shall:
			c ł	deliver to the Regulator evidence (including a copy of each such undertaking) that the Licensee has complied with its obligation to procure undertakings pursuant to condition B14.1;
			c s	nform the Regulator immediately in writing if the lirectors of the Licensee become aware that any such undertaking has ceased to be legally enforceable or that its terms have been preached; and
				comply with any direction from the Regulator to enforce any such undertaking;
			Regulat agreem of the L corpora	all not, save with the consent in writing of the or, enter (directly or indirectly) into any ent or arrangement with any Ultimate Controller Licensee or any of the subsidiaries of any such te Ultimate Controller (other than the aries of the Licensee) at a time when:
			(an undertaking complying with condition B14.1 is not in place in relation to that Ultimate Controller; or
			(ii) there is an unremedied breach of such undertaking; or

	Ι		
			(iii) the Licensee is in breach of the terms of any direction issued by the Regulator as referred to in condition B14.2.
		14.3	The Licensee must, on or before 31 July of each Charging Year, provide the Regulator with a schedule of the undertakings obtained in accordance with condition B14.1 that are in force at that time, together with a confirmation that the Licensee has sent each of the Ultimate Controllers concerned a letter, within the preceding 12 months, appraising the Ultimate Controller of the terms of the undertaking that it has given.
15.	Qualifying Acquisition	15.1	For the purposes of this condition and Schedule 7 (<i>Qualifying Acquisition Information</i>):
			"Qualifying Acquisition" means an event in which a person:
			(a) acquires direct or indirect control of the Licensee; or
			(b) acquires direct or indirect control of the T&S Network.
		15.2	For the purposes of condition B15.1(a) and subject to condition B15.4, "control" means either
			(a) where the percentage of shares or voting rights that the person holds in the Licensee increases:
			(i) from 25% or less to more than 25%;
			(ii) from 50% or less to more than 50%;
			(iii) from less than 75% to more than 75%, or
			(b) where the acquisition is of voting rights in the Licensee that (whether alone or together with other voting rights held by the person) enable the person to secure or prevent the passage of any class of shareholder resolution governing the affairs of the Licensee with respect to all or substantially all matters relating to the T&S Business.
		15.3	For the purposes of condition B15.1(b) and subject to condition B15.4, "control" means the acquisition of a right or interest such that the transferee is able:
			(a) to use the asset, or use it to a greater extent than prior to the acquisition; or
			(b) to direct or control how the asset is used, or direct or control how it is used to a greater extent than prior to the acquisition.

15.4	For the purposes of condition B15.1, "indirect control" means where the person does not directly hold the relevant right or interest, but holds a majority stake in an entity and that entity:
	(a) holds the interest or right; or
	(b) is part of a chain of entities and either (i) each of those entities other than the last has a majority stake in the entity immediately below it in the chain, and the last entity holds the interest or right, or (ii) if the last entity in the chain holds a majority stake in the Licensee, all but one of the other entities in the chain hold a majority stake in the entity immediately below it in the chain, and the other entity in the chain holds more than 25% of the shares or voting rights in the entity immediately below it.
15.5	For the purposes of condition B15.4, A has a "majority stake" in B if:
	(a) A holds a majority of voting rights in B,
	 (b) A is a member of B and has the right to appoint or remove a majority of the board of directors of B,
	(c) A is a member of B and controls alone, pursuant to an agreement with other shareholders or members, a majority of the voting rights in B, or
	 (d) A has the right to exercise, or actually exercises, dominant influence over B for the purposes of section 1162(2)(c) and paragraph 4 of schedule 7 of the Companies Act 2006,
	where "majority" means a simple majority of more than 50%.
15.6	In condition B15.5:
	(a) the reference to the right to appoint or remove a majority of the board of directors of an entity is to the right to appoint or remove directors holding a majority of the voting rights at meetings of the board on all or substantially all matters,
	(b) the reference to the board of directors, in the case of an entity that does not have such a board, is to be read as a reference to the equivalent management body of that entity.
15.7	The Licensee shall inform the Regulator in writing and provide the information set out in Schedule 7 (<i>Qualifying Acquisition Information</i>) of which it is aware or ought reasonably to be aware, having made reasonable

	15.8	enquiries, not less than [30] Business Days in advance of completion of any Qualifying Acquisition of which it is aware or ought reasonably to be aware, including in light of the provisions of condition B15.8. The Licensee shall procure from each company or other person which the Licensee knows or reasonably chould
		person which the Licensee knows or reasonably should know is at any time an Ultimate Controller of the Licensee a legally enforceable undertaking in favour of the Licensee that the Ultimate Controller (the "information covenantor") will give to the Licensee, and will procure that any person (including, without limitation, a corporate body) which is a Subsidiary Undertaking of, or is controlled by, the information covenantor (other than the Licensee and its subsidiaries) will give to the Licensee all such information as may be necessary to enable the Licensee to comply fully with the obligations imposed on it by condition B15.7. Such undertaking shall be obtained upon such corporate body or other person in question becoming an Ultimate Controller of the Licensee remains the holder of this licence and the information covenantor remains an Ultimate Controller of the Licensee.
and proper requirement	16.1	The Licensee must not appoint or have in place a person in a position of Significant Managerial Responsibility or Influence who is not a fit and proper person to occupy that role.
	16.2	The Licensee must:
		(a) have and maintain robust processes, systems and governance in place to ensure that any person holding a position of Significant Managerial Responsibility or Influence in the Licensee is fit and proper to occupy that role; and
		 (b) carry out regular assessments on such person(s) to ensure that they remain fit and proper to occupy that role.
	16.3	In complying with conditions B16.1 and B16.2, the Licensee must have regard to and take account of all relevant matters including, but not limited to, whether the individual has:
		 (a) been responsible for, contributed to or facilitated any serious misconduct or mismanagement (whether unlawful or not) in the course of carrying out a regulated activity (or, providing a service elsewhere which if provided in Great Britain, would be a regulated activity);
		 (b) any relevant unspent criminal convictions in any jurisdiction in particular fraud or money laundering;

			(c)	any insolvency history, including undischarged bankruptcy, debt judgements and County Court judgements;
			(d)	been disqualified from acting as a director of a company;
			(e)	been a person with Significant Managerial Responsibility or Influence at a current or former CCS Licensee in respect of whose T&S Network the Regulator issued a T&SCo of Last Resort Direction (including where they were a person with Significant Managerial Responsibility or Influence at that CCS Licensee within the 12 months prior to the T&SCo of Last Resort Direction being issued); and/or
			(f)	been refused, had revoked, restricted or terminated any form of authorisation, or had any disciplinary, compliance, enforcement or regulatory action taken by any regulatory body in any jurisdiction whether as an individual, or in relation to a business in which that person held Significant Managerial Responsibility or Influence.
		16.4	backg Great	Licensee must give particular regard to nstances in which the relevant person has a round in the carbon capture and storage sector in Britain and the previous actions of that person ed in or contributed towards significant market nent.
17.	Financial Resilience and Credit Quality	17.1	In accordance with the other provisions of this licence, the Licensee is required to ensure it has the financial resources and standing needed to enable the Licensee to comply with the terms and conditions of this licence. During the First Regulatory Period, the Licensee is not required to obtain or maintain an Investment Grade Issuer Credit Rating.	
		17.2	subject accord the Lid the F prevail are obtain Grade	ring the expiry of the First Regulatory Period and ct to condition B17.4 below, the Regulator may, in dance with the RES/RAS Process Guidance, require censee to undertake a RES/RAS Process to inform Regulator's assessment of whether the then iling conditions or circumstances of the Licensee appropriate to introduce an obligation to n/maintain would enable it to obtain an Investment Issuer Credit Rating of at least BBB (S&P, Fitch or Morningstar) or Baa2 (Moody's).
		17.3	and su	ying the Licensee undertaking a RES/RAS Process ubject to condition B17.5 below, the Regulator may ordance with the Investment Grade Issuer Credit g Guidance, require that the Licensee use all

				avours to obtain and maintain an e Issuer Credit Rating.
		17.4	under condition E prior thereto, it among other thing would need to app	all only be entitled to exercise its rights $[317.2 \text{ if, by no less than } [\bullet] months has issued guidance in relation to, gs, the conditions or circumstances that by before the Regulator introduces the the Licensee to undertake a RES/RAS$
		17.5	under condition E prior thereto, it among other thing would need to app requirement for t Investment Grade of the conditions would enable it t	all only be entitled to exercise its rights 317.3 if, by no less than [●] months has issued guidance in relation to, gs, the conditions or circumstances that oly before the Regulator introduces the he Licensee to obtain and maintain an e Issuer Credit Rating, taking account or circumstances of the Licensee that to obtain an Investment Grade Issuer prmed by the results of the RES/RAS
18.	Credit rating	18.1	for the Licensee accordance with provide to the Re reasonable adjust the cost of under	otifies the Licensee of the requirement to undertake a RES/RAS Process in condition B18.2, the Licensee shall egulator details of its analysis of any ment to the Opex to take into account taking the RES/RAS Process, and the termine any adjustment to the ex ante
		18.2	for the Licensee to Grade Issuer Creativity with effect during provide to the Re adjustments requirements inclu-	otifies the Licensee of the requirement to obtain and maintain an Investment dit Rating, pursuant to condition B17.3, a regulatory period, the Licensee shall egulator details of its analysis of the puired to take into account such uding but not limited to:
			a result o	tments to the Revenue Calculations as f the Investment Grade Issuer Credit cluding without limitation:
			any (inc sha allo Ope	reasonable increases or decreases to rex ante allowance (with cost sharing) cluding: (i) any change to the cost ring factor; and (ii) any ex ante wance in respect of SRAV Capex, ex, On-going Capex or On-going vex); and
			mai any prir	r reasonable amendments to the nner in which the Regulator will assess r ex post costs to apply the same nciples that the Licensee should not be cost risk for obtaining and maintaining

				Investment Grade Issuer Credit Rating; and/or
			(ii	ii) any adjustments to the WACC for the Operational Period as determined in accordance with condition H8 (WACC).
		18.3	requirem Grade Is	gulator will determine the impact of the ent to obtain and maintain the Investment suer Credit Rating on the Project/the Licensee adjustments to be provided to the Licensee.
		18.4	-	notification by the Regulator under condition he Licensee shall provide to the Regulator:
			co Li	ny additional information that the Regulator onsiders appropriate and requests from the icensee in relation to obtaining and maintaining ne Investment Grade Issuer Credit Rating; and
			de cc Re Ir th	there applicable in the context of an increase or ecrease of the Licensee's costs pursuant to ondition B18.3, satisfactory evidence to the egulator that obtaining and maintaining the nvestment Grade Issuer Credit Rating affects he cost impact to On-going Devex, Capex or pex.
		18.5	pursuant Licensee and main Rating af	bunt of any increases to the Licensee's costs to condition B18.3 shall be subject to the providing satisfactory evidence that obtaining ntaining the Investment Grade Issuer Credit fects the cost impact to On-going Devex, Capex pursuant to condition B18.4.
19.	Indebtedness	19.1	In addition to the requirements of condition B8 (<i>Disposal</i> of assets and restrictions on charges over receivables), the Licensee shall not without the prior written consent of the Regulator (following the disclosure by the Licensee of all material facts):	
			ai of ui oi ef	reate or continue or permit to remain in effect ny mortgage, charge, pledge, lien or other form f security or encumbrance whatsoever, ndertake any indebtedness to any other person r enter into or continue or permit to remain in ffect any guarantee or any obligation otherwise nan:
			(i) on an arm's length basis;
			(i	i) on normal commercial terms;
			(i	ii) for a Permitted Purpose; and
			(i	v) if the transaction is within the ambit of condition B8 (<i>Disposal of assets and</i>

	restrictions on charges over receivables) in accordance with that condition;
(b)	transfer, lease, license or lend any sum or sums, asset, right or benefit to any associate of the Licensee otherwise than by way of:
	 a dividend or other distribution out of distributable reserves;
	(ii) repayment of capital;
	 (iii) payment properly due for any goods, services or assets provided on an arm's length basis and on normal commercial terms;
	 (iv) a transfer, lease, licence or loan of any sum or sums, asset, right or benefit on an arm's length basis, on normal commercial terms and made in compliance with the payment condition referred to in condition B19.2;
	(v) repayment of or payment of interest on a loan not prohibited by sub-paragraph (a);
	 (vi) payments for group corporation tax relief calculated on a basis not exceeding the value of the benefit received; or
	 (vii) an acquisition of shares or other investments in conformity with paragraph 2 of condition B11 (<i>Restriction on activity</i> <i>and financial ring-fencing</i>) made on an arm's length basis and on normal commercial terms; or
(c)	enter into an agreement or incur a commitment incorporating a cross-default obligation or continue or permit to remain in effect any agreement or commitment incorporating a cross- default obligation, provided, however, that the provisions of condition B19.1 shall not prevent the Licensee from giving any guarantee permitted by and compliant with the requirements of condition B19.1(a).
the	e payment condition referred to in condition 0.1(b)(iv) is that the consideration due in respect of transaction in question is paid in full when the nsaction is entered into unless either:
(a)	the counter-party to the transaction has and maintains until payment is made in full an investment grade issuer credit rating; or

			(b) the obligations of the counter-party to the transaction are fully and unconditionally guaranteed throughout the period during which any part of the consideration remains outstanding by a guarantor which has and maintains an investment grade issuer credit			
20.	Regulatory Reporting	20.1	rating. The Licensee must comply with the regulatory and revenue reporting obligations which are to be set out in the regulatory and revenue reporting guidance published by the Regulator, providing all required information using the templates included in the guidance, by 31 July of the Charging Year to which the information relates.			
		20.2				
			Note: it is intended that guidance on regulatory reporting will be developed by the Regulator.			
21.	T&SCo of Last Resort	21.1	21.1 The Licensee shall at all times comply with any T&SCo of Last Resort Direction that has been given or varied by the Regulator pursuant to the licence.			
		21.2 The Regulator may, following consultation with the Licensee and any other CCS Licensees affected thereby, give a T&SCo of Last Resort Direction to provide carbon dioxide transport and storage services for a period not exceeding five years where the Regulator intends to revoke the licence of another CCS Licensee.				
		21.3	3 The Regulator may only give a T&SCo of Last Resort Direction to the Licensee if:			
			(a) it considers that the Licensee could comply with the T&SCo of Last Resort Direction without materially prejudicing the Licensee's ability to:			
			(i) continue to carry out its activities pursuant to this licence, and			
			(ii) fulfil its contractual obligations under the CCS Network Code;			
			 (b) it is satisfied that the Licensee is able to operate the relevant T&S network in an economic, efficient and effective manner; 			
			(c) it is satisfied that the Licensee will be able to finance the activities which are the subject of obligations to be imposed on it by or by virtue of the T&SCo of Last Resort Direction;			

			(d)	it is satisfied that the Licensee will be able to recover the costs of operating the relevant T&S network in an economic, efficient and effective manner, including a reasonable rate of return;
			(e)	it has given notice to the Licensee, pursuant to this condition, of its intention to give a T&SCo of Last Resort Direction and specified a reasonable period (not being less than 14 days from the date of publication of the notice) within which the Licensee may make representations to the proposed T&SCo of Last Resort Direction; and
			(f)	it has considered any representations made by the Licensee and not withdrawn the notice.
		21.4	unles modif to [se and o releva the Li trans	SCo of Last Resort Direction shall not take effect s the Regulator has formally proposed fications to the conditions of this licence, pursuant ection 13 of the Act], that will prescribe the rights obligations of the Licensee with respect to the ant T&S network, including the annual revenue that censee is able to earn for providing carbon dioxide port and storage services through the relevant T&S prk in an economic, efficient and effective manner.
		21.5	entitle Licens prope	icensee acknowledges that where the Regulator is ed to issue a T&SCo of Last Resort Direction, the see may be the intended transferee of designated erty, rights or liabilities under a Transfer Scheme erning another CCS Licensee.
		21.6	Direct on the of an	aking a decision on issuing a T&SCo of Last Resort tion the Regulator shall have regard to its guidance e T&SCo of Last Resort mechanism, but in the event y conflict between this licence condition and the nce, this licence condition will prevail.
		<i>be app to the networ the Ol</i>	lied wh regime rks. The FTO of	ended that a T&SCo of Last Resort regime will only pere reasonably practicable to implement, similarly that applies in the context of energy suppliers and e drafting in condition B21.3 is closely modelled on Last Resort standard condition included in the msmission licence standard licence conditions.
22.	Procurement obligations	22.1	which Licens terms	Licensee must ensure that all goods and services a are procured by the Licensee in the course of the see carrying out the T&S Business are procured on a which are designed to achieve the most economic, ent, and effective procurement of those goods and ces.
		22.2	other procu	out limiting the generality of condition B22.1, unless wise agreed by the Regulator, when the Licensee is ring goods and services in the carrying out of the Business, the Licensee must:

		(a)	follow procurement processes which:
			(i) treat potential providers equally and without discrimination;
			(ii) are transparent;
			(iii) are designed to ensure an effective competition by ensuring:
			 (A) such processes do not exclude particular providers, or types/categories of providers, or otherwise artificially restrict or narrow competition;
			(B) any contract award decision is to be made on the basis of the most economically advantageous tenders, or some other clear, transparent and objective award criteria, which in each case are made available in advance to potential providers, are linked to a clear and objective scoring methodology and linked to the subject matter of the contract being procured; and
			 (iv) are consistent with Good Industry Practice for the procurement of such goods and services;
		(b)	act in a transparent and proportionate manner;
		(c)	carry out a competitive tender process or where this is not possible, incorporate appropriate procurement benchmarking in its procurement processes;
		(d)	ensure that any contracts entered into as a result of the procurement process are:
			(i) on an arm's length basis; and
			(ii) on normal commercial terms; and
		(e)	have regard to any Local Content and other supply chain requirements, including but not limited to commitments under the North Sea Transition Deal that apply to the Licensee.
23.	Data Best Practice	-	ourpose of this condition is to set out the Licensee's ations to comply with Data Best Practice Guidance.

			23.2	This condition also sets out the process the Regulator will follow when issuing and amending Data Best Practice Guidance.		
			23.3	The Licensee must, when conducting work that involve working with or making decisions about the use Energy System Data, use its best endeavours to act accordance with Data Best Practice Guidance.		
			23.4	The Regulator will issue and amend Data Best Pract Guidance by direction.		
			23.5	The Regulator will publish Data Best Practice Guidand on the Regulator's website.		
			23.6	Data Best Practice Guidance will make provision about how the Regulator expects the Licensee to comply with data best practice to generate benefits for T&S Network Users, stakeholders or the public interest, including but not limited to ensuring services that involve Energy System Data are designed to meet the needs of T&S Network Users, stakeholders or the public interest.		
			23.7		e issuing Data Best Practice Guidance by direction, egulator will publish on the Regulator's website:	
				(a)	the text of the proposed guidance;	
				(b)	the date on which the Regulator intends the guidance to come into effect; and	
				(c)	a period during which representations may be made on the content of the guidance, which will not be less than 28 days.	
			23.8	Before amending Data Best Practice Guidance b direction, the Regulator will publish on the Regulator website:		
				(a)	the text of the amended guidance;	
				(b)	the date on which the Regulator intends the amended guidance to come into effect;	
				(c)	the reasons for the amendments to the guidance; and	
				(d)	a period during which representations may be made on the amendments to the guidance, which will not be less than 28 days.	
24.	Requirement Sufficiently Independent Directors	for	24.1	that must execu	ct to condition B24.11, except and to the extent the Regulator consents otherwise, the Licensee ensure that at all times it has at least two non- tive directors who meet the criteria set out in tions B23.2, B24.3 and B24.5 below. In this	

		ion B24, such directors are referred to as ciently independent directors".
24.2	A suffi	ciently independent director must:
	(a)	be a natural person; and
	(b)	in the reasonable opinion of the Licensee, have the skills, knowledge, experience, and personal qualities necessary to perform effectively as a non-executive director of the Licensee; and
	(c)	not have any executive duties within the T&S Business.
24.3	otherw indepe been d a direc	and to the extent that the Regulator consents vise, and subject to condition B24.4, a sufficiently endent director must not be, and must not have during the 12 months before their appointment as stor of the Licensee or the coming into force of this ion (whichever is the later):
	(a)	an employee of the Licensee; or
	(b)	a director or employee of an Associate of the Licensee.
24.4	not inc a whol incorpo	ference to "director" in condition B24.3(b) does clude appointment as a non-executive director of ly-owned subsidiary of the Licensee that has been orated by it solely for the purpose of raising e for a Permitted Purpose.
24.5	A suffi	ciently independent director must not:
	(a)	have, or have had during the 12 months before their appointment as a director or the coming into force of this condition (whichever is the later), any material business relationship with the Licensee or any Associate of the Licensee;
	(b)	hold a remit to represent the interests of any particular shareholder or group of shareholders of the Licensee or the interests of any Associate of the Licensee; or
	(c)	receive remuneration from the Licensee or any Associate of the Licensee apart from a director's fee and reasonable expenses.
24.6	For the respec	e purposes of conditions B24.5(a) and B24.5(c) tively:
	(a)	the holding of a small number of shares or associated rights shall not, of itself, be considered a material business relationship; and

	(b) the receipt or retention of any benefit accrued as a result of prior employment by or service with the Licensee or any Associate of the Licensee shall not be considered to be remuneration.
24.7	The Licensee must notify the Regulator of the names of its sufficiently independent directors within 14 days of [Licence Award] and must notify the Regulator within 14 days where any new directors are appointed to fulfil the obligation in condition B24.11.
24.8	The terms of appointment of each sufficiently independent director must include a condition stipulating that both the Licensee and the appointee will use their best endeavours to ensure that the appointee remains sufficiently independent during their term of office, having particular regard to the criteria set out in conditions B24.2, B24.3 and B24.5.
24.9	A term of appointment for a sufficiently independent director may not be for longer than eight years, but an individual may be reappointed thereafter provided that they continue to meet the criteria set out in conditions B24.2, B24.3 and B24.5.
24.10	The Licensee must notify the Regulator in writing within 14 days if any sufficiently independent director is removed from office or resigns, giving reasons for the removal or (to the extent that they are known to the Licensee) the resignation. For the purposes of this requirement, the reasons for a resignation may, if appropriate, be stated to be personal reasons.
24.11	If at any time the Licensee has fewer than two sufficiently independent directors because of a removal or resignation or other reason (including death or incapacity), the Licensee must use its reasonable endeavours to ensure that a new director is, or new directors are, appointed to fulfil the obligation in condition B24.1 as soon as is reasonably practicable to bring the number of sufficiently independent directors up to at least two.

Section C: Onshore Standard Conditions

1.	Application of Onshore Standard Conditions	1.1 These standard conditions apply where any part of the T&S Network to which this licence applies includes an Onshore Transportation System.			
2.	Obligations relating to onshore transport of CO ₂ and other obligations	2.1	2.1 The Licensee shall comply with all Legal Requirements that apply to the Onshore Transportation System.		
3.	Decommissioning obligations	3.1 The Licensee shall comply with the Onshore Decommissioning Requirements that apply to the Onshore Transportation System			
		3.2	3.2 The Licensee shall establish an Onshore Decommissioning Fund, which shall:		
			• •	e requirements of the Regulator in to ringfencing; and	
			(b) have suff	ficient funds to:	
				rry out the activities contemplated by e Onshore Decommissioning Plan; and	
			(ii) co	mply with condition C3.1.	
		3.3	nonths of Licen	mply with condition C3.2, within [12 ce Award], the Licensee shall prepare, le Regulator for approval:	
		(a) the proposed structure for the Onshore Decommissioning Fund which meets the ringfencing requirements of the Regulator;			
			-	guidance will be provided on the ng/holding requirements.	
			the de Transpor	ore decommissioning plan in relation to ecommissioning of the Onshore tation System ("Onshore hissioning Plan") which:	
			th to Tr co	accordance with the Onshore ecommissioning Requirements, sets out e measures that the Licensee proposes take to decommission the Onshore ansportation System and otherwise omply with the Onshore ecommissioning Requirements; and	
			th co	entains an estimate of the total cost of e measures proposed in it (including a entingency amount to cover unexpected ests); and	

	 (c) an investment strategy designed to [maintain the value] of the Onshore Decommissioning Fund ("Onshore Decommissioning Fund Investment Strategy").
	Note: general guidance relating to the investment strategy will be developed by Ofgem in due course. It is expected that a single investment strategy will apply across different onshore and offshore decommissioning funds established by the Licensee in accordance with the licence conditions.
3.4	The Licensee shall use all reasonable endeavours to ensure that the Onshore Decommissioning Plan accurately reflects:
	 (a) the measures required to decommission the Onshore Transportation System (and carry out any related activities); and
	(b) the economic, efficient and effective cost of those measures,
	in accordance with the Onshore Decommissioning Requirements that apply to the Onshore Transportation System.
3.5	The Regulator may require further information to be provided in relation to the Onshore Decommissioning Plan and the Onshore Decommissioning Fund Investment Strategy submitted to it under condition C3.3 and may also require the Licensee to make changes to any of these before they are approved.
3.6	Once the Regulator has provided its approval under condition C3.3, the Licensee shall calculate the Onshore Decommissioning Fund Contribution Amount, using the Onshore Decommissioning Fund Accrual Profile, and shall submit to the Regulator for approval:
	(a) the Onshore Decommissioning Fund Contribution Amount that the Licensee has calculated; and
	(b) details of how that calculation has been made, having regard to the cost estimate set out in the Onshore Decommissioning Plan, the Onshore Decommissioning Fund Investment Strategy and the expected operational life of the Onshore Transportation System.
3.7	The Regulator may require further information to be provided in relation to how the Licensee has calculated the Onshore Decommissioning Fund Contribution Amount and may also require the Licensee to make changes before approving the Onshore Decommissioning Fund Contribution Amount.

	3.8	condit Comm Transp the (Amoun Account	the Regulator has provided its approval under tion C3.6, in each Charging Year starting from the mercial Operations Date of the Onshore sportation System, the Licensee shall promptly pay Onshore Decommissioning Fund Contribution unt into the Onshore Decommissioning Fund unt [by the end of each calendar month of that ging Year].
			e: the hierarchy and mechanics of payment to be loped in the context of the financing workstream.
	3.9		icensee shall not make any withdrawals from the ore Decommissioning Fund Account other than for:
		(a)	the purpose of investment in accordance with the Onshore Decommissioning Fund Investment Strategy; and
		(b)	with the approval of the Regulator, for the purposes of carrying out any decommissioning of any part of the Onshore Transportation System (or related activities) as required by the Onshore Decommissioning Authority or any other regulator in accordance with the Onshore Decommissioning Requirements.
		fund v the e	e: further provisions to be developed to deal with windfall/shortfall, as well as access to the fund in event that the Licensee does not comply with its mmissioning obligations.
	3.10	Decom calcula	Onshore Decommissioning Plan, the Onshore mmissioning Fund Investment Strategy and the lation of the Onshore Decommissioning Fund ribution Amount shall be subject to review by the lator:
		(a)	at each Periodic Review; and
		(b)	outside of a Periodic Review if:
			 the Onshore Decommissioning Plan prepared by the Licensee no longer accurately reflects the estimated cost of the decommissioning measures (and related activities) that will need to be undertaken as a result of substantial changes to the Onshore Transportation System; or
			 (ii) other circumstances mean that there may be a shortfall in the funds required to carry out decommissioning of the Onshore Transportation System (or related activities) in accordance with the Onshore Decommissioning Requirements.

		3.11	The Licensee shall notify the Regulator, as soon as reasonably practicable, upon the occurrence of any circumstances referenced in conditions C3.10(b)(i) or C3.10(b)(ii).
		3.12	Where the Onshore Transportation System is significantly expanded, such that new decommissioning requirements are imposed by an Onshore Decommissioning Authority in relation to a new section of the Onshore Transportation System, the Regulator may direct the Licensee to prepare a new Onshore Decommissioning Plan (instead of updating an existing Onshore Decommissioning Plan) and establish a separate Onshore Decommissioning Fund in relation to the new section of the Onshore Transportation System, in accordance with this condition C3 (<i>Decommissioning obligations</i>).
4.	UK Emissions Trading Scheme	4.1	The Licensee shall comply with all obligations arising under the UK ETS in relation to all carbon dioxide that is delivered by Users into the Onshore Transportation System.
5.	Metering	5.1	The Licensee shall be responsible for ensuring that appropriate metering equipment is installed, operated and maintained to accurately measure the rate of flow and composition of carbon dioxide being delivered by Users into the T&S Network.
		5.2	Without limiting the generality of condition C5.1, the Licensee shall ensure that the metering on the T&S Network complies with the requirements of:
			(a) UK ETS;
			(b) the NSTA;
			(c) the CCS Network Code; and
			(d) any other Legal Requirements,
			including but not limited to any requirements in respect of the location and specification of the metering and any requirements to identify CO_2 leakage on the Onshore Transportation System, Offshore Transportation and Storage System and any Storage Complex.
		5.3	The Regulator may require the Licensee to provide evidence of verification by an independent technical expert of compliance of the metering equipment with the relevant standards and levels of accuracy, as required by this condition.
			Note: metering obligations remain subject to further consideration. In particular, a timeline for the UK ETS permit needs to be established which allows settlement

				ated issues under the licence, including design and g of metering across the T&S Network.
6.	Obligations in relation to carbon dioxide storage	6.1	the L Netwo	censee shall ensure that any carbon dioxide that icensee accepts is contained within the T&S rk in accordance with the requirements of the e and either:
			(a)	stored in a Storage Complex in relation to which the Licensee holds a Storage Licence and Storage Permit (where the Offshore Transportation and Storage System is also operated by the Licensee); or
			(b)	delivered to an Onshore Transportation System or an Offshore Transportation and Storage System operated by another CCS Licensee.

1.	Application of Offshore Standard Conditions	1.1	These standard conditions apply where any part of the T&S Network to which this licence applies includes an Offshore Transportation and Storage System.
2.	Obligations relating to offshore transport of CO ₂ and other obligations	2.1	The Licensee shall comply with all Legal Requirements that apply to the Offshore Transportation and Storage System.
3.	-	3.1	 The Licensee shall comply with the Offshore Decommissioning Requirements that apply to the Offshore Transportation and Storage System. Note: this condition sets out a preliminary framework for the Licensee's obligations relating to the establishment of the offshore decommissioning fund or funds. It will require further development to reflect the statutory regime that will underpin the establishment of an offshore decommissioning fund). It is expected that there will be fund coverage for installation and well decommissioning, post-decommissioning monitoring, post-licence termination monitoring costs and other contributions under the Storage of Carbon Dioxide (Termination of Licences) Regulations 2011, post-closure financial security costs, and post-closure Supplemental Compensation Agreement charges. The Licensee shall establish an Offshore Decommissioning Requirements and the Regulator in relation to ringfencing; and (b) have sufficient funds to: (i) carry out the activities contemplated by the Approved Offshore Decommissioning Programme; (ii) cover the post-closure insurances and Supplemental Compensation Agreement charges.
			Note : paragraph (b) above will be developed to deal in more detail with the different categories of costs referenced in the note under condition D3.1.

3.3	In order to comply with condition D3.2, within [12 months of Licence Award], the Licensee shall submit to the Regulator:
	(a) the proposed structure for the Offshore Decommissioning Fund which meets the ringfencing requirements of the Offshore Decommissioning Requirements and the Regulator;
	Note : guidance will be provided on the ringfencing/holding requirements.
	(b) a copy of the Approved Offshore Decommissioning Programme (including an estimate of the total cost of the measures proposed in it, which includes a contingency amount to cover unexpected costs, as approved by the Offshore Decommissioning Authority), together with evidence of its approval by the Offshore Decommissioning Authority; and
	Note: paragraph (b) above only deals with the Approved Offshore Decommissioning Programme. The process for approval of other costs is to be developed.
	 (c) an investment strategy designed to maintain the value of the Offshore Decommissioning Fund (the "Offshore Decommissioning Fund Investment Strategy"), which must be approved by the Regulator.
	Note: general guidance relating to the investment strategy will be developed by Ofgem in due course. It is expected that a single investment strategy will apply across different onshore and offshore decommissioning funds established by the Licensee in accordance with the licence conditions.
	Note: mechanism for dealing with payment to cover liability for decommissioning of Re-use Assets, into the Fund, to be developed.
3.4	The Regulator may require further information to be provided in relation to the Offshore Decommissioning Fund Investment Strategy submitted to it under condition D3.3 and may also require the Licensee to make changes to the Offshore Decommissioning Fund Investment Strategy before it is approved.
3.5	Once the Regulator has provided its approval under condition D3.3, the Licensee shall calculate the Offshore Decommissioning Fund Contribution Amount, using the Offshore Decommissioning Fund Accrual Profile, and shall submit to the Regulator for approval:

	(a)	the Offshore Decommissioning Fund Contribution Amount that the Licensee has calculated; and
	(b)	details of how that calculation has been made, having regard to the cost estimate set out in the Approved Offshore Decommissioning Programme, the Offshore Decommissioning Fund Investment Strategy and the expected operational life of the Offshore Transportation and Storage System.
3.6	provide the C Amour change	egulator may require further information to be ed in relation to how the Licensee has calculated Offshore Decommissioning Fund Contribution ht and may also require the Licensee to make es before approving the Offshore missioning Fund Contribution Amount.
3.7	conditi Comm Transp promp Contril Fund A	the Regulator has provided its approval under ion D3.5, in each Charging Year starting from [the ercial Operations Date of the Offshore portation and Storage System], the Licensee shall tly pay the Offshore Decommissioning Fund bution Amount into the Offshore Decommissioning account [by the end of each calendar month of that ng Year].
		the hierarchy and mechanics of payment to be peoped in the context of the financing workstream.
3.8		censee shall not make any withdrawals from the re Decommissioning Fund Account other than for:
	(a)	the purpose of investment in accordance with the Offshore Decommissioning Fund Investment Strategy; and
	(b)	with the approval of the Offshore Decommissioning Authority, for the purposes of carrying out any decommissioning of any part of the Offshore Transportation and Storage System (or related activities, as contemplated by condition D3.2(b)) as required by the Offshore Decommissioning Authority or any other regulator in accordance with the Approved Offshore Decommissioning Programme and the Offshore Decommissioning Requirements.
	fund v the ev	further provisions to be developed to deal with windfall/shortfall, as well as access to the fund in vent that the Licensee does not comply with its nmissioning obligations.

		3.9	The Offshore Decommissioning Fund Investment Strategy and the calculation of the Offshore Decommissioning Fund Contribution Amount shall be subject to review by the Regulator:
			(a) at each Periodic Review; and
			(b) outside of a Periodic Review if:
			 the Approved Offshore Decommissioning Programme has been updated and approved in accordance with the Offshore Decommissioning Requirements (including as a result of a periodic review of the Approved Offshore Decommissioning Programme by the Offshore Decommissioning Authority); or
			 (ii) either the Licensee or the Regulator considers that other circumstances mean that there may be a shortfall in the funds required to carry out decommissioning of the Offshore Transportation and Storage System (or related activities) in accordance with the Approved Offshore Decommissioning Programme.
		3.10	The Licensee shall notify the Regulator, as soon as reasonably practicable, upon the occurrence of any circumstances referenced in conditions D3.9(b)(i) or D3.9(b)(ii).
		3.11	Where there is a new Approved Offshore Decommissioning Programme in place in relation to the Offshore Transportation and Storage System or any part of it (for example, where the Offshore Transportation and Storage System has been significantly expanded or a new Storage Complex has been added to it), the Licensee may be required to establish a separate Offshore Decommissioning Fund in relation to that new Approved Offshore Decommissioning Programme, in accordance with this condition D3 (<i>Decommissioning</i> <i>obligations</i>).
4.	UK Emissions Trading Scheme	4.1	The Licensee shall comply with all obligations arising under the UK ETS in relation to all carbon dioxide that is delivered to the Offshore Transportation and Storage System.
5.	Metering	5.1	The Licensee shall be responsible for ensuring that appropriate metering equipment is installed, operated and maintained to accurately measure the rate of flow and composition of carbon dioxide being delivered by Users into the T&S Network.

		5.2	Without limiting the generality of condition C5.1, the Licensee shall ensure that the metering on the T&S Network complies with the requirements of:
			(a) UK ETS;
			(b) the NSTA;
			(c) the CCS Network Code; and
			(d) any other Legal Requirements,
			including but not limited to any requirements in respect of the location and specification of the metering and any requirements to identify CO_2 leakage on the Onshore Transportation System, Offshore Transportation and Storage System and any Storage Complex.
		5.3	The Regulator may require the Licensee to provide evidence of verification by an independent technical expert of compliance of the metering equipment with the relevant standards and levels of accuracy, as required by this condition.
			Note: metering obligations remain subject to further consideration. In particular, need to establish timeline for UK ETS permit which allows settlement of related issues under the licence, including design and costing of metering across the T&S Network.
6.	Obligations in relation to carbon dioxide storage	6.1	The Licensee shall ensure that any carbon dioxide that the Licensee accepts is contained within the T&S Network in accordance with the requirements of the licence and either:
			 (a) stored in a Storage Complex in relation to which the Licensee holds a Storage Licence and Storage Permit; or
			(b) delivered to an Onshore Transportation System or Offshore Transportation and Storage System operated by another CCS Licensee.
7.	Storage Licence and Storage Permit	7.1	Without limiting the generality of condition D2 (<i>Obligations relating to offshore transport of CO</i> ₂ and other obligations), the Licensee shall comply with the terms of the Licensee's Storage Licence and Storage Permit.

PART III: SPECIAL CONDITIONS

Section E: T&S Network Development

	_			
1.	T&S Network development	1.1	In the	e First Regulatory Period the Licensee shall:
			(a)	comply with its obligations under section F (<i>Construction Period Conditions</i>) in relation to the Handover Works;
			(b)	comply with its obligations under section G (<i>Commissioning Period Conditions</i>) in relation to the Commissioning Activities; and
			(c)	comply with the Approved Project Development Plan in relation to the development of the T&S Network.
2.	T&S Network expansion and additional	2.1		pproved Project Development Plan sets out (among things):
	construction,		(a)	the scope of the Project;
	commissioning and operational periods		(b)	the Approved T&S Network;
			(c)	Phase 1 activities to be completed prior to the Commercial Operations Date (i.e. the Handover Works and Commissioning Activities);
			(d)	Phase 2 activities to be completed post- Commercial Operations Date (i.e. activities with associated On-going Capex relating to the Approved T&S Network, which may include both construction and commissioning activities); and
			(e)	the Licensee's plans for the development and enhancement of the T&S Network beyond Phase 1 and Phase 2 (i.e. development activities in Phase 3 with associated On-going Devex).
		2.2	T&S N	icensee shall develop and enhance the Approved letwork through the application of On-going Devex vance Phase 3.
		2.3	the de Netwo the ne (for e APDP	Regulator and the Licensee each acknowledge that evelopment and enhancement of the Approved T&S ork as referred to in condition E2.2 may result in eed for the Licensee to propose Changes in Scope example, where On-going Devex envisaged in the reaches a stage where the Licensee can commence associated capex works), the impacts of which:
			(a)	shall be determined by the Regulator in accordance with this licence (see conditions F15 (<i>Change in Scope</i>), G12 (<i>Change in Scope</i>) and H24 (<i>Change in Scope</i>) below); and

		 (b) may result in additional investment decisions (which meet the requirements set out in the Project Programme and [section 5] of the APDP), construction periods, commissioning periods and commercial operations dates outside of the scope of the Phases, and, in such cases, the additional investment decisions, construction periods and commissioning periods shall
		neither:
		 (i) impact the Licensee's completion of the Handover Works, Commissioning Activities or the achievement of the Commercial Operations Date; nor
		(ii) impact the Licensee's ability to recover Allowed Revenue via Use of System Charges in respect of the Commercial Operations Date.
		2.4 Where the Regulator determines that a Change in Scope will require an additional construction period, commissioning period and commercial operations date, it is envisaged that these will follow the same processes and principles as set out in sections F and G (for example with the relevant costs during the construction and commissioning periods accruing to an SRAV before transferring to the RAV at the relevant commercial operations date, which is also the point that the Licensee's ability to begin to recover the relevant Allowed Revenue via Use of System Charges will be triggered).
3.	Definitions and interpretation	In the special conditions, terms defined in the standard conditions of this licence shall have the same meaning and, unless the context otherwise requires:
		"Additional or Expanded Storage Complex" has the meaning given in condition H20.9(a);
		"Availability" has the meaning given to it in condition H17.2;
		"Availability Correction Plan" has the meaning given to it in condition H17.14;
		"Availability Floor" means the floor in respect of Availability as set out in [section 3.3] of the APDP;
		"Availability Relief Factor" has the meaning given to it in condition H17.8;
		"Availability Target" means the target in respect of Availability as set out in [section 3.3] of the APDP;
		"Base Year" means [year to be specified];

"Carbon Storage and Development Plan" means the carbon storage development plan required to be prepared by the Licensee and submitted to the NSTA as part of the Licensee's application for a Storage Permit in relation to the relevant Storage Complex, in accordance with the requirements of the Storage of Carbon Dioxide (Licensing etc.) Regulations 2010; "Change in Law" means the coming into effect after Licence
Award of any change:
 (a) to any Legal Requirement which results in the Licensee being required to implement a new Legal Requirement; and/or
(b) to any Legal Requirement which applies to the Licensee and impacts on the capital or operational cost of the Project (including any change which results in such Legal Requirement ceasing to apply, being withdrawn or not being renewed); and/ or
(c) in the application, interpretation or effect of any Legal Requirement as the result of:
 (i) any judgment given by a court or tribunal of competent jurisdiction and in respect of which the period for making an appeal has expired; or
(ii) guidance published by the relevant body of competent jurisdiction,
which, in either case, applies (whether exclusively or otherwise) to the Licensee and/or the T&S Network and impacts on the capital or operational cost of the Project;
"COD Readiness" means that the Regulator has confirmed to the Licensee that the COD Readiness Activities have been completed pursuant to condition F4.3;
"COD Readiness Activities" has the meaning given to it in the APDP;
"Competent Auditor" has the meaning given to it in condition H20.6;
"Competent Person's Audit" has the meaning given to it in condition H20.2(a);
"Competent Person's Report" has the meaning given to it in condition H20.2(b);
"Constrained Registered Capacity" has the meaning given to it in condition H17.4(b);
"Construction Agreement" has the meaning given to it in the CCS Network Code;

"Corrective Measures" has the meaning given to it in The Storage of Carbon Dioxide (Licensing etc.) Regulations 2010;
"Corrective Measures Direction" means a direction in respect of Corrective Measures (and/or any measures for the protection of human health) given to the Licensee under The Storage of Carbon Dioxide (Licensing etc.) Regulations 2010;
" Corrective Measures Plan " has the meaning given to it in The Storage of Carbon Dioxide (Licensing etc.) Regulations 2010;
"Corrective Measures Requirements" has the meaning given to it in condition H19.2(b);
"CPR Year" has the meaning given in condition H20.3(a);
"Curtailed Nominations" has the meaning given to it in condition H17.4(a);
"Delay Notice" means a notice (or notices) served by Licensee and/or the First User under [clause 11.1] of the relevant Construction Agreement;
"Delay WACC" means an amount determined in accordance with conditions F13 (<i>WACC</i>) and G9 (<i>WACC</i>);
"Discriminatory Change in Law" means a Change in Law that specifically applies to:
 (a) the Project and not to the design, development, construction, completion, testing, commissioning, operation, maintenance and decommissioning of any other project;
(b) the T&S Network and not to any other network; or
(c) the Licensee and not to any other person;
"Event of First User Delay" has the meaning given to it in condition F5.2(a);
"First User" means any the User or Users which is/are needed supply the CO_2 required by the Licensee to carry out the Commissioning Activities;
"Force Majeure Event" means the occurrence of any of the following events after Licence Award:
(a) war, invasion, civil war, armed conflict, revolution, insurrection or encountering munitions of war;
(b) terrorism;
 (c) pressure waves caused by devices travelling at supersonic speeds;

	(d)	conta attrib	intering explosive materials, ionising radiation or mination by radioactivity except as may be utable to the Licensee's use of such explosives, cion or radioactivity;
	(e)	sourc Proje	ar, chemical or biological contamination unless the e or cause of the contamination is as a result of the ct or any act of the Licensee, its personnel/other byees or its subcontractors; or
	(f)	a Pub	lic Health Event,
	more	of its o	ly prevents the Licensee from performing one or bligations under the licence, if and to the extent that demonstrates to the Regulator that:
		(i)	such impediment is beyond its control;
		(ii)	such event was unforeseeable at Licence Award; and
		(iii)	the effects of the impediment could not have been avoided or overcome or otherwise mitigated by the Licensee;
			' means the confirmation by the Regulator of the of the bandover Works;
	["HSE	" mea	ns the Health and Safety Executive;]
	(i) ap Licens accord	prove see, to	ent Certifier" means a company or other person: d by the Regulator; and (ii) appointed by the act as an independent certifier to the Project in with an Independent Certifier's Deed of ;;
	deed	of app	ent Certifier's Deed of Appointment" means a pointment of an Independent Certifier in a form advance by the Regulator in writing;
	Persor	n's Rej	mpetent Person's Report" means a Competent port which was procured by the Licensee prior to rd following a Competent Person's Audit with:
	(a)	requi	audit carried out in accordance with the rements set out in condition H20 (<i>Store</i> rmance) for the conduct of a Competent Person's ; and
	(b)	set o	report produced in accordance with the requirements ut in condition H20 (<i>Store performance</i>) for the action of (or updating of) a Competent Person's t;
	Sched	luled C	Date" means the date which is 12 months following COD, as extended in accordance with condition G2 to achieve System Acceptance);

"Operational Period" has the meaning given to it in condition H1.1;
["OPRED" means the Offshore Petroleum Regulator for Environment and Decommissioning, which is a part of the Department for Energy Security and Net Zero;]
"Phase 3" means the development activities relating to the T&S Network with associated On-going Devex included in the APDP, as further described in [section 5] of the APDP;
"Phases" means Phase 1, Phase 2 and Phase 3, with each being a "Phase";
"Planned Initial Users" means the Users designated as such in [section4] of the APDP;
"Post-COD WACC" means the WACC which applies on and after COD as set out in conditions G9 (<i>WACC</i>) and H8 (<i>WACC</i>);
"Post Commissioning Review" has the meaning given to it in condition H2.2;
"Post Construction Review" has the meaning given in condition G10.2;
"Preceding Competent Person's Report" has the meaning given in condition H20.3(b);
"Pre-COD WACC" means the WACC which applies prior to COD, subject as set out in conditions G9 (<i>WACC</i>) and F13 (<i>WACC</i>);
" Pre-Licence Award Devex " means the development costs properly and efficiently incurred by the Licensee in relation to the T&S Business prior to Licence Award, being the amount set out in the Financial Settlement Schedule;
"Pre-operations Difference Payment" has the meaning given to it in the RSA;
"Project Programme" means the programme for the Project set out in the Technical Details Schedule;
"Proposed Revised Scheduled COD" has the meaning given to it in condition G2.5;
"Public Health Event" means any epidemic, pandemic, or other national, regional or localised public health event in each case which is recognised after Licence Award as subsisting by the World Health Organization, including without limitation any localised or widespread occurrence of an infectious disease caused by:
(a) a pathogen, whether bacterial or viral; or
(b) any other biological or natural agent,

which directly and adversely affects the operations of the T&S Network by the Licensee;					
"Qualifying	Change in Law" means an unforeseeable:				
(a) Discrii	minatory Change in Law; and				
(b) Specif	ic Change in Law;				
	nt" means the occurrence of any of the following Licence Award:				
	by persons other than the Licensee's personnel and employees of the Licensee and its subcontractors);				
	al catastrophe such as earthquake, tsunami, nic activity, hurricane or typhoon; and				
the Li	age and acts of vandalism (by persons other than icensee's personnel and other employees of the see and its subcontractors),				
Acceptance D	y prevents the Licensee from achieving the System Date by the Scheduled COD, if and to the extent that demonstrates to the Regulator that:				
(i)	such impediment is beyond its control;				
(ii)	such event was unforeseeable at Licence Award; and				
(iii)	the effects of the impediment could not have been avoided or overcome or otherwise mitigated by the Licensee;				
"Remediation H19.1;	on Plan" has the meaning given to it in condition				
"Return Dur in condition (ring Commissioning" has the meaning given to it G8.1;				
"Return Dur condition F12	ring Construction" has the meaning given to it in 2.1;				
"Revenue S support agre	<pre>upport Agreement" or "RSA" means the revenue ement;</pre>				
Operations D Technical De with conditio F17 (<i>Force</i> <i>achieve Sys</i> t	COD" means the date on which the Commercial Date is scheduled to be achieved as set out in the tails Schedule, as may be extended in accordance ons F15 (<i>Change in Scope</i>), F16 (<i>Change in Law</i>), <i>Majeure</i>), F18 (<i>Relief Event</i>), G2 (<i>Obligation to tem Acceptance</i>), G12 (<i>Change in Scope</i>), G13 aw), G14 (<i>Force Majeure</i>) or G15 (<i>Relief Event</i>);				
	Handover Date" means the date set out in the tails Schedule;				

"Specific Change in Law" means a Change in Law that specifically applies to:
 (a) the provision of services the same or similar to those provided by the Licensee across the T&S Network; or
 (b) the holding of shares in a company whose main business is providing services the same or similar to those provided by the Licensee across the T&S Network;
"SRAV Calculation Period" means a [<i>period to be specified</i>] within a Charging Year;
"SRMS Guidelines" means the Guidelines for the Applications of the CO ₂ Storage Resources Management System dated July 2022 and issued by the Society of Petroleum Engineers;
"Storage Site" has the meaning given to it in the Carbon Dioxide Storage Regulations;
Note: checks being run on definition in context of use.
"System Charging Platform" has the meaning given to it in condition H28.1(b);
"System Charging Statement" has the meaning given to it in condition H29.1;
"Technical Details Schedule" means Schedule 4 to this licence;
"Use of System Charges" means:
 (a) Flow Charges, which shall comprise (i) an Onshore Flow Charge; and (ii) an Offshore Flow Charge;
(b) Capacity Charges, which shall comprise (i) an Onshore Capacity Charge; and (ii) an Offshore Capacity Charge; and
 (c) Network Charges, which shall comprise (i) an Onshore Network Charge; and (ii) an Offshore Network Charge; and
Note: It is intended that, initially at least, no separate connection charges will be payable by Users.
"Use of System Charging Methodology" has the meaning given to it in condition H28 (<i>Use of System Charging Methodology</i>).
Unless the context otherwise requires, "carbon dioxide" (or "CO ₂ ") means a stream consisting primarily of carbon dioxide molecules and references to carbon dioxide being delivered by a User to the T&S Network assume that the stream being delivered is carbon dioxide.

Section F: Construction Period Conditions

1.	Construction Period	1.1	The Construction Period is the period from Licence Award to the Handover Date.
			Note : the Handover Date is the date on which the Regulator confirms that the Handover Works have been completed and the first day of the Commissioning Period.
		1.2	The Construction Period is within the First Regulatory Period.
2.	Obligation to	2.1	During the Construction Period, the Licensee shall:
	achieve Handover		(a) design, finance and construct the Handover Works;
			(b) in doing so, use reasonable endeavours to achieve completion of the Handover Works by the Scheduled Handover Date.
			Note : it is recognised that phasing of construction/commissioning activities for the works envisaged by the Approved Project Development Plan may be required depending on the Licensee's proposed technical solution. Tailoring of the licence may therefore be required for the inclusion of phasing mechanics, where relevant.
		2.2	The Licensee shall provide quarterly updates to the Regulator in each Charging Year during the Construction Period in the form agreed with the Regulator, setting out the Licensee's progress in relation to the Project Programme.
		2.3	As soon as reasonably practicable after the Licensee becomes aware that it is reasonably likely to fail to achieve Handover by the Scheduled Handover Date, it shall notify the Regulator in writing of such probable failure.
		2.4	These licence conditions for the Construction Period shall continue to apply until the Handover Date is achieved.
3.	Handover Date	3.1	The Handover Date shall be the date when Handover is achieved, as confirmed by the Regulator pursuant to this condition F3 (<i>Handover Date</i>).
		3.2	The Licensee shall give the Regulator no less than 25 Business Days' notice in writing of the date that it anticipates the Handover Works will be completed, and, in the event that such date changes, the Licensee shall as soon as reasonably practicable give the Regulator notice in writing of the revised date of anticipated completion.

		3.3	The Licensee shall provide to the Regulator eviden from an Independent Certifier in the form of a certifica certifying that the Independent Certifier is satisfied th the Handover Works have been completed accordance with this licence and the relevant Leg Requirements, together with any further evidence th the Regulator has notified the Licensee that it requir in respect of the completion of the Handover Works.			
		3.4	Within 20 Business Days of receipt, the Regulator s consider the evidence provided by the Licensee respect of the completion of the Handover Works shall:			
			(a)	confirm that the Handover Works are complete and Handover has been achieved; or		
			(b)	confirm that the Handover Works are not complete; or		
			(c)	require further evidence from the Licensee to confirm whether the Handover Works have been completed.		
		3.5	Where	the Regulator:		
			(a)	confirms under condition F3.4(b) that the Handover Works are not complete, the Licensee shall refer the matter back to the Independent Certifier and the procedures in conditions F3.3 and F3.4 shall be repeated; or		
			(b)	requires further evidence from the Licensee under condition F3.4(c), the Licensee shall provide such evidence to the Regulator and the procedure in condition F3.4 shall be repeated,		
				ch case until the Regulator confirms that the over Works are complete and Handover has been ved.		
4.	COD Readiness	4.1	the per- Licens Indeper- certify the C togeth has no	e the Licensee wishes to achieve COD Readiness for urposes of condition F5 (<i>First User Delay</i>), the ee shall provide to the Regulator evidence from an endent Certifier in the form of a certificate ing that the Independent Certifier is satisfied that OD Readiness Activities have been completed, her with any further evidence that the Regulator obtified the Licensee that it requires in respect of the etion of the COD Readiness Activities.		
		4.2	consid	[20] Business Days of receipt, the Regulator shall er the evidence provided in respect of the etion of the COD Readiness Activities and shall:		
			(a)	confirm:		

				(i)	that the COD Readiness Activities are complete;
				(ii)	that COD Readiness has been achieved; and
				(iii)	the date on which COD Readiness was achieved, being the date on which the Licensee provided all evidence in respect of the completion of the COD Readiness Activities required by the Regulator in accordance with this condition F4; or
			(b)		m that the COD Readiness Activities are not lete; or
			(c)	confir	re further evidence from the Licensee to m whether the COD Readiness Activities been completed.
		4.3	Where	e the Re	egulator:
			(a)	Readi Licens Indep	ms under condition F4.2(b) that the COD ness Activities are not complete, the see shall refer the matter back to the pendent Certifier and the procedures in tions F4.2 and F4.3 shall be repeated; or
			(b)	under provid	res further evidence from the Licensee condition F4.2(c), the Licensee shall de such evidence to the Regulator and the dure in condition F4.2 shall be repeated,
			that th	ne COD	until the Regulator confirms to the Licensee Readiness Activities are complete and COD is been achieved.
5.	First User Delay	5.1	Where	e the Li	censee:
			(a)	has a	chieved COD Readiness; and
			(b)	is cau	nably believes that one or more First User using, or shall cause, delay to COD beyond cheduled COD, the Licensee shall:
				(i)	provide to the Regulator a copy of the relevant Delay Notice(s); and
				(ii)	use reasonable endeavours to mitigate the impact of the action or inaction of such First User(s).
		5.2	Notico consio detail	es) fror dered t ed in th	ceipt of a copy of a Delay Notice (or Delay m the Licensee, the Regulator shall, having the circumstances of the relevant delay as the Delay Notice(s) provided by the Licensee tion F5.1(b) above:

		r		
			(a)	confirm that it is satisfied that the delay is, or was, caused by one or more First User(s), in which case, the Regulator shall confirm that an "Event of First User Delay" has occurred, the date on which such Event of First User Delay occurred, and the period of delay caused by the Event of First User Delay; or
			(b)	confirm that the delay is, or was, not caused by one or more First User(s); or
			(c)	require further evidence from the Licensee to confirm whether the delay is, or was, caused by one or more First User(s).
		5.3	the Li provid	e the Regulator requires further information from icensee under condition F5.2(c), the Licensee shall de such evidence to the Regulator and the dure in condition F5.2 shall be repeated.
6.	SRAV build-up	6.1	During	g the Construction Period:
			(a)	the Re-use Assets Valuation (if any);
			(b)	the SRAV Capex as referenced in condition F8 (SRAV Capex (Handover Works));
			(c)	the Pre-Licence Award Devex;
			(d)	the Opex;
			(e)	the On-going Devex as referenced in condition F11 (<i>On-going Devex</i>);
			(f)	[Pass Through Costs];
			(g)	the Return During Construction; and
			(h)	the licence fees payable in accordance with condition A2 (<i>Payments by the Licensee to the Regulator</i>),
				ccrue to an SRAV in accordance with the provisions s section F.
				<i>:</i> details of Pass Through Costs to be included in creatment paper.
		6.2	in ao <i>const</i> Perioo	, including Return During Construction calculated ccordance with condition F13 (<i>Return during ruction</i>), will build up over an SRAV Calculation d and will be moved on to the RAV in accordance condition G10 (<i>Post Construction Review</i>).
		6.3	the ca	e the Construction Period includes part of a year, alculation for that part of a Charging Year shall be tioned.

		6.4 SRAV will be measured and maintained in real CPIH terms, with reference to the Base Year. Where required, additions to SRAV will be adjusted to align with this price basis.							
7.	Re-use Assets	Note: it is recognised that certain oil and gas assets could be purchased and repurposed for T&S purposes (referred to here as the "Re-use Assets"), as a cost-saving mechanism by comparison to building completely new assets. The process for the valuation of any Re-use Assets is being developed separately.							
8.	SRAV Capex (Handover Works)	Note: T&SCo to refer to separate cost treatment paper.							
9.	Opex	Note: T&SCo to refer to separate cost treatment paper.							
10.	Pre-Licence Award Devex	10.1 The Pre-Licence Award Devex will be added to the SRAV at Licence Award at:							
		$Value_{nominal} \ x \ CPIH_{BaseYear} \ divided \ by \ CPIH_{LicenceAwardDate},$							
		where:							
		Value _{nominal} = the value in a nominal price base or in prices of a year other than the Base Year;							
		CPIH _{BaseYear} = the average value of each of the 12 monthly values of CPIH _m from 1 April to 31 March within the Base Year; and							
		$CPIH_{LicenceAwardDate}$ = the CPIH index at Licence Award.							
		Note:							
		• T&SCo to refer to separate guidance for cost assessment, which includes assessment of Pre-Licence Award Devex.							
		• Agreed Pre-Licence Award Devex to be included in the Financial Settlement Schedule.							
11.	On-going Devex	Note: T&SCo to refer to separate cost treatment paper.							
12.	Return During Construction	12.1 Return During Construction is an allowance for interest charges on the financing of and other costs of capital in relation to the Handover Works.							
		Note: position regarding specific financing costs such as commitment fees, arranging fees, hedging costs, hedge margin, etc., to be reviewed as part of the ongoing finance workstream.							

		12.2	The Return During Construction will be calculated as follows:
			For any SRAV Calculation Period, Return During
			Construction = WACC x SRAV
			where:
			WACC = an amount determined in accordance with condition F13 (WACC); and
			SRAV = the opening balance of the SRAV (calculated in accordance with condition F6 (SRAV build-up)) for that SRAV Calculation Period.
		Note: settled	to be updated once the calculation methodology has been
13.	WACC	13.1	Subject to condition F13.2:
			(a) WACC for the period up to (but excluding) the COD is the Pre-COD WACC, being the amount set out in the Financial Settlement Schedule for the Pre-COD WACC; and
			(b) WACC for the period from (and including) the COD is the Post-COD WACC, being the amount set out in the Financial Settlement Schedule for the Post-COD WACC,
			and any references to WACC in this section F (<i>Construction Period Conditions</i>) shall be construed accordingly.
		13.2	Delay WACC as set out in the Financial Settlement Schedule will apply as set out in conditions G9.1 to G9.2.
14.	Indexation	14.1	During the Construction Period, the SRAV will be measured and maintained in real CPIH terms, with reference to the Base Year.
		14.2	Where variable values need to be deflated from a nominal price base to Base Year prices, the following formula will be used:
			$Value_{BaseYear\ prices}$ = $Value_{nominal}\ x\ CPIH_{BaseYear}\ divided\ by\ CPIH_m$
			where:
			Value _{BaseYear prices} = the deflated/restated value in Base Year prices;
			Value _{nominal} = the value in a nominal price base or in prices of a Charging Year other than the Base Year;

			CPIH	BaseYear	 the average value of each of the 12 monthly values of CPIH_m from 1 April to 31 March within the Base Year; and
			CPIH	l _m	 the value of CPIH for the month m.
15.	Change in Scope	15.1	becom Const of suc its an	nes aw ruction h Char halysis	e shall notify the Regulator as soon as it vare of any Change in Scope during the Period and shall provide details in respect age in Scope to the Regulator together with of the effects, or likely effects, of such cope, including but not limited to:
			(a)	-	adjustments to the Revenue Calculations as ult of the Change in Scope, including without ation:
				(i)	any reasonable increases or decreases to any ex ante allowance (with cost sharing) (including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of SRAV Capex, Opex, On-going Capex or On-going Devex); and
				(ii)	the impact on any ex post assessments of the economic and efficient expenditure by the Licensee;
			(b)	obliga exten is de	adjustment in respect of the Licensee's ations under this licence where and to the at that the Licensee is unable to perform or layed in performing such obligations as a t of the Change in Scope:
				(i)	for such period as is necessary having regard to the Change in Scope; or
				(ii)	until such time as the obligations have been amended to reflect the impact of the Change in Scope;
			(c)	asses again specia	djustments to the targets or outputs (or the sment of the performance of the T&S Assets st such targets or outputs) under these al conditions to reflect the impact of the ge in Scope;
			(d)	Schee	adjustments to key milestone dates ding the Scheduled Handover Date, the duled COD and/or the Longstop Date) to account of the impact of the Change in e;

			(e)	the inclusion of new construction periods or operational periods (and associated SRAVs) in respect of the Change in Scope (see condition E2 (T&S Network expansion and additional construction, commissioning and operational periods) above); and/or
			(f)	any adjustments to the WACC for the Construction Period as determined in accordance with condition F13 (<i>WACC</i>).
		15.2		Regulator will determine the impact on the Project any adjustments to be provided to the Licensee.
		15.3	F15.1 additi appro	owing notification by the Licensee under condition 1, the Licensee shall provide to the Regulator any tional information that the Regulator considers ropriate and requests from the Licensee in respect uch Change in Scope.
16.	Change in Law	16.1	becom poten Const of suc Chang of the	Licensee shall notify the Regulator as soon as it mes aware of any Qualifying Change in Law, or ntial Qualifying Change in Law, during the truction Period and shall provide details in respect ch Qualifying Change in Law or potential Qualifying age in Law to the Regulator together with its analysis e effects, or likely effects, of such Qualifying Change w including but not limited to:
			(a)	any adjustments to key milestone dates (including the Scheduled Handover Date, the Scheduled COD and/or the Longstop Date) to take account of the impact of the Qualifying Change in Law;
			(b)	any adjustment in respect of the Licensee's obligations under this licence where and to the extent that the Licensee is unable to perform or is delayed in performing such obligations as a result of the Qualifying Change in Law:
				 for such period as is necessary having regard to the Qualifying Change in Law; or
				 (ii) until such time as the obligations have been amended to reflect the impact of the Qualifying Change in Law;
			(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Qualifying Change in Law;
			(d)	any adjustments to the Revenue Calculations as a result of the Qualifying Change in Law, including without limitation:

			 (i) any reasonable increases or decreases to any ex ante allowance (with cost sharing) (including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of SRAV Capex, Opex, On-going Capex or On-going Devex); and
			(ii) any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Qualifying Changes in Law; and/or
			(e) any adjustments to the WACC for the Construction Period as determined in accordance with condition F13 (WACC).
		16.2	The Regulator will determine the impact on the Project and any adjustments to be provided to the Licensee to reflect such impact.
		16.3	Following notification by the Licensee under condition F16.1, the Licensee shall provide to the Regulator:
			(a) any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Qualifying Change in Law or potential Qualifying Change in Law; and
			(b) where applicable in the context of an increase or decrease of the Licensee's costs pursuant to condition F16.2, satisfactory evidence to the Regulator that the Qualifying Change in Law affects construction and/or the cost impact to On-going Devex, Capex or Opex.
		16.4	The amount of any increases to the Licensee's costs pursuant to condition F16.2 shall be subject to the Licensee providing satisfactory evidence that the Qualifying Change in Law affects construction and/or the cost impact to On-going Devex, Capex or Opex pursuant to condition F16.3.
17.	Force Majeure	17.1	The Licensee shall notify the Regulator as soon as it becomes aware of any Force Majeure Event, or potential Force Majeure Event, during the Construction Period, and shall provide details in respect of such Force Majeure Event or potential Force Majeure Event to the Regulator together with its analysis of the effects, or likely effects, of such Force Majeure Event including but not limited to:
			(a) any adjustments to key milestone dates (including the Scheduled Handover Date, the Scheduled COD and/or the Longstop Date) to

		take account of the impact of the Force Majeure Event;
	(b)	any adjustment in respect of the Licensee's obligations under this licence where and to the extent that the Licensee is unable to perform or is delayed in performing such obligations as a result of the Force Majeure Event:
		(i) for such period as is necessary having regard to the Force Majeure Event; or
		 (ii) until such time as the obligations have been amended to reflect the impact of the Force Majeure Event;
	(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Force Majeure Event;
	(d)	any adjustments to the Revenue Calculations as a result of the Force Majeure Event, including without limitation:
		 any reasonable increases or decreases to any ex ante allowance (with cost sharing) (including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of SRAV Capex, Opex, On-going Capex or On-going Devex); and
		 (ii) any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Force Majeure Event; and/or
	(e)	any adjustments to the WACC for the Construction Period as determined in accordance with condition F13 (<i>WACC</i>).
17.2	and a	egulator will determine the impact on the Project by adjustments to be provided to the Licensee to such impact.
		ng notification by the Licensee under condition the Licensee shall provide to the Regulator:
	(a)	any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Force Majeure Event or potential Force Majeure Event; and

			(b) where applicable in the context of an increase or decrease of the Licensee's costs pursuant to condition F17.2, satisfactory evidence to the Regulator that the Force Majeure Event affects construction and/or the cost impact to On-going Devex, Capex or Opex.
		17.4	The amount of any increases to the Licensee's costs pursuant to condition F17.2 shall be subject to the Licensee providing satisfactory evidence that the Force Majeure Event affects construction and/or the cost impact to On-going Devex, Capex or Opex pursuant to condition F17.3.
18.	Relief Event	18.1	The Licensee shall notify the Regulator as soon as it becomes aware of any Relief Event, or potential Relief Event, during the Construction Period, and shall provide details in respect of such Relief Event or potential Relief Event to the Regulator together with its analysis of the effects, or likely effects, of such Relief Event on the Scheduled COD.
		18.2	The Regulator will determine the impact on the Scheduled COD and any extensions to be provided to the Scheduled COD to reflect such impact.
		18.3	Following notification by the Licensee under condition F18.1, the Licensee shall provide to the Regulator any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Relief Event or potential Relief Event.
		draftin	Relief Event mechanism to be amended in full form g to so that it will operate to extend the Handover Date ongstop Date (as applicable), as well as the Scheduled

Section G: Commissioning Period Conditions

1.	Commissioning Period	1.1	The Commissioning Period is the period from (and including) the Handover Date to the Commercial Operations Date.
			Note: the Commercial Operations Date will be when the T&S Network has passed all its commissioning tests, marking the end of the Commissioning Activities and the start of the Operational Period.
		1.2	The Commissioning Period is within the First Regulatory Period.
2.	Obligation to achieve System Acceptance	2.1	During the Commissioning Period, the Licensee shall:
			(a) complete the Commissioning Activities; and
			(b) in doing so, use reasonable endeavours to achieve System Acceptance by the Scheduled COD.
		2.2	The Licensee shall achieve System Acceptance by the Longstop Date.
		2.3	The Licensee shall provide quarterly updates to the Regulator in each Charging Year during the Commissioning Period in the form agreed with the Regulator, setting out the Licensee's progress in relation to the Project Programme.
		2.4	As soon as reasonably practicable after the Licensee becomes aware that it is reasonably likely to fail to achieve (i) the Scheduled COD; and/or (ii) System Acceptance by the Longstop Date:
			 (a) it shall notify the Regulator in writing of such probable failure; and
			(b) these licence conditions for the Commissioning Period shall continue to apply until the Commercial Operations Date.
		2.5	Where the Regulator has confirmed an Event of First User Delay pursuant to condition F5.2(a), promptly following a notification by the Licensee under condition G2.4 regarding a consequent delay to the Scheduled COD, the Licensee shall calculate the date on which the Licensee anticipates the delayed Commercial Operations Date shall occur (a "Proposed Revised Scheduled COD").

	2.6			ating a Proposed Revised Scheduled ensee shall:
		(a)		easonably and in accordance with Good stry Practice;
		(b)	incluc the ir	into account all relevant circumstances, ding the need to mitigate delay and/or mpact of delay on the Commissioning ties; and
		(c)	act i mann	n a transparent and proportionate her.
	2.7	Where	e condit	tions G2.4, G2.5 and G2.6 apply:
		(a)	Regul Scheo Regul that, Revis	Licensee shall promptly notify the lator of the relevant Proposed Revised duled COD and shall provide to the lator upon request satisfactory evidence when calculating such Proposed ed Scheduled COD, the Licensee lied with the requirements of condition
		(b)	condi consid Sched provid	ving receipt of the notice referred to in tion G2.7(a), the Regulator shall der the relevant Proposed Revised duled COD, together with any evidence ded by the Licensee in respect of the lation of such date and shall:
			(i)	confirm that such Proposed Revised Scheduled COD is accepted and, for the purposes of these licence conditions, shall apply by way of extension to the Scheduled COD; or
			(ii)	confirm that such Proposed Revised Scheduled COD is not accepted; or
			(iii)	require further evidence from the Licensee in respect of such Proposed Revised Scheduled COD and its calculation; and
		(c)	where	e the Regulator:
			(i)	confirms under condition G2.7(b)(ii) that a Proposed Revised Scheduled COD is not accepted, the Licensee shall revisit its calculation of the relevant Proposed Revised Scheduled COD and the procedures in conditions G2.7(a) and G2.7(b) shall be repeated; or

		 (ii) requires further evidence from the Licensee under condition G2.7(b)(iii), the Licensee shall provide such evidence to the Regulator and the procedure in condition G2.7(b) shall be repeated,
		in each case until the Regulator confirms that a Proposed Revised Scheduled COD is accepted and, for the purposes of these licence conditions, shall apply by way of extension to the Scheduled COD.
2.8	condit achiev and/o	e the Licensee (i) notifies the Regulator under tion G2.4 that it is reasonably likely to fail to ve System Acceptance by the Longstop Date; r (ii) fails to achieve System Acceptance by ongstop Date:
	(a)	the Licensee must provide a remedial plan to the Regulator, as soon as reasonably practicable, and, unless otherwise agreed by the Regulator, in any event within 20 Business Days of the date of (i) the notice; or (ii) failure (as the case may be), setting out the steps which the Licensee intends to take to address the cause(s) of delay to System Acceptance and to achieve System Acceptance;
	(b)	the Regulator shall be entitled to raise comments on any proposed remedial plan and the Licensee shall take account of such comments and, where appropriate, the Licensee shall issue a revised remedial plan to the Regulator as soon as reasonably practicable;
	(c)	the Regulator shall review the revised remedial plan and either confirm that its comments have been addressed or identify any of its comments which have not been addressed;
	(d)	if the Regulator identifies pursuant to condition G2.8(c) that any of its comments have not been addressed, the Licensee shall issue a further revised remedial plan which addresses such comments and/or shall set out in writing an explanation as to why it has not addressed any such comments; and
	(e)	the Licensee must use reasonable endeavours to implement and comply with any such remedial plan (as may be revised in accordance with conditions G2.8(b) to 2.8(d)) so as to achieve System Acceptance

				-	he date proposed in the relevant dial plan.
		2.9	additio Syster (as re 2.8(d) maxin Date	onal tin m Acce vised i)), pro num of	op Date will be extended by such me as the Licensee requires to achieve eptance as set out in its remedial plan n accordance with conditions G2.8(b) to ovided that the extension shall be a f 12 months from the original Longstop ut any extension pursuant to this 2.9.
3.	System Acceptance/Commercial Operations Date	3.1	when the by the	Systen e Reg	rcial Operations Date shall be the date n Acceptance is achieved, as confirmed ulator pursuant to this condition G3 eptance/Commercial Operations Date).
		3.2	25 Bus it antio be con change practio	siness cipates mplete es, the cable g	e shall give the Regulator no less than Days' notice in writing of the date that that the Commissioning Activities will d, and, in the event that such date ticensee shall as soon as reasonably ive the Regulator notice in writing of the of anticipated completion.
		3.3	evider of a Certifi Activit furthe Licens	nce fro certific ier is ties ha er evide see tha	ee shall provide to the Regulator m an Independent Certifier in the form cate certifying that the Independent satisfied that the Commissioning ve been completed, together with any ence that the Regulator has notified the t it requires in respect of the completion hissioning Activities.
		3.4	shall c	conside	usiness Days of receipt, the Regulator er the evidence provided in respect of on of the Commissioning Activities and
			(a)	confir	m:
				(i)	that the Commissioning Activities are complete;
				(ii)	that System Acceptance has been achieved; and
				(iii)	the date on which System Acceptance was achieved, being the date on which the Licensee provided all evidence in respect of the completion of the Commissioning Activities required by the Regulator in accordance with this condition G3;
			(b)		m that the Commissioning Activities ot complete; or

			(c)	require further evidence from the Licensee to confirm whether the Commissioning
				Activities have been completed.
		3.5	Where	e the Regulator:
			(a)	confirms under condition G3.4(b) that the Commissioning Activities are not complete, the Licensee shall refer the matter back to the Independent Certifier and the procedures in conditions G3.2 to G3.4 shall be repeated; or
			(b)	requires further evidence from the Licensee under condition G3.4(c), the Licensee shall provide such evidence to the Regulator and the procedure in condition G3.4 shall be repeated,
			Comn	ch case until the Regulator confirms that the nissioning Activities are complete and System stance has been achieved.
4.	SRAV build-up	4.1	During	g the Commissioning Period:
			(a)	the SRAV Capex as referenced in condition G5 (SRAV Capex (Commissioning Activities));
			(b)	the Opex;
			(c)	any On-going Devex as referenced in condition G7 (<i>On-going Devex</i>);
			(d)	[Pass Through Costs];
			(e)	the Return During Commissioning; and
			(f)	the licence fees payable in accordance with condition A2 (<i>Payments by the Licensee to the Regulator</i>),
			provis	accrue to an SRAV in accordance with the sions of this section G (<i>Commissioning Period tions</i>).
				details of Pass Through Costs to be included to the treatment paper.
		4.2	calcul <i>during</i> Calcu in a	, including Return During Commissioning ated in accordance with condition G8 (<i>Return</i> <i>g Commissioning</i>), will build up over an SRAV lation Period and will be moved on to the RAV accordance with condition H2 (<i>Post</i> <i>missioning Review</i>).

			1			
		4.3	Where the Commissioning Period includes part of a year, the calculation for that part of a Charging Year shall be apportioned.			
		4.4 SRAV will be measured and maintained in real CP terms, with reference to the Base Year. Whe required, additions to SRAV will be adjusted to all with this price basis.				
		Note: the working assumption is that delivery of CO_2 by First Users will be free of charge.				
5.	SRAV Capex (Commissioning Activities)	Note: T&SCo to refer to separate cost treatment paper.				
6.	Opex	Note: T&SCo to refer to separate cost treatment paper.				
7.	On-going Devex	Note: T&SCo to refer to separate cost treatment paper.				
8.	Return During Commissioning	8.1	Return During Commissioning is an allowance for interest charges on the financing of and other costs of capital in relation to the Commissioning Activities.			
		8.2 The Return During Commissioning will be calculat as follows:				
		For any SRAV Calculation Period, Return Dur Commissioning = WACC x SRAV				
			where:			
			WACC = an amount determined in accordance with condition G9 (WACC); and			
			SRAV = the opening balance of the SRAV for that SRAV Calculation Period.			
		8.3 If any Pre-operations Difference Payment become payable under the RSA, then the Return Durin Commissioning shall be reduced by reference to th amount of the Pre-operations Difference Paymen that relates to the allowed cost of debt.				
		Note: been se	to be updated once the calculation methodology has ettled.			
9.	WACC	9.1	Subject to condition G9.2:			
			(a) WACC for the period up to (but excluding) the COD is the Pre-COD WACC, being the amount set out in the Financial Settlement Schedule for the Pre-COD WACC; and			
			(b) WACC for the period from (and including) the COD is the Post-COD WACC, being the			

			amount set out in the Financial Settlement Schedule for the Post-COD WACC,
			and any references to WACC in this section G shall be construed accordingly.
		9.2	Where COD is not achieved by the Scheduled COD, Delay WACC shall apply from the Scheduled COD until COD, but will not apply to the extent Scheduled COD is subsequently extended and such extended Scheduled COD has not yet been reached and only from the date of the event which gives rise to the extended Scheduled COD.
10.	Post Construction Review	10.1	The Licensee shall provide the Regulator with all such information as the Regulator may require to enable it to carry out the Post Construction Review and other activities under this condition G10 (<i>Post Construction Review</i>), including:
			 (a) forecasted figures for the Construction Period, which shall be provided no later than [●] Business Days prior to the anticipated Handover Date; and
			 (b) actual figures for the Construction Period, which shall be provided no later than [●] Business Days following the Handover Date.
		10.2	Following the Handover Date, the Regulator shall undertake a review of the Construction Period (a "Post Construction Review"), which shall include a preliminary reconciliation of the SRAV for the Construction Period. This preliminary reconciliation will be based on forecasted figures for the Construction Period provided by the Licensee under condition G10.1(a).
		10.3	On the Commercial Operations Date, the reconciled SRAV (as referred to in condition G10.2 above), including the rolled-up Return During Construction, will be moved to the RAV.
		10.4	Adjustments for a final reconciliation based on the actual figures for the Construction Period provided by the Licensee under condition G10.1(b) will be applied to the calculation of Allowed Revenue and RAV as part of the first annual iteration process (referenced in condition H16.1(b)).
11.	Indexation	11.1	During the Commissioning Period, the SRAV will be measured and maintained in real CPIH terms, with reference to the Base Year.
		11.2	Where variable values need to be deflated from a nominal price base to Base Year prices, the following formula will be used:

			Value _E by CP	•	rices =	= Value _{nominal} x CPIH _{BaseYear} divided
			where	:		
			Value	BaseYear	=	the deflated/restated value in Base Year prices;
			Value	enominal		the value in a nominal price base or in prices of a Charging Year other than the Base Year;
			СРІН	BaseYear	=	the average value of each of the 12 monthly values of $CPIH_m$ from 1 April to 31 March within the Base Year; and
			CPIH	m	=	the value of CPIH for the month m.
12.	Change in Scope	12.1	becom Comm respec togeth	nes awa nissionin ct of su ner with s, of su	are of ng Pe ich C n its	notify the Regulator as soon as it f any Change in Scope during the eriod and shall provide details in hange in Scope to the Regulator analysis of the effects, or likely hange in Scope, including but not
			(a)	as a r	esult	ments to the Revenue Calculations of the Change in Scope, including hitation:
				(i)	(wit chai (ii) of	reasonable increases or reases to any ex ante allowance h cost sharing) (including: (i) any nge to the cost sharing factor; and any ex ante allowance in respect SRAV Capex, Opex, On-going ex or On-going Devex); and
				(ii)		impact on any ex post essments of the economic and cient expenditure by the Licensee;
			(b)	obliga the e perfor	tions xtent m o tions	ment in respect of the Licensee's under this licence where and to that the Licensee is unable to r is delayed in performing such as a result of the Change in
				(i)		such period as is necessary having ard to the Change in Scope; or
				(ii)	bee	l such time as the obligations have n amended to reflect the impact of Change in Scope;

			(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Change in Scope;
			(d)	any adjustments to key milestone dates (including the Scheduled COD and/or the Longstop Date) to take account of the impact of the Change in Scope;
			(e)	the inclusion of new construction periods or operational periods (and associated SRAVs) in respect of the Change in Scope (see condition E2 (<i>T&S Network expansion and</i> <i>additional construction, commissioning and</i> <i>operational periods</i>) above); and/or
			(f)	any adjustments to the WACC for the Commissioning Period as determined in accordance with condition G9 (<i>WACC</i>).
		12.2		egulator will determine the impact on the tand any adjustments to be provided to the ee.
		12.3	condit Regula Regula	ing notification by the Licensee under ion G12.1, the Licensee shall provide to the ator any additional information that the ator considers appropriate and requests from censee in respect of such Change in Scope.
13.	Change in Law	13.1	becom potent Comm respec potent togeth effects	censee shall notify the Regulator as soon as it hes aware of any Qualifying Change in Law, or cial Qualifying Change in Law, during the dissioning Period and shall provide details in ct of such Qualifying Change in Law or cial Qualifying Change in Law to the Regulator her with its analysis of the effects, or likely s, of such Qualifying Change in Law including bt limited to:
			(a)	any adjustments to key milestone dates (including the Scheduled COD and/or the Longstop Date) to take account of the impact of the Qualifying Change in Law;
			(b)	any adjustment in respect of the Licensee's obligations under this licence where and to the extent that the Licensee is unable to perform or is delayed in performing such obligations as a result of the Qualifying Change in Law:
				 (i) for such period as is necessary having regard to the Qualifying Change in Law; or

1		
		 (ii) until such time as the obligations have been amended to reflect the impact of the Qualifying Change in Law;
	(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Qualifying Change in Law;
	(d)	any adjustments to the Revenue Calculations as a result of the Qualifying Change in Law, including without limitation:
		 (i) any reasonable increases or decreases to any ex ante allowance (with cost sharing) (including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of SRAV Capex, Opex, On-going Capex or On-going Devex); and
		 (ii) any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Qualifying Changes in Law; and/or
	(e)	any adjustments to the WACC for the Commissioning Period as determined in accordance with condition G9 (<i>WACC</i>).
13.2	Projec	egulator will determine the impact on the tand any adjustments to be provided to the ee to reflect such impact.
13.3		ng notification by the Licensee under on G13.1, the Licensee shall provide to the tor:
	(a)	any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Qualifying Change in Law or potential Qualifying Change in Law; and
	(b)	where applicable in the context of an increase or decrease of the Licensee's costs pursuant to condition G13.2, satisfactory evidence to the Regulator that the Qualifying Change in Law affects construction and/or the cost impact to On-going Devex, Capex or Opex.
13.4		nount of any increases to the Licensee's costs ant to condition G13.2 shall be subject to the

			Qualit and/c	see providing satisfactory evidence that the fying Change in Law affects construction or the cost impact to On-going Devex, Capex bex pursuant to condition G3.3.		
14.	Force Majeure	14.1	14.1 The Licensee shall notify the Regulator as soon becomes aware of any Force Majeure Ever potential Force Majeure Event, during Commissioning Period, and shall provide deta respect of such Force Majeure Event or pot Force Majeure Event to the Regulator together its analysis of the effects, or likely effects, of Force Majeure Event including but not limited			
			(a)	any adjustments to key milestone dates (including the Scheduled COD and/or the Longstop Date) to take account of the impact of the Force Majeure Event;		
			(b)	any adjustment in respect of the Licensee's obligations under this licence where and to the extent that the Licensee is unable to perform or is delayed in performing such obligations as a result of the Force Majeure Event:		
				 (i) for such period as is necessary having regard to the Force Majeure Event; or 		
				 (ii) until such time as the obligations have been amended to reflect the impact of the Force Majeure Event; 		
			(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Force Majeure Event;		
			(d)	any adjustments to the Revenue Calculations as a result of the Force Majeure Event, including without limitation:		
				 (i) any reasonable increases or decreases to any ex ante allowance (with cost sharing) (including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of SRAV Capex, Opex, On-going Capex or On-going Devex); and 		
				 (ii) any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Force Majeure Event; and/or 		

			(e) any adjustments to the WACC for the Commissioning Period as determined in accordance with condition G9 (<i>WACC</i>).	
		14.2	The Regulator will determine the impact on the Project and any adjustments to be provided to the Licensee to reflect such impact.	
		14.3	Following notification by the Licensee under condition G14.1, the Licensee shall provide to the Regulator:	
			 (a) any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Force Majeure Event or potential Force Majeure Event; and 	
			(b) where applicable in the context of an increase or decrease of the Licensee's costs pursuant to condition G14.2, satisfactory evidence to the Regulator that the Force Majeure Event affects construction and/or the cost impact to On-going Devex, Capex or Opex.	
		14.4	The amount of any increases to the Licensee's costs pursuant to condition G14.2 shall be subject to the Licensee providing satisfactory evidence that the Force Majeure Event affects construction and/or the cost impact to On-going Devex, Capex or Opex pursuant to condition G14.3.	
15.	Relief Event	15.1	The Licensee shall notify the Regulator as soon as it becomes aware of any Relief Event, or potential Relief Event, during the Commissioning Period, and shall provide details in respect of such Relief Event or potential Relief Event to the Regulator together with its analysis of the effects, or likely effects, of such Relief Event on the Scheduled COD.	
		15.2	The Regulator will determine the impact on the Scheduled COD and any extensions to be provided to the Scheduled COD to reflect such impact.	
		15.3	Following notification by the Licensee under condition G15.1, the Licensee shall provide to the Regulator any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Relief Event or potential Relief Event.	
		draftin	Relief Event mechanism to be amended in full form g to so that it will operate to extend the Longstop as well as the Scheduled COD.	

Section	H:	Operational	Period	Conditions
---------	----	-------------	--------	------------

1.Operational Period1.1The Operational Period is the period from the end Commissioning Period to revocation of the licence	of the
means that the Operational Period will start durin First Regulatory Period (after the Construction and Commissioning Period) and continue in the s and subsequent regulatory periods.	e. This ng the Period
2. Post Commissioning 2.1 The Licensee shall provide the Regulator with information as the Regulator may require to enab carry out the Post Commissioning Review and activities under this condition H2 (<i>Post Commiss Review</i>), including:	e it to other
 (a) figures for the Commissioning Period, which be provided no later than [●] Business following the Commercial Operations Date 	Days
 (b) figures and other information required for purpose of determining (having regard relevant circumstances) the Allowed Revert each Charging Year following the Commo Distribution Date until the date of conclusions the first annual iteration process (referent condition H16.1(b)). 	to all ue for nercial ion of
2.2 Following the Commercial Operations Date Regulator shall undertake a review of Commissioning Period (a "Post Commissi Review"), which shall include calculating adjust for a reconciliation based on the figures for Commissioning Period provided by the Licensee condition H2.1(a) to be applied to the calculat Allowed Revenue and RAV as part of the first a iteration process (referenced in condition H16.1(b)	the oning ments r the under ion of annual
3. Allowed summary Revenue: 3.1 Allowed Revenue is the annual amount that the Lice is entitled to recover, calculated according to building blocks set out in condition H5 (Allowed Recalculation).	o the
3.2 Allowed Revenue will be calculated and verified annual basis and calibrated periodically (for exa every five years) during a Periodic Review.	
3.3 Allowed Revenue for the initial period of the Opera Period shall be calculated as set out in conditio (<i>Post Construction Review</i>). The Allowed Revenu then be adjusted to reflect the reconciliation of the for the Commissioning Period undertaken at the Commissioning Review (see condition H2 <i>Commissioning Review</i>)).	n G10 1e will SRAV
4. Regulatory period 4.1 Following the end of the First Regulatory Period, Pe	riodic

		4.2	information as carry out eac purpose of det circumstances) Year during eac	the ch terr) th ch I ual	all provide the Regulator with such e Regulator may require to enable it to Periodic Review, including for the mining (having regard to all relevant e Allowed Revenue for each Charging Periodic Review period (subject to any iteration process (referenced in))).	
		4.3	regulatory peri advance of the the regulatory synchronised,	iod e re pe if	ne second (or another subsequent) may be adjusted by the Regulator (in gulatory period commencing) so that eriods of different T&S networks are it is considered desirable for price nt T&S networks to take place at the	
			include referen	nce n se	ded that the economic guidance will to the factors that the Regulator will etting the length of the second or atory periods.	
5.	Allowed Revenue calculation	5.1	For the Operati calculated as fo		al Period, the Allowed Revenue will be ws:	
			Allowed Rev AdjNominal _t +		iue_t = Allowed Revenue before jNominal _t	
			Allowed Revenue before $AdjNominal_t = Calculated$ Revenue (real) _t x CPIH _t / CPIH _{BaseYear} + Adj _t			
			Calculated Re Decom _t +Tax _t	eve	nue (real) _t = RoC_t + Depr _t + Opex _t +	
			where:			
			RoCt	=	$RAV_t \times WACC_i;$	
			RoC	=	the return on capital which the Licensee is expected to make during the First Regulatory Period;	
			RAV	=	an amount calculated in accordance with condition H6.3 (<i>RAV</i>);	
			WACC	=	an amount calculated in accordance with condition H8 (WACC);	
			Depr	=	an amount calculated in accordance with condition H9 (<i>Depreciation</i>);	
			Opex	=	an amount calculated in accordance with condition H10 (<i>Opex</i>);	
			Decom	=	an amount calculated in accordance with condition H13 (<i>Decommissioning allowance</i>);	

			Тах	-	an amount calculated in accordance with condition H15 (<i>Tax</i>);
			Adj	V	an amount calculated in accordance with condition H16 (<i>Adjustments</i>); and
			AdjNominal	v	an amount calculated in accordance with condition H23 (<i>Under / overutilisation adjustment</i>).
			calculated in	real t	ed Revenue (real) amounts will be terms, with the exception of some llowance which will be on a nominal
				∕ has	updated once the calculation been settled. Details of Pass be included in cost treatment paper.
6.	RAV	6.1	The RAV sha following form		calculated in accordance with the
					ansferred SRAV _t + On-going Capex _t t] - Depr _t – Disposals _t + AdjRAV
			where:		
			Transferred SRAV	=	SRAV which is to be moved to the RAV in accordance with H6 (<i>Transferred SRAV and additional SRAV(s)</i>);
			On-going Capex	=	an amount as referenced in condition H11 (<i>On-going Capex</i>);
			[On-going Devex	=	an amount as referenced in condition H12 (<i>On-going Devex</i>);]
					Note: mechanism to be developed – see condition H12 (On-going Devex).
			Depr	=	an amount calculated in accordance with condition H9 (<i>Depreciation</i>);
			Disposals	=	an amount calculated in accordance with condition H14 (<i>Disposals</i>). At the Post Commissioning Review there will be a reconciliation of SRAV for the Construction Period and the Commissioning Period. At this point, the rolled-up Return

				During Construction will be included in the SRAV; and
			AdjRAV =	an amount calculated in accordance with condition H16 (<i>Adjustments</i>).
		6.2	terms, with reference	ured and maintained in real CPIH ce to the Base Year. Where required, Il be adjusted to align with this price
		6.3		e calculation of RoC will be the NPV- base which will be calculated as the d Discounted RAV _t
			where:	
			Discounted RAV	= RAV / (1+WACC); and
			WACC	 an amount calculated in accordance with condition H8 (WACC).
7.	Transferred SRAV and Additional SRAV(s)	7.1	reconciliation of th	moved to the RAV following the ne SRAV at the Post Construction dance with condition G10 (<i>Post v</i>).
		7.2	accrual of additiona phasing of the Appr or in the event of e	lude mechanics which allow for the al SRAVs (for example to reflect any roved T&S Network by the Licensee expansion of the T&S Network) and additional SRAVs to the RAV.
		7.3	accordance with p (Construction Period Period Conditions).	RAVs will accrue and transfer in provisions equivalent to sections F d Conditions) and G (Commissioning . This means that any additional ransfer to the RAV following the nissioning Review.
8.	WACC	8.1	WACC, being the Settlement Schedul as set out in cond	e First Regulatory Period, Post-COD amount set out in the Financial e for the Post-COD WACC, will apply dition G9.1, and any references to n H shall be construed accordingly.
		8.2	WACC is the regu determined by the	subsequent regulatory periods, the ulated average return on capital, e Regulator, and adjusted by the me to time according to market
		8.3		onal Period includes part of a year, hat part of a Charging Year shall be

9.	Depreciation	9.1 The Licensee will be paid back a portion of its RAV in each period over the economic life of the asset.
		9.2 The depreciation profile for at least the First Regulatory Period is to be straight line real terms by reference to the expected life of the T&S network assets as set out in the Technical Details Schedule, with depreciation commencing at COD for amounts transferred from the SRAV or in the year following spend for On-going Capex.
10.	Орех	Note: T&SCo to refer to separate cost treatment paper.
11.	On-going Capex	Note: T&SCo to refer to separate cost treatment paper.
12.	On-going Devex	Note: T&SCo to refer to separate cost treatment paper.
13.	Decommissioning Allowance	13.1 The Decommissioning Allowance building block is intended to cover the costs that will be incurred by the Licensee in funding the decommissioning and monitoring liability related to the T&S Network, which shall be calculated on the basis of:
		 (a) the Onshore Decommissioning Fund Contribution Amount referenced in condition C3 (Decommissioning obligations);
		 (b) the Offshore Decommissioning Fund Contribution Amount referenced in condition D3 (Decommissioning obligations).
		Note: The initial estimate of decommissioning costs will be determined as part of the initial settlement.
14.	Disposals	14.1 Where there is a Disposal of a T&S Asset, which is permitted under standard condition B8 (<i>Disposal of assets and restrictions on charges</i>), such Disposal shall result in an adjustment to the RAV determined by the Regulator.
		14.2 The Regulator shall determine a share of the positive difference between (a) the actual sale value of a T&S Asset subject to a permitted Disposal; and (b) the value ascribed to such T&S Asset in the RAV, which shall be accounted for as part of the annual iteration process (referenced in condition H16.1(b)).
15.	Тах	15.1 An allowance for the expected tax costs of the Licensee, using an approach that aligns with the Regulator's regulatory precedent, including a notional tax approach that holds the Licensee harmless to changes in tax resulting from changes to gearing, compliance obligations including the 'tax review', and the 'tax compliance statement'.

16.	Adjustments	16.1	Adjustments will be applied to the calculation of Allowed Revenue and RAV including for:
			(a) adjustments, including the store performance / availability adjustment (see condition H17 (<i>Availability adjustment</i>) below); and
			(b) true-ups, corrections and reconciliations in accordance with an annual iteration process administered by the Regulator, which shall include reconciliation of any relevant negative Monthly QRP Adjustments (as defined in the RSA) under the RSA.
			Note: it is envisaged that the Licensee will be required to provide details of revenues collected for assurance by the Regulator in order to facilitate the true- ups/reconciliations.
17.	Availability adjustment	17.1	In each Charging Year during the Operational Phase, the Licensee shall:
			(a) use reasonable endeavours to achieve the Availability Target, which shall apply to incentivise the Licensee to maximise store performance, minimise outages and to schedule planned outages in an efficient manner; and
			(b) achieve the Availability Floor.
			Notes:
			• Full form drafting to address scenario where COD occurs part way through a Charging Year, i.e. where there is a "short" first Charging Year.
			• the Availability Target is set out in [section 3] of the APDP and is 95% Availability. The Availability Target will take into account a reasonable and justified level of planned maintenance and unplanned outage in a Charging Year.
			• The Availability Floor [is set out in [section 3] of the APDP] and is 75% Availability. In circumstances where Availability falls below the Availability Target, the Availability incentive regime pursuant to conditions H17 and H19 shall apply, whether or not Availability remains above the Availability Floor.
			• Where the Licensee does not achieve the Availability Floor but (i) has an approved Availability Correction Plan and an approved associated Remediation Plan (if applicable) in place in accordance with conditions H17-H19; and (ii) is implementing and complying with such plan(s) in all material respects, the Licensee will not be in breach of condition H17.1(b).

17.2	Network Delivery "Availa	was Poir bility	ging Year, the amount of CO ₂ that the T&S capable of accepting at each of the ints shall be assessed to determine " for that Charging Year _t as follows: $1 - \frac{\left(\sum_{h=1}^{H} \sum_{i=1}^{N} CRC_{ih} + CN_{ih}\right)}{\sum_{h=1}^{H} \sum_{i=1}^{N} RC_{ih}}\right)$
	where:		
	А	=	Availability for Charging Year _t ;
	CN	=	curtailed nominations, i.e. reduction in the amount of nominated capacity that can be delivered affecting Delivery Point <i>i</i> in hour h (tCO_2), but excluding any curtailed nominations to the extent to which such curtailed nominations were directly caused by an Availability Relief Factor;
	CRC	=	constrained registered capacity, i.e. reduction in the amount of registered capacity that can be nominated affecting Delivery Point <i>i</i> in hour <i>h</i> (tCO_2), but excluding any constrained registered capacity to the extent to which such constrained registered capacity was directly caused by an Availability Relief Factor;
	Н	=	total number of hours in Charging Year _t , being 8760 hours;
	Ν	=	total number of Delivery Points in the T&S Network; and
	RC	=	registered capacity, i.e. registered capacity held at Delivery Point <i>i</i> in hour h (tCO_2).
			ligated Network Capacity will be set out in of the APDP.
	and curi context	tailed ions/r	finitions of constrained registered capacity nominations remain under review in the of development of the renominations procedure under the CCS c.
17.3	Availabil Charging to in cor	lity Re g Yea nditior	shall keep written records of any potential elief Factors that have occurred during a r, which shall include the details referred is H17.5(a) and H17.5(b) below in respect potential Availability Relief Factor.

	17.4	shall referre other keep,	otly upon request by the Regulator, the Licensee provide to the Regulator copies of the records ed to in condition H17.3 above, as well as any written records which the Licensee is required to publish and/or supply to Users under the CCS ork Code relating to:
		(a)	curtailed nominations, i.e. reduction in the amount of nominated capacity that can be delivered affecting Delivery Point <i>i</i> in hour <i>h</i> , howsoever caused ("Curtailed Nominations"); and
		(b)	constrained registered capacity, i.e. reduction in the amount of registered capacity that can be nominated affecting Delivery Point <i>i</i> in hour <i>h</i> , howsoever caused ("Constrained Registered Capacity").
	17.5	Availa specifi with a the Li which any p releva	icensee shall put forward its calculation of the bility for each Charging Year within [<i>period to be</i> <i>ied</i>] of the end of that Charging Year, together ny further evidence that the Regulator has notified censee that it requires in respect of the same, shall include written notice of the occurrence of potential Availability Relief Factors during the ant Charging Year, together with evidence actory to the Regulator in respect of:
		(a)	the nature, cause(s) and anticipated duration of each such potential Availability Relief Factors; and
		(b)	the anticipated impact of each such potential Availability Relief Factor on nominations (i.e. the nominated capacity that can be delivered affecting Delivery Point i in hour h) and/or registered capacity (i.e. the amount of registered capacity that can be nominated affecting Delivery Point i in hour h), including the anticipated duration of such impact.
	17.6	the L	egulator shall consider the evidence provided by icensee in respect of the calculation of the bility for a Charging Year and shall:
		(a)	confirm that the calculation of the Availability for such Charging Year is accepted; or
		(b)	confirm that the calculation of the Availability for such Charging Year is not accepted; or
		(c)	require further evidence from the Licensee to confirm whether the calculation of Availability for such Charging Year can be accepted,

	and where the Regulator requires further evidence from the Licensee under condition H17.6(c), the Licensee shall promptly provide such evidence to the Regulator. In considering the evidence provided by the Licensee in respect of the calculation of the Availability for a Charging Year, as referred to in condition H17.6 above, the Regulator shall consider the evidence provided by the Licensee in respect of the occurrence of potential Availability Relief Factors and shall, in each case:	
17.7		
	(a)	confirm that an Availability Relief Factor has occurred;
	(b)	confirm that an Availability Relief Factor has not occurred; or
	(c)	require further evidence from the Licensee to confirm whether an Availability Relief Factor has occurred,
	and where the Regulator requires further evidence from the Licensee under condition H17.7(c), the Licensee shall promptly provide such evidence to the Regulator and the procedure in condition H17.6 shall be repeated.	
17.8	An "Availability Relief Factor" , subject to H17.9 to H17.11 below, is any of the following factors:	
	(a)	the occurrence of a Force Majeure Event;
	(b)	the occurrence of a Qualifying Change in Law;
		Note: this should include for example a new HSE requirement to the extent it falls within the scope of Discriminatory Change in Law or Specific Change in Law, but not administration of existing licences/permits e.g. the licence and/or the Storage Permit.
	(c)	disruption to power supply resulting from an electricity transmission or distribution network outage (including any action by the electricity system operator) but not including any disruption to power supply resulting from events on the Licensee's side of a supply point connecting the T&S Network to the electricity transmission or distribution network, the impact of which the Regulator determines in accordance with condition H17.9 is beyond the reasonable control of the Licensee;
	(d)	an act or omission of a User, the impact of which the Regulator determines in accordance with condition H17.9 is beyond the reasonable control of the Licensee; and

	Note: this would include a User emergency and/or User delivery of off-specification CO_2 , to the extent it met the criteria.
(e)	asset damage caused by third parties or natural causes, subject to compliance with [reference T&SCo obligation to procure as noted in note (i) below.],
	Note:
	<i>Provisions reflecting the following principles to be incorporated:</i>
	(<i>i</i>) T&SCo to be under an obligation to procure asset damage insurance and associated BI insurance to the extent available;
	(ii) Code to provide for suspension of T&S charges to the extent covered by the BI insurance procured pursuant to the above obligation; and
	<i>(iii) RSA to treat amounts recoverable under the BI insurance procured pursuant to the above obligation as Market Revenues.</i>
	Detailed drafting to be developed, including scoping of procurement obligation and definition of amounts recoverable.
evider directl Consti Charg avoide Licens	th case provided that the Licensee has provided nee to the Regulator's satisfaction that such factor y caused Curtailed Nominations and/or rained Registered Capacity in the relevant ing Year which, in each case, could not have been ed or overcome or otherwise mitigated by the ee acting in accordance with Good Industry ce; and
evider factor Consti	h case to the extent that the Licensee has provided nee to the Regulator's satisfaction that neither such nor any such Curtailed Nominations and/or rained Registered Capacity were caused or puted to by acts and/or omissions of the Licensee
	(i) any of its Affiliates, Related Undertakings or Ultimate Controllers; or
	 (ii) any of its agents, employees, contractors, subcontractors or other suppliers (of any tier) working on its behalf.
Availa the d	e the Licensee notifies the Regulator of a potential ability Relief Factor within limbs (c) and/or (d) of efinition of "Availability Relief Factor" set out in tion H17.8 above, in determining whether or not

		the Li reasor	levant impact is beyond the reasonable control of censee the Regulator shall consider whether it is nable in all the circumstances to expect that the see, acting in accordance with Good Industry ce:
		(a)	should have been aware of the relevant triggers and/or contributory factors for: (i) the relevant potential Availability Relief Factor; and (ii) the impact of the same; and
		(b)	should have controlled or changed some or all of such triggers or contributory factors to prevent the relevant potential Availability Relief Factor from occurring or to prevent or minimise its impact,
		circum	ed that lack of funds shall not, in any stances, be interpreted as a cause beyond the able control of the Licensee.
	17.10	Where	e the Regulator:
		(a)	confirms under condition H17.6(b) that the calculation of the Availability for such Charging Year is not accepted, the Licensee shall reassess and resubmit its calculation to the Regulator and the procedures in condition H17.6 shall be repeated; or
		(b)	requires further evidence from the Licensee under condition H17.6(c), the Licensee shall provide such evidence to the Regulator and the procedures in condition H17.6 shall be repeated,
			ch case until the Regulator confirms that the ation of the Availability for such Charging Year is ed.
	17.11		e Availability, as confirmed by the Regulator in dance with condition H17.6(a), in Charging Year $_{\rm t}$:
		(a)	[save in the 12 month period following COD in respect of which condition H17.12 shall apply,] exceeds the Availability Target for Charging Year _t by any amount up to and including a maximum of 99% Availability, the Licensee will benefit from a positive adjustment to the Allowed Revenue, up to a cap calculated by reference to 25% of regulated return on equity for such Charging Year _t , such positive adjustment to apply to the Allowed Revenue in accordance with condition H17.6 above in Charging Year _{t+2} ; or
			Note: regulated return on equity means allowed equity return (cash-equivalent of the cost of

			equity, defined as CoE*equity portion of RAV), rather than return on regulatory equity (RoRE).
		(b)	[save in the 12 month period following COD in respect of which condition H17.12 shall apply,] fails to meet the Availability Target for Charging Year _t , the Licensee will suffer a negative adjustment to the Allowed Revenue, up to a cap calculated by reference to 100% of the regulated return on equity for such Charging Year _t , to apply to the Allowed Revenue in accordance with condition H16 (<i>Adjustments</i>) above in Charging Year _{t+2} .
	17.12	COD adjust decrea equity adjust not ad	ESNZ intends to offer a 12 month period post- in which (i) the maximum potential positive ment where the Availability Target is exceeded is ased to 12.5% of the relevant regulated return on ; and (ii) the maximum potential negative ment applicable where the Availability Target is chieved is decreased to 50% of the relevant ted return on equity.
	17.13	the Re falls b adjust	e Availability in a Charging Year, as confirmed by egulator in accordance with condition H17.6(a), below the Availability Floor, no further negative ments shall apply under condition H17.11(b) [or ion H17.12].
	17.14	Regula below promp Availal shall in Plan'' the pro- to imp and/or econor	ilability in a Charging Year, as confirmed by the ator in accordance with condition H17.6(a), falls the Availability Target, the Licensee shall otly submit to the Regulator for approval an bility correction plan, the implementation of which mprove Availability (an "Availability Correction), together with evidence of the justification for oposed works and/or activities that are necessary prove Availability (including any alternative works r activities considered by the Licensee), and the mic, efficient and effective costs of the works r activities.
	17.15	Where	::
		(a)	Availability falls below the Availability Target as a result of or in connection with an issue or issues arising at the Storage Site and/or otherwise as a result of or in connection with store performance; and
		(b)	new capital expenditure is required as part of the related Availability Correction Plan,
			the Licensee shall submit a Remediation Plan ant to condition H19 (<i>Remediation works</i>).

		(subject to a s		where new capex hreshold) is required an.
	17.16	provided an Av for approval u	ndition H17.19, whe vailability Correction nder condition H17.10 vailability Correction	Plan to the Regulator D, the Regulator shall
		(a) confirm approve	that the Availabilit ed; or	y Correction Plan is
			n that the Availability (ed and provide its re al.	
	17.17	condition H1 economic, efficient implementing into account, related spend	an Availability Cor 7.16, the Regulato cient and effective cos the Availability Cor among other thin I in respect of whic provided to the Licens	or will assess the sts directly related to rection Plan, taking gs, any correction- ch allowances have
		Availability reg	apex/opex position gime is set out in the rrective Measures app	
			-	
			Capex	Opex
		Availability remediation (Storage Site)	<i>Capex</i> <i>80/20 cost share</i> <i>Condition H19</i> <i>(Remediation Plan)</i>	<i>Opex</i> 80/20 cost share Cost treatment paper
		remediation (Storage Site)	80/20 cost share Condition H19 (Remediation Plan)	80/20 cost share Cost treatment paper
		remediation (Storage	80/20 cost share Condition H19	80/20 cost share Cost treatment
	17 18	remediation (Storage Site) Availability remediation (non- Storage Site)	80/20 cost share Condition H19 (Remediation Plan) 40/60 cost share Cost treatment paper	80/20 cost share Cost treatment paper 40/60 cost share Cost treatment
	17.18	remediation (Storage Site) Availability remediation (non- Storage Site) Where the Reg (a) confirm Availab License Availab License	80/20 cost share Condition H19 (Remediation Plan) 40/60 cost share Cost treatment paper gulator: as under condition pility Correction Plan ility Correction pility Correction plan ility correction Plan eshall promptly promptly ame	80/20 cost share Cost treatment paper 40/60 cost share Cost treatment paper H17.16(a) that the h is approved, the lement the approved or
	17.18	remediation (Storage Site) Availability remediation (non- Storage Site) Where the Reg (a) confirm Availab License Availab License Availab License Availab	80/20 cost share Condition H19 (Remediation Plan) 40/60 cost share Cost treatment paper gulator: as under gulator: as under condition ility Correction paper gulator: as under as under paper as under paper gulator: as under condition ility Correction paper as under condition ility Correction paper paper	80/20 cost shareCosttreatmentpaper40/60 cost shareCosttreatmentpaperH17.16(a) that thehement the approved, thelement the approved orH17.16(b) that theis not approved, theend and resubmit theto the Regulator andonH17.16 shall be

			will ı (<i>Rem</i> e	not re <i>ediatio</i>	Plan is approved under condition H17.16 equire reapproval under condition H19 n works). vailability Correction Plan process is used to
		assess different options which might include different combinations of capex and/or opex. It is envisaged that once an overall Availability Correction Plan solution has been approved, the capex element may need a further approval process under H19, unless sufficient detail has been provided and approved under the Availability Correction Plan process.			
18.	Corrective Measures	18.1			icensee is required to undertake Corrective respect of the T&S Network:
			(a)	Plan	aant to the Licensee's Corrective Measures as included in the Storage Permit; and/or a active Measures Direction; and/or
			(b)	meas requi	spect of any Corrective Measures (and/or ures for the protection of human health) red under the Storage Permit which do not ithin condition H18.1(a) above,
			the Li	censee	e shall promptly provide to the Regulator:
				(i)	the details of such Corrective Measures Plan, Corrective Measures Direction and Corrective Measures (if any); and
				(ii)	any related and/or additional information that the Regulator considers appropriate and requested from the Licensee in respect of such Corrective Measures Plan, Corrective Measures Direction and Corrective Measures (if any).
			Meası opex intent	regim tion is	opex costs arising pursuant to Corrective ill need to be reviewed in the context of the e under the cost treatment process. The for the treatment of opex and capex in orrective Measures to be consistent.
		18.2	to con expen have of the	ndition nditure alread e same	e shall submit a Remediation Plan pursuant H19 (<i>Remediation works</i>) where new capital (that is in addition to any allowances that y been provided to the Licensee in respect activity) is required directly as a result of tion with:
			(a)	Plan a	mplementation of a Corrective Measures and/or any Corrective Measures (and/or any ures for the protection of human health); or
			(b)	in res	sponse to a Corrective Measures Direction,

			in each case as notified by the Licensee to the Regulator pursuant to condition H18.1.
			Note: this is designed to apply where substantive new capex (subject to a specified de minimis threshold) is required pursuant to Corrective Measures.
19.	Remediation works	19.1	Where:
			(a) condition H17.15 applies;
			(b) condition H18.2 applies; and/or
			(c) the Licensee otherwise proposes to undertake remediation works relating to the Storage Site, the implementation of which shall (i) require new capital expenditure; and (ii) improve Availability in the future,
			the Licensee shall promptly provide to the Regulator for approval a remediation plan setting out the remediation works that the Licensee proposes to undertake to the T&S Network (a "Remediation Plan"), together with:
			 evidence of the justification for the remediation works (including any alternative remediation works considered by the Licensee), and the economic, efficient and effective costs of the remediation works; and
			 (ii) any related and/or additional information that the Regulator considers appropriate and requested from the Licensee in respect of such Remediation Plan.
			Note: separate consideration is being given to the need for the Licensee to procure asset damage and business interruption insurance in the context of the licence, the CCS Network Code and the RSA.
		19.2	The Licensee shall include in any Remediation Plan provided to the Regulator under condition H19.1 such remediation works as are necessary:
			 (a) in the case of conditions H19.1(a) and H19.1(c), to improve Availability; and
			(b) in the case of condition H19.1(b) and subject to condition H19.3, to meet the requirements of the relevant Corrective Measures Plan, Corrective Measures Direction or any Corrective Measures (and/or any measures for the protection of human health) (as the case may be) ("Corrective Measures Requirements").

	19.3	neces	e the Remediation Plan includes remediation works sary to meet any Corrective Measures rements, the Licensee shall:
		(a)	provide evidence that it has informed [the NSTA, HSE, OPRED] of such Remediation Plan and, where required by [the NSTA, HSE, OPRED], that it has discussed the Remediation Plan with [the NSTA, HSE, OPRED]; and
		(b)	take into account any comments, and comply with any directions, provided by [the NSTA, HSE, OPRED] on the Remediation Plan (if any).
	19.4	the Re	e the Licensee has provided a Remediation Plan to egulator for approval under condition H19.1, the ator shall review such plan and shall:
		(a)	subject to the Licensee having complied with condition 19.3, confirm that the Remediation Plan is approved; or
		(b)	confirm that the Remediation Plan is not approved and provide its reasons for such non-approval.
	19.5	Where	e the Regulator:
		(a)	confirms under condition H19.4(a) that the Remediation Plan is approved, the Licensee shall promptly implement the approved Remediation Plan; or
		(b)	confirms under condition H19.4(b) that the Remediation Plan is not approved, the Licensee shall promptly amend and resubmit the Remediation Plan to the Regulator and the procedures in condition H19.4 shall be repeated.
	19.6	assess	proving a Remediation Plan, the Regulator will s the economic, efficient and effective remediation and, subject always to condition H19.7:
		(a)	where condition H19.1(a) and/or condition H19.1(c) applies, 80% of the Licensee's economic, efficient and effective remediation spend shall accrue to the RAV; and
		(b)	where condition H19.1(b) applies, 100% of the Licensee's economic, efficient and effective remediation spend shall accrue to the RAV.
		capex	ESNZ/Ofgem considering whether remediation may in certain circumstances be funded through lowed Revenue adjustment rather than RAV al.

		1		1
		19.7		essing remediation spend which may accrue to the he Regulator will take into account:
			(a)	the extent to which the remediation spend directly relates to the implementation of the Remediation Plan;
			(b)	any remediation spend in respect of which allowances have already been provided to the Licensee;
			(c)	any remediation spend which is covered by commercial insurances and/or the Supplemental Compensation Agreement; and
			(d)	the extent to which:
				 a failure to meet the Availability Target is caused or contributed to by acts and/or omissions of the Licensee, including any failure by the Licensee to take reasonable steps that are consistent with Good Industry Practice to manage, minimise and/or otherwise mitigate the effects on Availability; and/or
				(ii) the need to perform a Corrective Measure is attributable to any fault, negligence and/or misconduct of the Licensee.
			Target omissi Correc neglige percer RAV u	to the extent that a failure to meet the Availability is caused or contributed to by acts and/or ions of the Licensee, or the need to perform a ctive Measure is attributable to any fault, ence and/or misconduct of the Licensee, the ntage of remediation spend that accrues to the under condition H19.6 may be reduced on a rtionate basis.
20.	Store performance	20.1	prepar Netwo respec	censee shall monitor store performance and shall re capacity forecasts in respect of the T&S rk, and shall provide reports to the Regulator in ct of the same in accordance with the programme re format specified by the Regulator.
		20.2	The Li	censee shall procure that:
			(a)	independent audits of the Licensee's capacity assumptions (including models) and corresponding Carbon Storage and Development Plan are undertaken and completed ("Competent Person's Audit"); and
			(b)	the results of any such audit are promptly provided to the Regulator in a report ("Competent Person's Report") or as an

	update to a previously provided Competent Person's Report,
	in each case in accordance with the requirements of conditions H20.3 to H20.7.
20.3	Following the Commercial Operations Date, the Licensee shall procure that:
	 (a) a further Competent Person's Audit is carried out at least once within the 12-month period following the Commercial Operations Date and, subject to condition H20.4 below, thereafter further Competent Person's Audits are carried out on an annual basis (each a "CPR Year"); and
	(b) following any such further audit, the Initial Competent Person's Report as updated in respect of the preceding audit (the "Preceding Competent Person's Report") (if relevant) is promptly updated to reflect the results of such further audit; and
	(c) the updated Competent Person's Report is provided to the Regulator no later than 12 months from the date that the Preceding Competent Person's Report was provided to the Regulator.
20.4	The frequency of Competent Person's Audits and updates to the Competent Person's Report during the second and/or subsequent regulatory periods may be adjusted by the Regulator.
20.5	Where the Regulator considers that there has been a significant and adverse deviation from the expected behaviour of the Storage Complex (such expected behaviour being based on the findings of previous Competent Person's Audits as recorded in the relevant updated Competent Person's Report), the Regulator shall be entitled to determine the relevant parts and/or elements of such Competent Person's Report that are to be updated to take into account such deviation, and the Licensee shall procure that such relevant parts and/or elements of the Competent Person's Report are promptly updated.
20.6	A Competent Person's Audit and Competent Person's Report (and any updates thereto) must be undertaken by a "Competent Auditor" , being an independent and qualified auditor who has:
	(a) experience in undertaking audits of carbon capture, usage and storage and/or oil and gas development projects similar in scope, character and complexity to the Project and has the

		expertise to assess and report on the matters that are set out in condition H20.7; and
	(b)	been appointed in accordance with the following process:
		 the Licensee shall send details of the proposed auditor to the Regulator, including information about how the auditor satisfies the criteria set out in paragraph (a);
		(ii) the Regulator shall consider the information provided by the Licensee under condition H20.6(b)(i) and shall advise the Licensee whether it approves (or does not approve) the auditor selected by the Licensee; and
		(iii) if the Regulator has approved the auditor then the Licensee shall appoint that auditor, but if the Licensee has failed to provide its approval then the Licensee shall seek the Regulator's approval to an alternative auditor in accordance with the requirements of this condition H20.6.
20.7		Competent Person's Report (and any updates to) must address the following matters:
	(a)	an assessment of the Storage Capacity and Resources (as each such term is defined in the SRMS Guidelines), i.e.:
		 (i) 1P, 2P and 3P (as each such term is defined in the SRMS Guidelines) which will be based on forecast user demand and licence duration; and
		 (ii) the capacity assuming unconstrained user demand for the same technical assumptions, based on maximum throughput of the designed facilities of the T&S Network and the Storage Complex, distinguishing previously stored quantities; and
		 (iii) contingent resources, i.e. 1C, 2C and 3C (as each such term is defined in the SRMS Guidelines);
	(b)	outstanding uncertainties and current level of residual risk associated with 1P, 2P and 3P (as each such term is defined in the SRMS Guidelines);

		•		at in respect of conditions H20.7(a) to ove, the required information shall include:
			(i)	the actual information in respect of the preceding Charging Year (save where the year during which the report is being prepared is the first Charging Year);
			(ii)	confirmation of the information that had previously been forecast in respect of the Charging Year referred to in limb (i) above (or where the year during which the report is being prepared is the first Charging Year, the first Charging Year);
			(iii)	confirmation of the information that had previously been forecast for all Charging Years subsequent to the Charging Year referred to in limb (i) above (or where the year during which the report is being prepared is the first Charging Year, subsequent to the first Charging Year);
			(iv)	current revised forecasts for the Charging Year in which the report is being prepared and all subsequent Charging Years.
			-	red contents of each Competent Person's further considered by ESNZ.
	20.8	Repor accord	t (or a dance <i>mance</i>	time as providing any Competent Person's any update thereto) to the Regulator in with this condition H20 (<i>Store</i>), the Licensee shall provide a copy of the
		(a)	the NS	STA; and
		(b)		e such report is produced during, or relates e First Regulatory Period, the Secretary of
	20.9		/repoi blex/e	rting in relation to additional Storage
		(a)	in Sc Storag expan "Addi Comp of pr Regula then Comp Addition	e the Licensee intends to propose a Change ope which would involve an additional ge Complex or Storage Complexes or an sion to an existing Storage Complex (an itional or Expanded Storage olex "), the Licensee shall have, in advance oposing such Change in Scope to the ator, procured that the Competent Auditor appointed undertake and complete a etent Person's Audit in respect of the onal or Expanded Storage Complex, and produce to the Regulator (with a copy to the

		 NSTA and, during the First Regulatory Period, the Secretary of State), a Competent Person's Report in accordance with the requirements of condition H20.7 reporting the results of such audit, such report forming part of the Licensee's Change in Scope proposal to the Regulator. (b) Following the date of completion of an Competent Person's Audit referred to in condition H20.9(a) in respect of an Additional or Expanded Storage Complex, the Licensee shall procure that during construction and commissioning of the same (if any), and following the relevant commercial operations date associated with the Additional or Expanded Storage Complex (if any);
		 any): (i) the relevant Additional or Expanded Storage Complex shall thereafter be audited as part of the T&S Network-wide Competent Person's Audits which take place under and in accordance with condition H20.2 and conditions H20.3 to H20.7; and
		 (ii) the results of any Competent Person's Audit which relate to such Additional or Expanded Storage Complex shall be included in the relevant updates to the T&S Network-wide Competent Person's Report as referred to in condition H20.3.
		<i>Note:</i> the Regulator will review the data supplied under condition H20 in order to assess at each Periodic Review any updates needed to the availability/store performance regime in condition H17.
21.	Bad Debt Allowance	Note: the mechanics of the bad debt allowance remain subject to on-going development. The working assumption is that the bad debt allowance will be set at zero for at least the First Regulatory Period given other protections on offer to the Licensee (such as the RSA) and value for money considerations.
22.	Market engagement costs	Note: remains subject to review as part of the cost assessment process.
23.	Under/overutilisation adjustment	23.1 In accordance with section 7.3 of the RSA, where in any month the Monthly IDP Amount is less than any negative Monthly QRP Adjustment (each as defined in the RSA), the difference between these amounts shall be addressed through the annual iteration process pursuant to condition H16.1(b).
		23.2 The amount of any difference referred to in condition H23.1 above will be taken into account as part of the "AdjNominal" for Charging Year_{t+2}.

24. Chan	nge in Scope	24.1	becom Opera such (analys	Licensee shall notify the Regulator as soon as it mes aware of any Change in Scope during the rational Period and shall provide details in respect of Change in Scope to the Regulator together with its ysis of the effects, or likely effects, of such Change tope, including but not limited to:
			(a)	any adjustments to the Revenue Calculations as a result of the Change in Scope, including without limitation:
				 (i) any reasonable increases or decreases to any ex ante allowance (with cost sharing) including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of Opex, On-going Capex, On-going Capex or On-going Devex; and
				 the impact on any ex post assessments of the economic and efficient expenditure by the Licensee;
			(b)	any adjustment in respect of the Licensee's obligations under this licence where and to the extent that the Licensee is unable to perform or is delayed in performing such obligations as a result of the Change in Scope:
				(i) for such period as is necessary having regard to the Change in Scope; or
				 (ii) until such time as the obligations have been amended to reflect the impact of the Change in Scope;
			(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Change in Scope;
			(d)	any adjustments to key milestone dates to take account of the impact of the Change in Scope;
			(e)	the inclusion of new construction periods or operational periods (and associated SRAVs) in respect of the Change in Scope (see condition E2 (<i>T&S Network expansion and additional</i> <i>construction, commissioning and operational</i> <i>periods</i>) above); and/or
			(f)	any adjustments to the WACC for the Operational Period as determined in accordance with condition H8 (<i>WACC</i>).

-		1			
		24.2		-	or will determine the impact on the Project ustments to be provided to the Licensee.
		24.3	H24.1 additi appro	, the L onal ir priate	tification by the Licensee under condition icensee shall provide to the Regulator any oformation that the Regulator considers and requests from the Licensee in respect age in Scope.
		24.4	where Expan	e a Ch Ided S	iting the generality of condition H24.3, ange in Scope involves an Additional or torage Complex, the provisions of H20.9 n respect of the same.
25.	Change in Law	25.1 The Licensee shall notify the Regulator as soon as it becomes aware of any Qualifying Change in Law, or potential Qualifying Change in Law, during the Operational Period and shall provide details in respect of such Qualifying Change in Law or potential Qualifying Change in Law to the Regulator together with its analysis of the effects, or likely effects, of such Qualifying Change in Law including but not limited to:			
			(a)		djustments to key milestone dates to take nt of the impact of the Qualifying Change v;
			(b)	obliga exten is del	adjustment in respect of the Licensee's ations under this licence where and to the t that the Licensee is unable to perform or ayed in performing such obligations as a of the Qualifying Change in Law:
				(i)	for such period as is necessary having regard to the Qualifying Change in Law; or
				(ii)	until such time as the obligations have been amended to reflect the impact of the Qualifying Change in Law;
			(c)	asses again specia Qualif Availa	djustments to the targets or outputs (or the sment of the performance of the T&S Assets st such targets or outputs) under these al conditions to reflect the impact of the fying Change in Law (other than the ability Target, which is addressed in tion H17 (<i>Availability adjustment</i>));
			(d)	a res	djustments to the Revenue Calculations as sult of the Qualifying Change in Law, ling without limitation:
				(i)	any reasonable increases or decreases to any ex ante allowance (with cost sharing) including: (i) any change to the cost sharing factor; and (ii) any ex ante

				(ii)	allowance in respect of Opex, On-going Capex, On-going Capex or On-going Devex; and any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Qualifying Changes in Law; and/or
			(e)	Period	ljustments to the WACC for the Operational as determined in accordance with ion H8 (<i>WACC</i>).
		25.2	and a	-	or will determine the impact on the Project ustments to be provided to the Licensee to mpact.
		25.3		-	ification by the Licensee under condition censee shall provide to the Regulator:
			(a)	consid Licens	dditional information that the Regulator lers appropriate and requests from the ee in respect of such Qualifying Change in r potential Qualifying Change in Law; and
			(b)	decrea condit Regula affects	applicable in the context of an increase or ase of the Licensee's costs pursuant to ion H25.2, satisfactory evidence to the ator that the Qualifying Change in Law s construction and/or the cost impact to ing Devex, Capex or Opex.
		25.4	pursua Licens Qualif the co	ant to see pro ying Cl ost imp	of any increases to the Licensee's costs condition H25.2 shall be subject to the oviding satisfactory evidence that the nange in Law affects construction and/or pact to On-going Devex, Capex or Opex condition H25.3.
26.	Force Majeure	26.1	becom Force shall p Event togeth	nes awa Majeur provide or pote ner with	e shall notify the Regulator as soon as it are of any Force Majeure Event, or potential e Event, during the Operational Period, and details in respect of such Force Majeure ential Force Majeure Event to the Regulator is analysis of the effects, or likely effects, e Majeure Event including but not limited
			(a)		djustments to key milestone dates to take nt of the impact of the Force Majeure ;
			(b)	obliga	djustment in respect of the Licensee's tions under this licence where and to the that the Licensee is unable to perform or

		is delayed in performing such obligations as a result of the Force Majeure Event:
		 (i) for such period as is necessary having regard to the Force Majeure Event; or
		 (ii) until such time as the obligations have been amended to reflect the impact of the Force Majeure Event;
	(c)	any adjustments to the targets or outputs (or the assessment of the performance of the T&S Assets against such targets or outputs) under these special conditions to reflect the impact of the Force Majeure Event (other than the Availability Target, which is addressed in condition H17 (<i>Availability adjustment</i>)); and
	(d)	any adjustments to the Revenue Calculations as a result of the Force Majeure Event, including without limitation:
		 (i) any reasonable increases or decreases to any ex ante allowance (with cost sharing) including: (i) any change to the cost sharing factor; and (ii) any ex ante allowance in respect of Opex, On-going Capex, On-going Capex or On-going Devex;
		 (ii) any reasonable amendments to the manner in which the Regulator will assess any ex post costs to apply the same principles that the Licensee should not be on cost risk for any Force Majeure Event; and/or
	(e)	any adjustments to the WACC for the Operational Period as determined in accordance with condition H8 (WACC).
26.2	and a	Regulator will determine the impact on the Project my adjustments to be provided to the Licensee to t such impact.
26.3		ring notification by the Licensee under condition , the Licensee shall provide to the Regulator:
	(a)	any additional information that the Regulator considers appropriate and requests from the Licensee in respect of such Force Majeure Event or potential Force Majeure Event; and
	(b)	where applicable in the context of an increase or decrease of the Licensee's costs pursuant to condition H26.2, satisfactory evidence to the Regulator that the Force Majeure Event affects

			construction and/or the cost impact to On-going Devex, Capex or Opex.
		26.4	The amount of any increases to the Licensee's costs pursuant to condition H26.2 shall be subject to the Licensee providing satisfactory evidence that the Force Majeure Event affects construction and/or the cost impact to On-going Devex, Capex or Opex pursuant to condition H26.3.
27.	Regulated/non-regulated revenue	27.1	Any revenues generated by the Licensee in relation to the use of the T&S Network are to be considered as regulated revenue.
			Note: it is not anticipated that the Licensee will earn any non-regulated revenue (for example fees obtained in relation to the non-regulated transport of CO ₂ from markets outside of the licence area) in the First Regulatory Period.
		27.2	The Licensee shall not be entitled to undertake any non- regulated activities or earn any non-regulated revenue without the written consent of the Regulator.
		27.3	The Regulator will determine the treatment of any non- regulated revenues received by the Licensee.
			Note: for the avoidance of doubt, any non-regulated revenue received by the Licensee will be taken into account in the calculations made pursuant to the RSA – see the RSA heads of terms for further details.
28.	, 55	28.1	The Licensee shall:
	Methodology		(a) have in place a Use of System Charging Methodology, which sets out the methodology for the determination of the Use of System Charges, through which the Licensee can recover its Allowed Revenue for each Charging Year; and
			(b) design, establish, operate and maintain a functioning platform including all required interfaces to operate the Use of System Charging Methodology (the "System Charging Platform").
		28.2	The Use of System Charging Methodology that applies from Licence Award is the Use of System Charging Methodology set out in CCS Network Code as at Licence Award.
		28.3	The Licensee shall, for the purpose of ensuring that the Use of System Charging Methodology achieves the Relevant Objectives, keep the Use of System Charging Methodology at all times under review.
		28.4	The Licensee shall, in accordance with the provisions of the CCS Network Code, make such modifications to the

			Use of System Charging Methodology as may be requisite for the purpose of better achieving the Relevant Objectives, provided that any such modification must be approved by the Regulator.
29.	Use of System Charging Statement	29.1	For each Charging Year, the Licensee shall prepare a statement of its Use of System Charges for using the T&S Network and how those charges have been calculated in accordance with the Use of System Charging Methodology (the "System Charging Statement").
		29.2	The Licensee shall submit the draft System Charging Statement for each Charging Year to the Regulator for approval by [31 December] before the start of the Charging Year to which the draft System Charging Statement relates. Note: it is also envisaged that the Licensee will provide details of forecasts for years t+1 and t+2, which will also be reviewed by the Regulator.

Schedule 1: Specified Area

Schedule 1: Specified Area

UK

Schedule 2: Revocation

- 1. The Regulator may at any time revoke the licence by giving not less than 30 days' notice (24 hours' notice, in the case of a revocation under sub-paragraph 1(f)) in writing to the Licensee:
 - (a) if the Licensee agrees in writing with the Regulator that the licence should be revoked;
 - (b) if any amount payable under standard condition [A2] (*Payments by Licensee to the Regulator*) is unpaid 30 days after it has become due and remains unpaid for a period of 14 days after the Regulator has given the Licensee notice that the payment is overdue provided that no such notice shall be given earlier than the sixteenth day after the day on which the amount payable became due;
 - (c) if the Licensee fails:
 - (i) to comply with a final order (within the meaning of section [●] of the Act) or with a provisional order (within the meaning of that section) which has been confirmed under that section and (in either case) such failure is not rectified to the satisfaction of the Regulator within three months after the Regulator has given notice of such failure to the Licensee provided that no such notice shall be given by the Regulator before the expiration of the period within which an application under section [●] of the Act could be made questioning the validity of the final or provisional order or before the proceedings relating to any such application are finally determined; or
 - (ii) to pay any financial penalty (within the meaning of section [●] of the Act) by the due date for such payment and such payment is not made to the Regulator within three months after the Regulator has given notice of such failure to the Licensee - provided that no such notice shall be given by the Regulator before the expiration of the period within which an application under section [●] of the Act could be made questioning the validity or effect of the financial penalty or before the proceedings relating to any such application are finally determined;
 - (d) if the Licensee fails to comply with:
 - (i) an order made by the court under section 34 of the Competition Act 1998;
 - (ii) an order made by the Regulator under sections 158 or 160 of the Enterprise Act 2002;
 - (iii) an order made by the Competition and Markets Authority under sections 76, 81, 83, 84 and 161 of the Enterprise Act 2002;
 - (iv) an order/decision made by the Secretary of State under sections 66, 147, 160 or 161 of the Enterprise Act 2002;

Note: the statutory references in paragraph (d) above are subject to further review.

- (e) if the Licensee:
 - (i) has ceased to carry on the development and operation of a network for the transportation and storage of carbon dioxide; or
 - (ii) has not commenced carrying on the activity referred to in paragraph (i) above within 30 days of the date on which the licence comes into force;

- (f) if the Licensee:
 - (i) is unable to pay its debts (within the meaning of section 123(1) or (2) of the Insolvency Act 1986, but subject to paragraphs 2, 3 and 4 of this schedule) or has any voluntary arrangement proposed in relation to it under section 1 of that Act or enters into any scheme of arrangement (other than for the purpose of reconstruction or amalgamation upon terms and within such period as may previously have been approved in writing by the Regulator);
 - (ii) has a receiver (which expression shall include an administrative receiver within the meaning of section 251 of the Insolvency Act 1986) of the whole or any material part of its assets or undertaking appointed;
 - (iii) has an administration order under schedule B1 of the Insolvency Act 1986 made in relation to it;
 - (iv) passes any resolution for winding-up other than a resolution previously approved in writing by the Regulator; or
 - (v) becomes subject to an order for winding-up by a court of competent jurisdiction.
- 2. For the purposes of sub-paragraph 1(f)(i), section 123(1)(a) of the Insolvency Act 1986 shall have effect as if for "£750" there was substituted "£250,000" or such higher figure as the Regulator may from time to time determine by notice in writing to the Licensee.
- 3. The Licensee shall not be deemed to be unable to pay its debts for the purposes of subparagraph 1(f)(i) if any such demand as is mentioned in section 123(1)(a) of the Insolvency Act 1986 is being contested in good faith by the Licensee with recourse to all appropriate measures and procedures or if any such demand is satisfied before the expiration of such period as may be stated in any notice given by the Regulator under paragraph 1.
- 4. Any rights the Regulator may have to revoke the licence under paragraphs 1 and 6 of this schedule shall be automatically suspended whilst the Licensee is in transport and storage administration under Chapter 4 of Part 1 of the Act until such time as the Regulator has determined, having consulted with the T&S administrator (as defined in section 49 of the Act), that achieving the objective of the transport and storage administration (as defined in section 43 of the Act), is no longer reasonably practicable.
- 5. The Regulator may at any time revoke the licence by giving no less than seven days' notice in writing to the Licensee where the Regulator is satisfied that there has been a material misstatement (of fact) by, or on behalf of the Licensee, in making its application for the licence.
- 6. The Regulator may at any time revoke the licence by giving not less than 30 days' notice in writing to the Licensee where the following events occur:
 - (a) revocation of any legal permission which has a material adverse effect on the operation of the T&S Network; or
 - (b) issuance of a Discontinuation Notice by the Secretary of State.

Note: in relation to 6(a) above, in deciding whether to exercise the revocation power, the Regulator will take into account all the relevant circumstances, including the ability and timescales of the Licensee to remedy the position. The Regulator has the discretion but not an obligation to exercise its revocation powers.

- 7. Where the Regulator is entitled to revoke the licence, the Licensee acknowledges that a Transfer Scheme may be applied.
- 8. Where the Licensee is insolvent and the Regulator is entitled to revoke the licence under this Schedule 2, in making a decision on revocation the Regulator shall have regard to the [Enforcement Guidance], but in the event of any conflict between these licence terms and the guidance, the licence terms will prevail.

Schedule 3: Financial Settlement Schedule

The Financial Settlement Schedule sets out details of the financial settlement determined before Licence Award, as set out below.

Item reference	Item	Details
Conditions F13.2 and G9.2	Delay WACC	[0%]
Condition F10	Pre-Licence Award Devex	[to be inserted]
Conditions F13, G9 and H8	WACC	Pre-COD WACC
		Post-COD WACC
Condition F7	Re-Use Asset Valuation	[to be inserted]
Conditions F8 and G5	SRAV Capex	<i>Note:</i> T&SCo to refer to separate cost treatment paper.
Conditions F11, G7 and H12	On-going Devex	<i>Note:</i> T&SCo to refer to separate cost treatment paper.
Condition H11	On-going Capex	<i>Note:</i> T&SCo to refer to separate cost treatment paper.
Condition F9, G6 and H10	Opex	<i>Note:</i> T&SCo to refer to separate cost treatment paper.

Schedule 4: Technical Details Schedule

The Technical Details Schedule sets out procedural, technical and programming details relating to the construction and commissioning of the T&S Network, as set out below.

Item reference	Item	Details
Condition F2	Scheduled Handover Date	[include initial Scheduled Handover Date]
Condition G2 Scheduled COD		[include initial Scheduled COD]
Condition G2	Longstop Date	[include initial Longstop Date plus 12 months]
Condition H9	Asset life	[Include asset life for the Approved T&S Network]
Conditions F2.2 and G2.3	Project Programme	[Include high level information on delivery plan/programme with subsections as set out below.]
		Phase 1
		[Include details key milestones for Phase 1, e.g. mechanical completion of sub-systems, Scheduled Handover Date, anticipated date of COD Readiness, Scheduled COD and Longstop Date.]
		Phase 2
		[Include the expected date of mechanical completion and completion of commissioning of individual sub-systems, as well as details of any other key milestones for Phase 2.]
		Phase 3
		[Include details of key milestones in respect of Phase 3, e.g. start and end of individual stages such as Appraise, Define, Execute and initial views on financial investment decisions, construction periods, commissioning periods and commercial operations dates.]

Schedule 5: Derogations Schedule

Note: it is recognised that in some instances the manner in which a Licensee is taking forward the development of a T&S network, in terms of matters such as project structure, procurement and operating model, may not be fully in compliance with the standard conditions, but is nonetheless considered to be efficient, economic and effective. In such circumstances, at licence award, the licensee may be granted certain "derogations" from full compliance with specific obligations under certain standard licence conditions or alternatively a licence condition may only come into effect upon a direction from the Regulator. How these derogations are issued and documented will be considered on a case by case basis and will depend on the nature of the derogation (e.g., temporary vs enduring, complexity etc.). The decision on what forms of derogations from the standard conditions provided by the Licensee. The method of implementation of any agreed derogations remains under review, but options include:

- 1. regulatory consent process contained within a licence condition;
- 2. *derogation from a licence obligation via Regulator direction;*
- 3. regulator direction to give (or not give) a licence condition effect; and
- 4. Derogations Schedule.

Schedule 6: Template Form of Approved Project Development Plan

Approved Project Development Plan for [insert details of Licensee/Project]

1. Acronyms and definitions

[Include a section setting out relevant acronyms and definitions.]

2. Scope of Project and cluster overview

[Include a description of the scope of the proposed Project, cross referring to the Technical Details Schedule as appropriate. This description should include a schematic of each of Phase 1 and Phase 2, the overall capacity of the Approved T&S Network following each such Phase and the location of the Planned Initial Users. A second schematic should show the Phase 3 expansion projects relative to the Approved T&S Network.]

3. Approved T&S Network overview

[Phase 1 and Phase 2 together comprise the works and activities required to ensure that the Approved T&S Network can provide at least the CO₂ capacity required for the Planned Initial Users during the First Regulatory Period.

Include details of the Approved T&S Network, cross referring to the Technical Details Schedule as appropriate, with subsections as set out below.]

3.1 Phase 1

[Include details of Phase 1 works and activities which are those works and activities that will be completed prior to the Commercial Operations Date, i.e. Handover Works (see below) and Commissioning Activities (see below) with associated SRAV Capex. Cross refer to the Technical Details Schedule as appropriate, and include (i) an overview of the T&S Network; (ii) details of storage system; (iii) details of offshore and onshore transportation; (iv) details of relevant Independent Certifier; and (v) definition of "COD Readiness Activities" (see below).

- This section must define the "Handover Works", being the works and activities to be undertaken by the Licensee during the Construction Period as part of Phase 1 that are required to ensure that Phase 1 of the Approved T&S Network is able to commence commissioning, as referred to in the APDP. Some indicative examples of the works and/or activities that are likely to form part of the Handover Works include but are not limited to:
 - design and construction of the assets of Phase 1 of the Approved T&S Network (including any Offshore Pipeline Infrastructure, Offshore Transportation and Storage System, Storage Complex(s), metering or systems);
 - carrying out non-operational testing;
 - obtaining all required permits/consents for Phase 1 of the Approved T&S Network;
 - organisational activities (such as the hiring/training of operational staff).
- This section must define the "Commissioning Activities", being the activities to be undertaken by the Licensee during the Commissioning Period as part of Phase 1 in order to demonstrate that Phase 1 of the Approved T&S Network is able to commence operations. Some indicative examples of the activities that are likely to form part of the Commissioning Activities include but are not limited to:

- *designing, establishing, operating and maintaining the System Charging Platform; and*
- carrying out pre-commissioning activities (e.g. functional / statutory testing, drying, pressure testing and SAT);
- post construction integrity and function testing;
- a review ensuring all outstanding works relating to Phase 1 of the Approved T&S Network (i.e. those works not completed as part of the Handover Works but required to be completed prior to System Acceptance), performance testing and pre-COD contractual obligations have been completed;
- function testing of the offshore system through the onshore master control station;
- flooding, cleaning and gauging of offshore equipment;
- pipeline pressure testing;
- carry out injection system barrier leak tests;
- *de-watering, drying and nitrogen purging of the system;*
- final dryness check and nitrogen packing;
- [wells and subsurface pre-commissioning activities;]
- system pressurisation and first fill;
- initial flow and line testing;
- metering and data / control system interface operability, venting;
- testing of rejection system responsiveness User and Licensee side;
- testing of Storage Complex responsiveness;
- NSTA requirements;
- OPRED requirements;
- proving of assets (e.g. leakage levels on end-to-end operations, energy, and CO₂ performance (MWh/tCO₂ stored, availability (for a specified duration)), operability, start-up/shutdown operations, injectivity and Storage Complex responsiveness);
- proving of operational capability (e.g. emergency response, event response (dry runs), leakage analysis process, control and safety systems); and
- organisational activities (such as the hiring/training of operational staff); and
- end to end operation for a period of time.

- This section must define the "COD Readiness Activities", being the Commissioning Activities that can be completed without CO₂, which shall include all of the Commissioning Activities other than:
 - system pressurisation and first fill;
 - *initial flow and line testing;*
 - [testing of Storage Complex responsiveness, other than some testing such as, but not limited to, near wellbore permeability using a fluid like KCI brine];
 - end to end operation for a period of time; and
 - proving of assets (e.g. leakage levels on end-to-end operations, energy, and CO₂ performance (MWh/tCO₂ stored, availability (for a specified duration)), operability, start-up/shutdown operations, injectivity and Storage Complex responsiveness).]

Note: "COD Readiness Activities" drafting to be kept in-line with the Commissioning Activities drafting above.

3.2 Phase 2

[Include details of Phase 2 of the Project, i.e. those works and activities in respect of the Approved T&S Network that will be not completed prior to Commercial Operations Date, including details of any development activities in respect of the Approved T&S Network with associated On-going Capex, including, as relevant, development of the storage system and offshore/onshore transportation and any connection of Planned Initial Users to the Approved T&S Network (to the extent not connected as part of Phase 1). The descriptions of such works and activities should be provided in the same way as for Phase 1.]

3.3 Availability Target and Availability Floor

[Include details of the Availability Target, i.e. [95]% of Availability, and Availability Floor, i.e. [75%] of Availability.]

3.4 Obligated Network Capacity

[Include details of the Obligated Network Capacity.]

	Level	COD	First Regulatory Period	End of First Regulatory Period
Flow rate (Mtpa)	[●]	System Acceptance test to provide evidence based on cost assessment process.	Availability incentive applies	Review of Availability incentive by Regulator.
Overall store capacity (MtCO ₂)	[●]	No further tests beyond FID assurance.	No incentive. Reporting on store performance applies	Store performance incentive to be set up by the Regulator.

Note: level is to be updated by reference to Change in Scope and/or Storage Permit (as applicable), which in either case increases flow rate and/or store capacity.

Schedule 6: Template Form of Approved Project Development Plan

4. Users

[Include details of the assumed Users, with subsections as set out below.]

4.1 Planned Initial Users

[Include details of the Planned Initial Users, i.e. [those announced as part of the [•]].]

4.2 Phase 1 Users

[Include details of the commissioning Users, i.e. Planned Initial Users delivering CO_2 at COD.]

4.3 Phase 2 Users

[Include details of any subsequent Users, i.e. Planned Initial Users commencing CO₂ deliveries post-COD.]

4.4 Potential Phase 3 Users

[Include details of any potential subsequent Users that are envisaged to require expansion of the T&S Network.]

5. Development/expansion of Approved T&S Network

[Phase 3 will comprise such activities beyond Phase 1 and Phase 2. Include details of Phase 3, i.e. development activities relating to the T&S Network with associated On-going Devex during the First Regulatory Period. The Phase 3 activities should be described in the same way as outlined for Phase 1 works and activities based on current assumptions.

Include also information regarding conditions/stage gates to investment/implementation of Changes in Scope, cross referring to the Technical Details Schedule and Financial Settlement Schedule as appropriate.]

Schedule	7:	Qualifying	Acquisition	Information
----------	----	------------	-------------	-------------

No.	Item	Deta	il
1.	Acquirer contact details and related	(a)	number of all persons involved in the Qualifying Acquisition
	notifications	(b)	name of person(s) involved
		(c)	contact details of the person(s)
		(d)	details on any notification to any UK or overseas investment screening regimes that person (or persons) has submitted within the last 12 months
2.	Acquisition details	(a)	the percentage of the shares or voting rights that the person(s) holds in the entity
		(b)	expected date of the Qualifying Acquisition
		(c)	approvals required by UK regulators
		(d)	rationale for the Qualifying Acquisition
3.	Entity details	(a)	name and address of the entity
		(b)	contact details of the authorised individual in entity
		(c)	companies house registration number
		(d)	standard industrial classification SIC code.
		(e)	details on all licences held to operate within the UK.
		(f)	description of dual use item
		(g)	details on entity relationship with the UK Government
		(h)	pre-Qualifying Acquisition structure chart of the entity
		(i)	details on any non-UK government which have a direct or indirect role in the operation or decision making of the entity
4.	Acquiring person(s)	(a)	name, address, and business registration details
	details	(b)	details on any non-UK government, or representative of any non-UK government, have share ownership or voting rights in the person(s)
		(c)	details on any non-UK government which have a direct or indirect role in the operation or decision making of the acquirer
		(d)	when the Qualifying Acquisition completes, details on if there will be any contractual arrangements in place regarding share ownership or voting rights between the person(s) and any other party

No.	Item	Detail
		(e) structure chart of person(s) which includes the % ownership and country of incorporation
5.	Additional information	Other relevant documentation and information on the Qualifying Acquisition.

Annex B: Government Support Package

CCUS: Government Support Package for T&SCo

Introduction and Indicative Heads of Terms

Note: these indicative heads of terms are subject to the "Disclaimer" section at the front of the explanatory note document to which they are annexed.

A. INTRODUCTION

1.1 Background

- (a) The indicative heads of terms in section C (Supplemental Compensation Agreement), section D (Discontinuation Agreement) and section E (Liaison Agreement) of this document set out the basis upon which HM Government ("HMG") would consider providing a contingent Government Support Package ("GSP") to a company licensed to provide transport and storage services ("T&SCo") as part of HMG's CCUS Programme (the "Programme").
- (b) The contents of the heads of terms in section C (Supplemental Compensation Agreement), section D (Discontinuation Agreement) and section E (Liaison Agreement) of this document are indicative only and do not constitute an offer by HMG and do not create a basis for any form of expectation or reliance. Any GSP arrangements that are developed in the future will be subject to approval by Ministers, in consultation with the economic regulator ("Regulator"), and the development and completion of necessary contractual documentation.
- (c) The introduction set out in this section A does not form part of the indicative heads of terms contained in section C (*Supplemental Compensation Agreement*), section D (*Discontinuation Agreement*) and section E (*Liaison Agreement*) of this document and is intended only to provide an overview of the rationale and assumptions for the provision of the GSP.

1.2 Rationale

- (a) The GSP has been structured to cover certain high impact, but low probability risks beyond those which are manageable by operation of the Economic Regulatory Regime ("ERR") and the Revenue Support Agreement ("RSA"), which the investors and/or supply chain, including insurers, of T&SCo cannot take, or cannot price at an efficient level which is good value for money for UK taxpayers, consumers and Users.
- (b) The risks to be covered by the GSP and the detailed terms of coverage remain under consideration, but these heads of terms set out current HMG thinking on the support which could be provided by the GSP. This will depend upon the final terms of the ERR and RSA and the target risk profile.

1.3 Description

- (a) This document comprises of the following:
 - (i) section A: introduction;
 - (ii) section B: conditions precedent to the GSP and other activities pre-Licence Award;

- section C: heads of terms for the supplemental compensation agreement ("Supplemental Compensation Agreement" or "SCA") and the lead insurer agreement ("LIA") at Appendix 2;
- (iv) section D: heads of terms for the discontinuation agreement ("Discontinuation Agreement" or "DA"); and
- (v) section E: heads of terms for the liaison agreement ("Liaison Agreement").
- (b) The scope of the areas of specific protection set out in the indicative heads of terms in section C (Supplemental Compensation Agreement), section D (Discontinuation Agreement) and section E (Liaison Agreement) of this document remains subject to review, in particular in the context of the detail of the ERR and the RSA.

1.4 **Initial assumptions**

1

- (a) This document is based on the following assumptions in respect of the T&S Network:
 - (i) ownership of the T&S Network: the "onshore" (i.e. Onshore Transportation Systems) and "offshore" (i.e. Offshore Transportation and Storage Systems) elements of the T&S Network will be owned by T&SCo. In particular, T&SCo will hold any Crown Estate (including Crown Estate Scotland) lease and licence/permit issued by the NSTA (or licensing authority of a devolved administration);
 - development and operation of the T&S Network: T&SCo will be responsible for the development, construction, operation and maintenance of the T&S Network, including obtaining all necessary permits and approvals;
 - (iii) expansion of the T&S Network: in the future T&SCo may need to expand the T&S Network under the terms of the ERR and will provide access to new Users of the T&S Network in accordance with the terms of its Licence, the CCS Network Code and all relevant laws and regulations;¹
 - (iv) decommissioning: T&SCo will be responsible for the decommissioning and monitoring of:
 - (A) the Offshore Transportation and Storage System in accordance with the Petroleum Act 1998 as applied under the Energy Act 2008 (as may be amended or supplemented) and with regards to post-closure and monitoring obligations, T&SCo will be responsible as set out in its storage permit and licence granted by the NSTA; and
 - (B) the Onshore Transportation System in accordance with the relevant decommissioning conditions of any development/planning consent;
 - (v) T&S charges: Users of the T&S Network will pay T&SCo charges for provision of T&S services which will be regulated under the ERR;
 - (vi) CO₂ ownership: title and risk in CO₂ delivered to the T&S Network will pass to T&SCo at the Delivery Point;

Consideration is being given to changes required to the existing third party access rules as a result of the new regulatory regime.

(vii) RSA: in the development of the revenue model for the T&S Network, HMG has identified that there may be certain demand related revenue risks to T&SCo, where T&SCo's actual revenues may fall short of T&SCo's Allowed Revenue under the ERR. Such demand related revenue risks will be addressed through a number of mitigation measures in the ERR and where the ERR is not sufficient, T&SCo will have access to the RSA as a mechanism which will enable the recovery by T&SCo of its Allowed Revenue in the event of a shortfall.

1.5 **Defined terms**

Capitalised terms not otherwise defined in this document have the meaning given to them in the Licence heads of terms.

B. CONDITIONS PRECEDENT TO THE GSP AND OTHER ACTIVITIES PRE-LICENCE AWARD

No.	Item	Commercial Terms
1.	Conditions Precedent	1.1 It is proposed that the GSP will be conditional on certain conditions precedent including those set out at Appendix 1 to section E (<i>Liaison Agreement</i>).
		Note: conditions precedent to be kept under review.
2.	Procurement of insurances prior to Licence Award	2.1 Prior to Licence Award, T&SCo will use reasonable endeavours to secure commercial insurance in respect of each of the Insured Risks and Insured Losses in accordance with the requirements of the Insurance Schedule appended to the SCA.
		2.2 T&SCo shall ensure that any commercial insurances are placed and maintained with a commercial insurer of Good Standing (as defined in section 26 (<i>Definitions</i>) of section C (<i>Supplemental Compensation Agreement</i>)).
3.	Insurance assurance	3.1 T&SCo shall provide evidence, including confirmation from independent insurance brokers that T&SCo has satisfied its obligation to secure commercial insurances in respect of each of the Insured Risks and Insured Losses pursuant to section 3.3 and section 3.4 of section C (<i>Supplemental Compensation Agreement</i>), save to the extent that it is established that commercial insurances for any such Insured Risks or Insured Losses are Unavailable.
		3.2 The SoS will verify that T&SCo has satisfied its obligation under section 3.1.
		3.3 The SoS will undertake a review of the commercial insurances procured by T&SCo in respect of each of the Insured Risks and Insured Losses to determine whether commercial insurance for any of the Insured Risks or Insured Losses is Unavailable at Licence Award.
4.	Lead Insurer Agreement	4.1 A Lead Insurer Agreement (as set out in Appendix 2 to section C (<i>Supplemental Compensation Agreement</i>)).
		Note : provisions to address the interface with SoS and insurers remain subject to development. Such provisions will be entered into at Licence Award and will set out the relevant requirements and process in respect of claims handling, subrogation and loss adjustment.
5.	T&SCo's financing structure/due diligence	5.1 T&SCo's financing structure will be the subject of due diligence checks for the purpose of assessment against the terms of the Discontinuation Agreement (including, but not limited to, any compensation on termination payable).
		5.2 T&SCo's financing structure shall be in accordance with the Agreed Financing Principles and Approved Hedging Policy,

No.	Item	Commercial Terms
		each as set out in Appendix 1 of section D (<i>Discontinuation Agreement</i>).
6.	Definitions	In this section B (<i>Conditions precedent to the GSP and other activities pre-Licence Award</i>), terms defined in the Licence heads of terms shall have the same meaning, and:
		"Agreed Financing Principles" means the agreed financing principles set out in Appendix 1 of section D (<i>Discontinuation Agreement</i>);
		"Approved Hedging Policy" means the approved hedging policy set out in Appendix 1 of section D (<i>Discontinuation Agreement</i>);
		"GSP" has the meaning given to it in section E (<i>Liaison Agreement</i>);
		"Insured Losses" has the meaning given in section C (Supplemental Compensation Agreement);
		"Insured Risks" has the meaning given in section C (<i>Supplemental Compensation Agreement</i>);
		"Insurance Schedule" means the insurance schedule appended to the SCA;
		"Lead Insurer Agreement" or "LIA" means the agreement to be entered into between the SoS and the lead insurers in respect of, amongst other things, claims handling and subrogation rights as set out at Appendix 2 of section C (<i>Supplemental Compensation</i> <i>Agreement</i>);
		"Licence" means the economic licence to be granted under HM Government's CCUS programme;
		"SCA" means the supplemental compensation agreement; and
		"Unavailable" has the meaning given to it in section C (Supplemental Compensation Agreement).

C. SUPPLEMENTAL COMPENSATION AGREEMENT (SCA)

No.	Heading	Descri	Description		
1.	Parties	1.1	The Parties to the SCA will be:		
			(a) SoS; and		
			(b) T&SCo.		
2.	Commencement and Duration	2.1	The term of the SCA shall commence on and from Licence Award and shall expire:		
			 (a) on revocation of the Licence, where SoS has not exercised its right to discontinue under the Discontinuation Agreement; or 		
			(b) on the date on which a Transfer Scheme (transferring T&SCo's Storage Permit to the transferee) comes into effect, where SoS has exercised its right to discontinue under the Discontinuation Agreement.		
3.	Insurance Schedule	3.1	The SCA will include an insurance schedule which will detail the relevant insurance requirements in respect of each of the Insured Risks and Insured Losses (the "Insurance Schedule"). A draft version of the Insurance Schedule is set out in Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>) and this may be updated from time to time in accordance with section 14 (<i>Variation to Insurance Schedule</i>).		
		3.2	The Insurance Schedule provides the basis for protection for T&SCo under the Supplemental Compensation Agreement – see sections on Excess Loss Compensation and Unavailability Loss Compensation below.		
		3.3	T&SCo has, prior to Licence Award, procured the placement of commercial insurances for each of the Insured Risks and Insured Losses in respect of the "initial period" as set out in paragraph 1.4(a) of the Insurance Schedule, other than in respect of:		
			(a) [<i>specified</i>] Insured Risks; and		
			(b) [<i>specified</i>] Insured Losses,		
			for which commercial insurance is Unavailable at Licence Award.		
		3.4	T&SCo has, prior to Licence Award, procured commitment to provision of commercial insurances for each of the Insured Risks and Insured Losses in respect of the "secondary period" as set out in paragraph 1.4(b) of the Insurance Schedule, other than in respect of:		

No.	Heading	Descriptio	n
		(a	[specified] Insured Risks; and
		(b) [<i>specified</i>] Insured Losses,
			which commercial insurance is Unavailable at Licence vard.
		tei Ris pa	or to Licence Award, T&SCo shall procure under the rms of its commercial insurance that where an Insured sk has occurred and caused an Insured Loss under ragraph 1.3(a) of the Insurance Schedule, any surance proceeds shall be payable:
		(a) to T&SCo, from Licence Award to the date immediately preceding the Discontinuation Date; or
		(b	to SoS, on and from the Discontinuation Date.
		fol	te : loss payee mechanics to be considered further lowing feedback from insurance market on the cash ttlement position.
4.	Other insurances of T&SCo	ot its Ne int pr co CC	e requirements of the SCA are without limitation to any ner insurances that T&SCo elects to procure as part of ordinary course in managing risk in operating the T&S twork, including but not limited to cover for: business erruption; construction risk, employers' liability, ofessional indemnity, credit risk, property/pipeline ver, property damage, cover for delayed start-up and D_2 leakage coverage beyond the requirements of the A.
		ex	te: wider insurance may play an important role, for ample in the context of the Code, the wider obligations the Licence and NSTA financial security.
5.	Unavailability		bject to section 5.2, in relation to any Insured Risks or sured Losses, if at Licence Award or on a Renewal Date:
		(a) commercial insurance is not available in respect of the Project (and any other Licensable Activities in the United Kingdom) in the worldwide insurance market with reputable insurers of Good Standing; or
		(b	the premium, sub-limits or retention under the insurance to cover the relevant Insured Risks or Insured Losses is such that the insurance cover is not commercially viable in accordance with section 5.3, or
		(c)	there has been a material adverse change in the terms (excluding the premium, sub-limits and

No.	Heading	Descrip	tion	
				policy retentions) on which insurance in respect of any Insured Risks or Insured Losses is available,
			Insure	commercial insurance in respect of each such d Risks or Insured Losses shall be regarded as ailable" and Unavailability shall be construed ingly.
		5.2	-	ommercial insurance in respect of Insured Risks Insured Losses shall only be treated as Unavailable
				such Unavailability has not occurred as a result of any criminal activity, fraud or wilful default by T&SCo
				a prudent board of directors of a company undertaking Licensable Activities would, acting reasonably and in the best interests of their company, cease to operate such business as a result of such Unavailability; and
				such Insured Risks and/or Insured Losses are not covered by Commercial Insurances at Licence Award or continue to be covered by Commercial Insurances beyond the Renewal Date.
		5.3	and/or viable premiu directo would best i comme	mmercial insurance for the relevant Insured Risks Insured Losses will be considered not commercially pursuant to section 5.1(b) if the cost of the im or other terms are such that a prudent board of ors of a company undertaking Licensable Activities in similar circumstances, acting reasonably in the interests of the company, resolve that such ercial insurance for the relevant Insured Risks Insured Losses is not commercially viable.
6.	Excess Loss Compensation	6.1	exceed Insura the Ins Policy	event results in claims for Insured Losses that I the Applicable Policy Limit on a Commercial nce, the SoS will meet the amounts in respect of sured Losses which are in excess of the Applicable Limit, up to the Sums Insured set out in the nce Schedule:
				only in respect of the loss which is in excess of the Applicable Policy Limit;
				to the extent such loss would have been recoverable pursuant to the terms and conditions of such relevant Commercial Insurance but for the application of the Applicable Policy Limit; and
			(c)	provided that liability of the provider(s) of the relevant Commercial Insurances (the " Primary

No.	Heading	Description	า		
		pra in a	deter insur settlen ctical the	Trance Provider(s) ") has been agreed or rmined under the terms of the applicable rance policies up to the Applicable Policy Limit. Thent of these claims will follow as far as is terms of the Primary Insurance Providers and ce with the usual market position in respect of urances.	
7.	Unavailability Loss Compensation	7.1 The SoS will provide supplemental compensation protection in the case of the Unavailability of commercial insurance in respect of any Insured Risks and/or Insured Losses up to the Sums Insured set out in the Insurance Schedule.			
		Terms of co	ver		
		con	nmercial	nental compensation for the Unavailability of insurance in respect of any Insured Risks red Losses shall at the SoS's election:	
		(a)	Insu	cate the terms of the relevant Commercial rances (if any) immediately prior to such vailability; or	
		(b)	wher	e:	
			(i)	no Commercial Insurances applied immediately prior to such Unavailability; or	
			(ii)	the SoS considers (acting reasonably) that the Commercial Insurances that applied immediately prior to such Unavailability are no longer appropriate,	
			Insu Insu Una	icate the terms of the other Commercial irances as extended to cover the relevant ired Risks or Insured Losses that are vailable, on terms determined by the SoS ing reasonably).	
		Exclusions			
		crin insu Sch T&S	ninal ac ured part iedule, a	vailability has occurred as a result of any tivity, fraud or wilful default by a named ty under the insurances listed in the Insurance ny loss suffered by that party is excluded from titlement to claim under the SCA in relation to ilability.	
		<u>Defences, e</u>	xclusions	s and retentions	

No.	Heading	Description		
		7.4 The SoS shall be entitled to rely on any defence exclusions and retentions that apply in accordance wit the provisions of section 7.2.		
8.	Subsequent Recovery	8.1 If and to the extent claims paid pursuant to the SCA as subsequently recovered from third parties, the SoS w expect these monies to be returned in full to prever double payment of claims.		
		8.2 The SoS will require the same rights of subrogation as ar relevant Primary Insurance Provider(s).		
		8.3 The Lead Insurer Agreement will address the relationsh with Lead Insurers with regards to the rights subrogation.		
9.	Claims Handling	9.1 Claims handling under the SCA is to be consistent insofa as practicable with usual insurance market practices.		
		9.2 The Lead Insurer Agreement will address the relationsh with Lead Insurers with regards to claims handling ar loss adjustment.		
10.	Failure to maintain Commercial Insurances	10.1 If T&SCo fails to maintain any Commercial Insurances a set out in Appendix 1 to this section C (<i>Supplement Compensation Agreement</i>) (other than as a result of ar Unavailability), then the SoS may either:		
		(a) pay any premiums required to keep th Commercial Insurances in force; or		
		(b) procure the Commercial Insurances itself,		
		and in each case, T&SCo shall be liable to the SoS in respect of any premiums paid by the SoS.		
11.	Renewal of Insurances	11.1 T&SCo undertakes to provide prior notice to the SoS respect of amendments to or the renewal of an Commercial Insurance policies in accordance with section 11.2. To the extent Commercial Insurances are amended or renewed in the absence of SoS approval, the SoS sharnot be liable to pay any amount of Excess Los Compensation (see section 6) or Unavailability Los Compensation (see section 7) which would not have bee incurred had such amendments to the relevant insurance policies not been made.		
		11.2 The process around renewals of the Commerci Insurance will involve:		
		(a) T&SCo giving not less than [sixty (60)] Busines Days' notice to SoS of T&SCo's proposals in relation		

No.	Heading	Description	
			to the Commercial Insurances prior to their renewal or expiry.
		(b)	The notice pursuant to section 11.2(a) shall include an outline of T&SCo's proposals on:
			 the changes to the Commercial Insurances since the date of the SCA or the previous Renewal Date; and
			 (ii) any proposed changes in the settlement value of all assets required to be insured, insured amounts in respect of all risks insured, the amount of retentions and the applicable policy limit and any claims experience in respect of the existing insurance.
		(c)	T&SCo shall then give not less than [thirty (30)] Business Days' notice to the SoS of the full details of T&SCo's proposals in relation to the Commercial Insurances prior to their renewal or expiry.
		(d)	SoS will have an opportunity to comment on the outline of the insurance proposals notified by T&SCo pursuant to section 11.2(a) and also the full details of the proposals notified by T&SCo pursuant to section 11.2(c).
		(e)	SoS will have [ten (10)] Business Days from receipt of notice of T&SCo's full insurance proposals pursuant to section 11.2(c) to consider whether it agrees or not with the full details of the insurance proposals notified by T&SCo pursuant to section 11.2(c) and SoS should not be entitled to unreasonably withhold or delay its consent to the proposals where such insurances are to be renewed on terms consistent with the Insurance Schedule.
		(f)	If the SoS does not consent to the insurance proposals provided by T&SCo (acting reasonably pursuant to section 11.2(e)) or if T&SCo amends the terms of the insurances without the consent of SoS, T&SCo may continue to place the insurances but the SoS shall not be obliged to pay any amount of Excess Loss or Unavailability Loss to the extent that such loss would not have been incurred but for T&SCo placing or amending the insurance on such terms.
			Note: as a general note, time periods in these heads of terms remain under consideration by ESNZ.

No.	Heading	Descri	ption			
12.	T&SCo insurance market testing	12.1	T&SCo shall undertake testing of the commercial insurance market as part of the process for renewal of Commercial Insurances pursuant to section 11 (<i>Renewal</i> <i>of Insurances</i>) to assess whether commercial insurances are available in respect of any Insured Risks or Insured Losses that were previously Unavailable. 11.2(c).			
		12.2	At the SoS's request (such requests to be made no more than once in any twelve (12) month period and only where the SoS considers that commercial insurances may be available in respect of any Insured Risks or Insured Losses that were previously Unavailable), T&SCo shall undertake additional testing of the commercial insurance market.			
13.	Insurance assurance	13.1	13.1 T&SCo shall provide an independent insurance broker's undertaking to the SoS (in a form and substance satisfactory to the SoS) on each Renewal Date, confirming that T&SCo has sufficiently tested the commercial insurance market in respect of each of the Insured Risks and Insured Losses and any Unavailability of commercial insurances.			
		13.2	Following receipt of the independent insurance broker's undertaking pursuant to section 13.1, the SoS will satisfy itself that adequate and suitable quantum and coverage has been sought effectively by T&SCo and, where relevant has been obtained, or is otherwise Unavailable.			
		13.3	The SoS may, at its own discretion, seek independent advice and request further information in respect of the undertaking provided by T&SCo pursuant to section 13.1 and T&SCo shall promptly comply with any such further information requests made by the SoS.			
14.	Variation to Insurance Schedule	14.1	Subject to section 14.2, the SoS shall be entitled, on notice to T&SCo, to vary or amend the Insurance Schedule, including the Insured Risks and Insured Losses.			
		14.2	The SoS shall not make any amendment to the Insurance Schedule if:			
			(a) such amendment would materially and adversely affect:			
			(i) the risk profile of T&SCo and/or			
			(ii) the ability of T&SCo to operate the T&S Network,			
			and T&SCo has provided evidence, to the satisfaction of the SoS, that this would be the effect of such amendment; and/or			

No.	Heading	Descri	ption			
			 (b) the SoS does not have the legal power or capacity to require the implementation of such an amendment, unless the SoS has obtained the prior written consent of 			
			T&SCo.			
		14.3	Subject to section 14.4, T&SCo shall be entitled to request a variation or amendment to the Insurance Schedule, including the Insured Risks and Insured Losses, and such request shall be supported by evidence and an insurance broker's undertaking.			
		14.4	SoS shall consider whether or not to implement such request and shall notify T&SCo of its decision whether to accept, request amendments or reject such request from T&SCo. This decision shall be at the sole discretion of the SoS.			
		14.5	Any variation or amendment to the Insurance Schedule, including to the Insured Risks and Insured Losses, made pursuant to this section 14 shall take effect on and from the next Renewal Date following notice by the SoS to T&SCo, subject to such notice being given by the SoS to the T&SCo no later than [sixty (60)] Business Days prior to the next Renewal Date.			
15.	Consequences of Expiry	15.1	Where the SCA expires pursuant to section 2.1, this will be without prejudice to the SoS's obligation to pay supplemental compensation for third party liabilities and claims:			
			 (a) in respect of events arising prior to or on the date of expiry of the SCA; and 			
			(b) notified to the SoS no later than, [three (3) years] after the date of expiry of the SCA.			
16.	Utmost Good Faith	16.1	T&SCo acknowledges and agrees that it owes a duty of utmost good faith to the SoS.			
17.	SCA Fee	17.1	The SoS will charge fees to T&SCo for support under the SCA (the "SCA Fee").			
			Note:			
			• Methodology of calculation of the SCA Fee remains under consideration.			
			• The SCA Fee will reflect the pricing elements used on the commercial insurance market and will be based on a conservative estimate of the premium			

No.	Heading	Description
		 that T&SCo would be charged on the commercial insurance market (potentially plus a buffer). Consideration is being given to the SCA Fee including both a "commitment" fee and an "arrangement" fee. The commitment fee would provide for standby cover to address changes in availability of coverage on the commercial market. The arrangement fee would cover the costs of arranging SCA cover. Consideration is also being given to the level of retention applied. This should, as a minimum, be based on an assessment of the market position. 17.2 On each Renewal Date, the SoS shall have the right to adjust the SCA Fee.
18.	Undertakings	 18.1 T&SCo will provide certain information and other undertakings to the SoS in support of its obligations to maintain Commercial Insurances and other obligations under the SCA.
19.	Dispute Resolution Procedure	19.1 Any dispute arising out of or in connection with the SCA, including in relation to any amounts payable under or pursuant to the SCA (a "Relevant Dispute"), shall be resolved in accordance with the Dispute Resolution Procedure set out in this section 18.
		19.2 In the event of a Relevant Dispute, the matter shall first be referred to the Liaison Committee for resolution by the Parties.
		19.3 If the Relevant Dispute is not resolved by the Liaison Committee after a period of [ten (10)] Business Days on and from such referral (or such other period as agreed between the Parties in writing), either Party shall be entitled to refer the Relevant Dispute to identified senior representatives for resolution.
		19.4 If the Relevant Dispute is not resolved by the senior representatives after a further period of [ten (10)] Business Days on and from such referral to identified senior representatives (or such other period as agreed between the Parties in writing), then:
		 (a) in the case of disputes arising pursuant to section 11 (<i>Renewals of Insurances</i>) or section 12 (<i>T&SCo</i> <i>insurance market testing</i>), either Party may refer the dispute for expert determination in accordance with the following procedure:

No.	Heading	Descrip	tion		
				(i)	the Parties shall first seek to agree the appointment of an expert within [five (5) Business Days];
				(ii)	if the Parties are unable to agree on the appointment of an expert within those [five (5) Business Days], the Parties agree that an expert shall be appointed by [the chair of the Association of British Insurers (ABI)/the Chartered Insurance Institute]; and
				(iii)	the Relevant Dispute shall be referred to the expert for determination; and
			(b)	of an sectio	case of any other disputes (and any appeal expert determination made pursuant to n 19.4(a)), either Party may refer the ant Dispute to the courts of England and s.
20.	Representatives and personnel	20.1			e representatives of the Parties are to be ncluding delegated authority.
		20.2	T&SC	o repre	sentative shall:
			(a)		ge and oversee compliance by T&SCo with its and obligations under the SCA; and
			(b)	exerci	ise any right and perform any obligation isable by or to be performed by the T&SCo sentative under the SCA.
		20.3	SoS re	epreser	ntative shall:
			(a)		ge and oversee compliance by SoS with its and obligations under the SCA; and
			(b)	exerci	ise any right and perform any obligation isable by or to be performed by the SoS sentative under the SCA.
21.	Assignment and Sub-Contracting	21.1	assigr	nees an	all benefit and bind Parties, their permitted d respective successors and any reference in ny party shall be construed accordingly.
		21.2	transf the S writte unrea be per its sec	er by T CA (in n cons sonably rmitted cured ci	include restrictions on the assignment and/or &SCo of any of its rights or obligations under whole or in part except) without the prior sent of the SoS (such consent not to be y withheld or delayed), save that T&SCo will to create a security assignment in favour of reditors (and SoS will assist in facilitating this, t all costs and expenses properly incurred by

No.	Heading	Description
		SoS in giving effect to such assignment are paid by T&SCo).
		21.3 The SoS may, subject to the conditions to be set out in the SCA, transfer its rights or novate its obligations under the SCA to any Minister of the Crown or any entity directly wholly-owned by a Minister of the Crown.
22.	Confidentiality	22.1 The SCA will set out confidentiality provisions which will apply to all Parties in respect of the Project. There will be customary exceptions.
		22.2 The SCA sets out that the SoS is, and that T&SCo may become subject to, the requirements of the FOIA and the Environmental Information Regulations and each Party shall facilitate compliance by the other Party with its information disclosure requirements pursuant to the FOIA and the Environmental Information Regulations.
23.	Survivorship	23.1 Following the expiry of the SCA, each Party shall cease to have rights or obligations under the SCA, save that the rights in respect of claims under the SCA for Insured Losses in respect of any Insured Risk that occurred prior to expiry of the SCA, under section 6 (<i>Excess Loss Compensation</i>), section 7 (<i>Unavailability Loss Compensation</i>), section 8 (<i>Subsequent Recovery</i>), section 15 (<i>Consequences of Expiry</i>), section 19 (<i>Dispute Resolution Procedure</i>), section 22 (<i>Confidentiality</i>) and section 25 (<i>Governing Law</i>) will continue in full force and effect.
24.	Rights of Third Parties	24.1 A person who is not a Party to the SCA shall have no rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any of its terms.
25.	Governing Law	25.1 The SCA shall be governed by and construed in accordance with English law and the courts of England and Wales shall have exclusive jurisdiction.
26.	Definitions	In this section C (<i>Supplemental Compensation Agreement</i>), terms defined in the Licence heads of terms shall have the same meaning and:
		"Applicable Policy Limit" means, in respect of each of the Commercial Insurances, the applicable policy indemnity limits for such Commercial Insurance;
		"Approved T&S Network" has the meaning given to it in the Licence heads of terms, as at Licence Award;
		"Commercial Insurances" means the commercial insurances placed by T&SCo in respect of each of the Insured Risks and Insured Losses;

No.	Heading	Description
		"Corrective Measures" has the meaning given to that term in the Carbon Dioxide Storage Regulations;
		"Discontinuation Date" has the meaning given in section D (<i>Discontinuation Agreement</i>);
		"Discontinue" has the meaning given in section D (<i>Discontinuation Agreement</i>);
		"Dispute Resolution Procedure" means the procedure for the resolution of disputes set out in section 19 (<i>Dispute Resolution Procedure</i>) of this section C (<i>Supplemental Compensation Agreement</i>);
		"Environmental Information Regulations" means the Environmental Information Regulations 2004 (as amended);
		"Excess Loss Compensation" means the compensation payable by SoS on the terms set out at section 6 (<i>Excess Loss</i> <i>Compensation</i>) of this section C (<i>Supplemental Compensation</i> <i>Agreement</i>);
		"FOIA" means the Freedom of Information Act 2000 (as amended);
		"Good Standing" means in relation to a commercial insurer that such insurer is:
		 a UK authorised insurer (that is, regulated by the Prudential Regulation Authority or any successor body) or European Economic Area (EEA) authorised insurer operating in the UK); and
		 (ii) an issuer that has a long-term rating of a minimum of A- by S&P Global Ratings (Standard & Poor's) or a minimum of A3 by Moody's Investors Service, or an equivalent rating by another recognised ratings agency;
		"Insurance Schedule" has the meaning given to it in Appendix 1 (<i>Insurance Schedule</i>) of this section C (<i>Supplemental Compensation Agreement</i>);
		"Insured Losses" has the meaning given in Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>);
		"Insured Risks" has the meaning given in Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>);
		"Lead Insurer Agreement" or "LIA" means the agreement to be entered into between the SoS and the lead insurers in respect of, amongst other things, claims handling and subrogation rights

No.	Heading	Description
		as set out at Appendix 2 of section C (<i>Supplemental Compensation Agreement</i>);
		" Lead Insurers " has the meaning given to it in Appendix 2 of this section C (<i>Supplemental Compensation Agreement</i>);
		"Leakage" means any release of CO_2 from the Storage Complex;
		"Liaison Committee" has the meaning given to it in Section E (<i>Liaison Agreement</i>);
		"Licence" means the economic licence to be granted under HM Government's CCUS programme;
		"Licensable Activities" means such activities described at section 2 of the Act;
		"Minister of the Crown" shall have the meaning given to it in the Minister of the Crown Act 1975;
		"Post Closure Period" has the meaning given to that term in the Carbon Dioxide Storage Regulations;
		" Primary Insurance Provider(s) " has the meaning given to it in section 6 (<i>Excess Loss Compensation</i>) of this section C (<i>Supplemental Compensation Agreement</i>);
		" Principal Exclusions " are as set out in paragraph 1.10 of Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>);
		"Relevant Dispute" has the meaning given to it in section 19.1 (<i>Dispute Resolution Procedure</i>) of this Section C (<i>Supplemental Compensation Agreement</i>);
		"Renewal Date" means a renewal date in respect of any Commercial Insurances or any date on which T&SCo undertakes market testing of the commercial insurance market pursuant to section 12 (<i>T&SCo insurance market testing</i>) of this section C (<i>Supplemental Compensation Agreement</i>);
		"SCA Fee" has the meaning given to it in section 17.1 (<i>SCA Fee</i>) of this Section C (<i>Supplemental Compensation Agreement</i>);
		"Significant Irregularity" has the meaning given in Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>);
		"Specified Significant Irregularity" or "SSI" has the meaning given in Appendix 1 (<i>Insurance Schedule</i>) to this section C (<i>Supplemental Compensation Agreement</i>);
		"Sums Insured" means the sums set out at paragraph 1.5 of section 3 (<i>Insurance Schedule</i>) of Appendix 1 (<i>Insurance</i>

No.	Heading	Description
		<i>Schedule)</i> to this section C (<i>Supplemental Compensation Agreement</i>);
		"Storage Complex" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Complex is identified as part of the Approved T&S Network;
		"Storage Permit" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Permit relates to the Approved T&S Network;
		"Storage Site" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Site is identified as part of the Approved T&S Network;
		"Transfer Scheme" means a transfer scheme instituted by the SoS in respect of the T&S Network under section [50] of the Act;
		"T&SCo Successor" means the successor entity to T&SCo pursuant to a Transfer Scheme;
		"Unavailable" has the meaning given to it in section 7 (<i>Unavailability Loss Compensation</i>) of this section C (<i>Supplemental Compensation Agreement</i>); and
		"Unavailability Loss Compensation" has the meaning given to it in section 7 (<i>Unavailability Loss Compensation</i>) of this section C (<i>Supplemental Compensation Agreement</i>).

Appendix 1 to Section C (*Supplemental Compensation Agreement*): Draft Insurance Schedule

Insurance Schedule

Note: the SCA heads of terms and the Insurance Schedule in particular, remain subject to consideration and engagement with the commercial insurance market and ongoing diligence by technical advisers.

1.1 Insured

- (a) T&SCo;
- (b) the Secured Creditors;
- (c) the sub-contractors of T&SCo;
- (d) the regulators (Ofgem, the North Sea Transition Authority (NSTA) and Offshore Petroleum Regulator for Environment and Decommissioning (OPRED), the Health and Safety Executive (HSE));
- (e) the SoS;
- (f) suppliers, manufacturers, professional consultants and engineers for their site activities only in connection with the carrying out of works; and
- (g) employees, directors and officers of the above,

each for their respective rights and interests.

1.2 Insured Risks

Insured Risks means any of the following risks:

- (a) physical damage to infrastructure assets, including that attributable to machinery breakdown, at a Storage Site; or
- (b) any physical damage to a well (including any legacy well) or a well blowout at a Storage Complex; or
- (c) geological conditions at a Storage Complex,

which, in each case, results in:

(A) during the period of injection of CO_2 at that Storage Site or Storage Complex:

- Leakage, that obliges T&SCo under its Storage Permit to suspend the injection of CO₂ at that Storage Site (including where such suspension results from Corrective Measures or a direction from NSTA); and/or
- (ii) in the case of paragraph (c), a Specified Significant Irregularity that obliges T&SCo under its Storage Permit to suspend the injection of CO₂ at that Storage Site for a minimum period of three (3) years (including where such suspension results from Corrective Measures or a direction from NSTA), or

(B) during the Post Closure Period at that Storage Site, any Leakage or Specified Significant Irregularity at that Storage Complex.

(the "Insured Risks").

Note: definition of Insured Risks subject to further insurance/technical input.

The Insured Risks may be caused either by a single event or may emerge over a longer period of time through monitoring of the Storage Complex.

"Significant Irregularity" means any irregularity in the injection or storage operations at the Storage Site or in the condition of the Storage Complex, which implies the risk of a Leakage and/or risk to the environment or human health;

"Specified Significant Irregularity" or **"SSI"** means a Significant Irregularity which implies the risk of a Leakage and which arises from one of the following implied risks:

- unexpected, measured or interpreted pressures and/or plume locations that imply the risk of failure to geological containment of CO2 (including seals, cap rocks, faults, adjoining aquifers or other means of geological containment) at that Storage Complex, when operating within the permitted operating limits within the Storage Permit;
- (ii) any seismicity that implies the risk of Leakage at that Storage Complex;
- (iii) any geological aseismic movement caused by CO2 injection at that Storage Complex that implies the risk of Leakage at that Storage Complex; and/or
- (iv) an environmental event caused by CO2 injection at that Storage Complex (which may additionally include significant unplanned leakage of brine, heavy metals, oil, gas, H2S, and/or other toxins) that implies the risk of Leakage at that Storage Complex.

1.3 Insured Losses

Insured Losses means the following costs, losses and liabilities arising from the Insured Risks (up to the Sums Insured and subject to the Principal Exclusions):

- (a) costs resulting from (as applicable):
 - (i) reinstatement of infrastructure at the Storage Site, by way of implementation of Corrective Measures or a direction from NSTA; or
 - (ii) reinstatement of a well (including any legacy well) or control of a well blow out at the Storage Complex, by way of implementation of Corrective Measures or a direction from NSTA; or
 - (iii) geological remediation of the Storage Complex, by way of implementation of Corrective Measures or a direction from NSTA,

provided that, in the case of abandonment of the Storage Site, the cash settlement value at paragraph 1.9(a) shall apply;

 (b) delay in start-up/business interruption costs (covering loss of Use of System Charges) in the event that suspension of injection of CO₂ at the Storage Site (as specified in paragraph 1.2(A)) is required;

Note: *limb* (*b*) *remains under review in the context of consideration of the interplay between business interruption insurance and the Licence, RSA, GSP and Code. Provisions reflecting the following principles are to be incorporated:*

(i) T&SCo to be under an obligation to procure asset damage insurance and associated BI insurance to the extent available;

(ii) Code to provide for suspension of T&S charges to the extent covered by the BI insurance procured pursuant to the above obligation; and

(iii) RSA to treat amounts recoverable under the BI insurance procured pursuant to the above obligation as Market Revenues.

Detailed drafting to be developed, including scoping of procurement obligation and definition of amounts recoverable.

Note: position regarding use of CO2 for commissioning remains under review with NSTA.

- (c) any environmental remediation costs necessary to comply with relevant legal requirements at the time of the occurrence of the Insured Risk and/or any Storage Permit obligations, including costs associated with clean-up and third-party damage; and
- (d) costs relating to the purchase of UK ETS allowances in respect of Leakage,

(the "Insured Losses").

Note: definition of Insured Losses subject to further insurance/technical input

1.4 **Period**

(a) Initial Period – Construction and Commissioning

The initial period of cover will run from Licence Award and will remain in effect during the Construction Phase and Commissioning Phase, ending on the Commercial Operations Date. The initial period of cover will not require cover for Insured Risks that result in Specified Significant Irregularities.

(b) Secondary Period – First phase of operations

The secondary period of cover will run from the Commercial Operations Date up to the end of the First Regulatory Period (unless otherwise required by SoS to achieve an economic and efficient outcome).

(c) Subsequent Periods

The subsequent periods of cover (with renewals at each Periodic Review under the Licence unless otherwise required to achieve an economic and efficient outcome) will run:

- (i) during the period from the end of the First Regulatory Period until start of the Post Closure Period at the Storage Site; and
- (ii) during the Post Closure Period until termination pursuant to The Storage of Carbon Dioxide (Termination of Licences) Regulations 2011.

During the Post Closure Period the following elements of this Insurance Schedule will cease to apply:

(i) settlement costs in the event of total abandonment of the Storage Site pursuant to paragraph 1.3(a); and

(ii) delay in start-up/business interruption pursuant to paragraph 1.3(b).

1.5 Sums Insured

The Sums Insured for the purposes of Excess Loss Compensation and Unavailability Loss Compensation in respect of the Insured Losses are as follows:

Insured Loss	Reference	Sum insured (indexed)
Infrastructure remediation	1.3(a)	Reinstatement / remediation costs
Geological remediation	1.3(a)	[To be specified]
Business interruption	1.3(b)	[To be specified]
Environmental liabilities	1.3(c)	[To be specified]
Carbon UK ETS liabilities	1.3(d)	[To be specified]

Note: position on indexation of Sums Insured subject to confirmation by ESNZ.

1.6 **Geographical Limits**

Geographical limits for the SCA insurance requirement relate to the management of CO_2 leakage risk offshore, specifically arising at the Storage Complex [which is to include the precise location and delimitation of the Storage Site and the Storage Complex].

1.7 Retentions

Insurance retentions relate to the first loss amount borne by the T&SCo in the event an Insured Risk occurring. These retentions should align with the market standard.

Where there is an opportunity for the T&SCo to decide upon the level of premia versus the level of retentions, there will be pre-agreed parameters determined upfront with SoS.

1.8 **Principal Extensions**

Any extension to the cover in this Insurance Schedule should be priced separately unless there is no differential.

1.9 **Principal Inclusions**

Principal inclusions for cover will include the following:

- (a) cash settlement clause in respect of abandonment of the Storage Site referenced at paragraph 1.3(a); and
- (b) market standard non-vitiation clause/innocent insured clause.

1.10 **Principal Exclusions**

Principal exclusions from cover will include the following:

- (a) any fines and penalties borne by T&SCo;
- (b) tangible environmental damage or environmental remediation costs to third parties beyond what is necessary to comply with relevant legal requirements and Storage Permit requirements;
- (c) any normal operating costs incurred by T&SCo, including operating, maintenance

(whether preventative or reactive which should be carried out as a matter of good industry practice) and management costs associated with the operation of the T&S Network in the Operational Phase and, in each case as eligible for assessment as part of the T&SCo's Allowed Revenue;

- (d) any costs related to constraints on the level of CO₂ injection or the capacity of the Storage Complex either before or after (i) any Leakage or (ii) a Specified Significant Irregularity at the Storage Complex, has occurred, where such constraints do not amount to suspension of CO₂ injection at the Storage Site (as specified in paragraph 1.2(A)); and
- (e) any costs or liabilities of Users arising from the Insured Risks.

1.11 Loss Payee

Where an Insured Risk has occurred and caused an Insured Loss under paragraph 1.3(a) of this Insurance Schedule, any amount shall be payable:

- (a) to T&SCo, from Licence Award to the date immediately preceding the Discontinuation Date; or
- (b) to SoS, on and from the Discontinuation Date.

Note: loss payee mechanics to be considered further following feedback from insurance market on the cash settlement position.

Appendix 2 to Section C (*Supplemental Compensation Agreement*): Lead Insurer Agreement heads of terms

Note: provisions to address the interface with SoS and insurers remain subject to development and subject to insurance market feedback.

No.	Heading	Descri	ption	
1.	Parties	1.1	The Parties to the LIA will be:	
			(a) SoS; and	
			(b) Lead Insurers.	
2.	Purpose	2.1	The purpose of the LIA shall be to govern the relationship between the Parties in respect of the sharing of information and the assessment and settlement of claims for which the SoS may be liable in accordance with the terms of the SCA.	
		2.2	It is anticipated that each Lead Insurer in respect of the Insurances will put arrangements in place with the Insurers to allow that Lead Insurer to comply with its obligations under this Agreement.	
3.	Commencement and Duration	3.1	The term of the LIA shall commence on and from Licence Award and shall expire on expiry of the SCA in accordance with its terms.	
4.	Noting of SoS's interest	4.1	The Lead Insurers shall acknowledge the SoS's role and interest as provider of contingent financial support under the SCA in respect of:	
			(a) Excess Loss; and	
			(b) claims where a risk covered by an Insurance becomes Unavailable,	
			in each case as set out in the SCA.	
5.	Acknowledgment of	5.1	Each Lead Insurer shall, in respect of the Insurances:	
	the LIA by all Lead Insurers		 (a) ensure that each Insurer in that Lead Insurer's layer of the Insurance from time to time is given a copy of the LIA; and 	
			(b) use reasonable endeavours to procure that each Insurer acknowledges the terms of the LIA by signing an acknowledgment and returning it to relevant Lead Insurer who shall then provide a copy to the SoS.	
6.	Sharing of information, reports and notices	6.1	The Lead Insurers for the Insurance shall (directly or through the project appointed broker or loss adjuster) promptly, and in any event within [ten (10)] Business Days of becoming aware, notify the SoS whether in its or their reasonable opinion any claim or series of claims in	

No.	Heading	Descri	ption			
			actua Applio cover	il) is lik cable P · provic	the Insurances (in each case anticipated or ely to exceed (a) [fifty per cent (50%)] of the olicy Limit for such Insurance and/or (b) the ded by the relevant Lead Insurer's layer of nce (a "Threshold Claim").	
		6.2	Each Lead Insurer, directly or through a nominated third party, shall provide the SoS with a quarterly summary of all claims, decisions, determinations, settlements and payments under the Insurances or layer of Insurance (as the case may be) in respect of which it is the Lead Insurer.			
		6.3	If requested by the SoS, each Lead Insurer shall, as soon as reasonably practicable provide the SoS with any formal risk, survey or other report, and any other information referred to in section 6.4 in relation to the Insurances, provided that:			
			(a)	such	egal professional privilege attaching to any information shall not be waived by the sion of such information;	
			(b)	strictl confic LIA a	onfidentiality of any such information shall be ly maintained in accordance with the dentiality provisions that will be set out in the and shall not be disclosed without the prior en consent of the Lead Insurer; and	
			(c)	where	e any report or advice requested by the SoS:	
				(i)	relates solely to a Disputed Claim, the Lead Insurers shall be under no obligation to provide such report or advice to the SoS; or	
				(ii)	refers to a Disputed Claim but covers topics or information in addition to the relevant Disputed Claim, the Lead Insurers may redact the relevant report or advice to the extent it relates to the Disputed Claim from the copy provided to the SoS.	
		6.4	section report by m and a	on 6.3 ts or e ore tha any sit	information that may be requested under may include expert reports, loss adjuster xternal legal advice (available to or received an one Insurer), risk surveying programmes re issues raised by any Lead Insurer's risk connection with the Project.	
		6.5	advic Insur	e in c er shal	ad Insurer obtains or procures any reports or connection with the Insurances, that Lead I use reasonable endeavours to ensure that ntitled to rely on such reports or advice.	

No.	Heading	Descri	ption	
7.	Appointments of loss adjusters	7.1	The Lead Insurer shall notify the SoS of any proposed appointment of any loss adjuster, other expert or adviser in relation to any Insurance	
		7.2	The Lead Insurer shall, at the request of the SoS, ensure that the appointment allows the SoS to:	
			(a) access the advice and reports prepared by the relevant loss adjuster, expert or adviser; and	
			(b) rely upon that appointment,	
			in each case on the terms set out in section 12 (<i>Costs and Expenses</i>).	
		7.3	Where an advice or report of any loss adjuster, expert or adviser referred to above:	
			(a) relates solely to a Disputed Claim, the Lead Insurers shall be under no obligation to provide such report or advice to the SoS; or	
			(b) refers to a Disputed Claim but covers topics or information in addition to the relevant Disputed Claim, the Lead Insurers may redact the relevant advice and report to the extent it relates to the Disputed Claim from the copy provided to SoS.	
8.	Participation in insurance meetings by the SoS	8.1	The Lead Insurers shall invite the SoS to, and the SoS shall be entitled to attend and participate in, all reporting, monitoring and claims related meetings and conference calls (other than any internal meetings or calls held by individual Insurers) relating to the Insurances.	
		8.2	To the extent that a meeting or conference call referred to in section 8.1 relates to a Disputed Claim, the SoS shall not be entitled to attend and participate in the part of the meeting in which the attendees discuss the Disputed Claim.	
9.	Notification of Claims	9.1	In relation to any Threshold Claim, the relevant Lead Insurer shall provide to the SoS:	
			 (a) any notices received or issued by the relevant Lead Insurer, including in respect of any notices received by other Insurers in the relevant Lead Insurer's layer that have been copied to the Lead Insurer; 	
			 (b) details relating to the appointment of claims handlers and loss adjusters, lawyers and other experts or advisers; 	

No.	Heading	Descri	escription		
			(c)	any reports commissioned or received by the relevant Lead Insurer; and	
			(d)	information on claim assessments, any proposed settlement and actual settlements.	
		9.2	each	ation to any other claim in relation to the Insurances, Lead Insurer shall promptly supply the information It above to the SoS upon the SoS's request.	
		9.3		ation to any Threshold Claim under the Insurances, of the Lead Insurers shall:	
			(a)	following any decision or determination in respect of settlements or payments under the Insurances, promptly notify the SoS in writing of such decision or determination, including details as to quantum; and	
			(b)	following any payment under the Insurances, promptly notify the SoS in writing of the details of such payments.	
10.	Conduct of proceedings in respect of Threshold Claims	10.1	shall, consu	lation to any Threshold Claim, the Lead Insurers ensure that any appointed loss adjuster shall, It with the SoS in relation to the initiation and ict of any proceedings.	
11.	Subrogation	11.1	The Lead Insurers shall consult with the SoS in relation to any proposed exercise of subrogation rights in relation to the Insurances and the process of any exercise of subrogation rights which have been instituted.		
12.	Costs and expenses	12.1	The SoS shall not be required to pay any amounts or fees in relation to the provision to it of any information under the LIA or for the SoS's attendance and participation in meetings under the LIA.		
		12.2	Subject to section 12.3, in relation to the costs and expenses incurred with respect to any loss adjuster(s) and any other experts or advisers that have been appointed by the SoS and a Lead Insurer, such costs and expenses shall be shared between the SoS proportionally to the ratio that the Insurers' liability incurred under the Insurances bears to the SoS's liability incurred under the SCA for the relevant claim(s).		
		12.3	and releva that tl	SoS instructs a loss adjuster appointed by the SoS a Lead Insurer to undertake activities and the ant Lead Insurer(s) do not (acting reasonably) agree he relevant activities are necessary, the SoS shall be nsible for the costs and expenses specified in section	

No.	Heading	Descri	ption
			12.2 that are incurred solely in connection with such activities.
13.	Accession of a replacement or additional Lead Insurer	13.1	Where a replacement or additional Lead Insurer is appointed for any Insurance after the date of the LIA, the Parties acknowledge that the replacement or additional Lead Insurer shall be required to execute a deed of adherence to accede to the terms of the LIA.
14.	Removal of Lead Insurer	14.1	Where a Lead Insurer ceases to be a Lead Insurer after the date of the LIA, the relevant Lead Insurer shall be released from its obligations to the SoS under this Agreement, without prejudice to any existing claims or liabilities of that Lead Insurer as at the date they cease to be a Lead Insurer.
15.	Confidentiality	15.1	The LIA will set out confidentiality provisions which will apply to all Parties in respect of the Project. There will be customary exceptions.
		15.2	The LIA sets out that the SoS is, and that Lead Insurers may become subject to, the requirements of the Freedom of Information Act ("FOIA") and the Environmental Information Regulations and each Party shall facilitate compliance by the other Party with its information disclosure requirements pursuant to the FOIA and the Environmental Information Regulations.
16.	Assignment	16.1	The LIA shall benefit and bind Parties, their permitted assignees and respective successors and any reference in the SCA to any Party shall be construed accordingly.
		16.2	The LIA will include restrictions on the assignment and/or transfer by T&SCo of any of its rights or obligations under the LIA (in whole or in part except) without the prior written consent of the SoS (such consent not to be unreasonably withheld or delayed), save that T&SCo will be permitted to create a security assignment in favour of its secured creditors (and SoS will assist in facilitating this, provided that all costs and expenses properly incurred by SoS in giving effect to such assignment are paid by T&SCo).
		16.3	The SoS may, subject to the conditions to be set out in the LIA, transfer its rights or novate its obligations under the LIA to any Minister of the Crown or any entity directly wholly-owned by a Minister of the Crown.
17.	Dispute Resolution Procedure	17.1	The courts of England and Wales shall have exclusive jurisdiction to settle any dispute arising out of or in connection with the LIA.

No.	Heading	Description				
18.	Governing Law	18.1 The LIA shall be governed by and construed in accordance with English law and the courts of England and Wales shall have exclusive jurisdiction.				
19.	Definitions	"Applicable Policy Limit" means, in respect of each Insurance, the applicable policy indemnity limits for such Insurance set out in paragraph 1.5 (<i>Sums</i> Insured) of Appendix 1 (<i>Draft Insurance</i> <i>Schedule</i>) to the SCA, as those policy indemnity limits may be amended from time to time;				
		"Disputed Claim" means a claim under the Insurances made by a named insured for which:				
		 (a) the Insurers have denied indemnity and liability under the Insurances to the relevant named insured; and 				
		 (b) such named insured is disputing or challenging the relevant denial of indemnity and liability under the Insurances; 				
		"Environmental Regulations" means the Environmental Information Regulations 2004 (as amended);				
		"FOIA" means the Freedom of Information Act 2000 (as amended);				
		"Insurances" means the insurance policies referred to in Insurance Schedule appended to the SCA heads of terms;				
		"Insurers" means, in respect of each Lead Insurer, any insurers within its layer of the Insurances and/or any re-insurers;				
		"Lead Insurers" has the meaning given to that term in section 1 (<i>Parties</i>), being the lead insurers for each layer of each of the Insurances (as amended from time to time by section 13 (<i>Accession of a replacement or additional Lead Insurer</i>) and section 14 (<i>Removal of Lead Insurer</i>)), as listed as at the date of the LIA in a schedule and each of the Lead Insurers shall be a "Lead Insurer";				
		"Minister of the Crown" shall have the meaning given to it in the Minister of the Crown Act 1975;				
		"Project" means the CCUS project;				
		"SCA" means the supplemental compensation agreement; and				
		"Threshold Claim" has the meaning given to it in section 6 (<i>Sharing of information, reports and notices</i>) of this heads of terms.				

D. DISCONTINUATION AGREEMENT (DA)

No.	Heading	Descri	ption
1.	Parties	1.1	The Parties to the Discontinuation Agreement will be:
			(a) SoS; and
			(b) T&SCo.
			Note: ESNZ would anticipate that there would be provision for assignment of the Discontinuation Agreement to the Security Trustee under T&SCo's security arrangements – it is therefore not proposed that the security trustee is party to the Discontinuation Agreement.
2.	Commencement and Duration	2.1	Subject to section 23 (<i>Survivorship</i>), the term of the Discontinuation Agreement shall commence on and from Licence Award and shall expire on the later of:
			(a) expiry of the SCA; and
			(b) expiry of the RSA.
3.	Discontinuation	3.1	Subject to sections 3.2–3.5, the SoS will have the right (but not the obligation) in its absolute discretion, to Discontinue the provision of the GSP on or where:
			 (a) claims, individually or in aggregate with all other previous claims, under the SCA have exceeded a minimum of two hundred million pounds (£200,000,000) (indexed) in the aggregate or such higher amount as arises from adjustments in accordance with section 4.4 ("SCA Discontinuation Threshold");
			(b) pursuant to section 5 (Unavailability) of section C (Supplemental Compensation Agreement), the conditions for supplemental compensation protection in respect of Unavailability of commercial insurance for any of the Insured Risks or Insured Losses have been satisfied at any Renewal Date;
			(c) if the aggregate claims for Difference Payments under the RSA are forecast to exceed sixty (60)% of Allowed Revenue in each of the next three Charging Years, as forecast in accordance with section 5.1 and subject to section 5.3 ("RSA Discontinuation Threshold"); or
			(d) the SoS and T&SCo agree in writing that the GSP should be Discontinued,

No.	Heading	Descri	ption			
			(each	a "Dis	continuation Event").	
		3.2	Either Party may, at any time prior to service of Discontinuation Notice by the SoS pursuant to section 3.5 submit a proposal to the Liaison Committee that the GS should be Discontinued, and upon such referral the Liaiso Committee shall consider and discuss such proposal(s) a soon as reasonably practicable and in any event withi [ten (10)] Business Days.			
		3.3	Either Party may, at any time prior to service of a Discontinuation Notice by the SoS pursuant to section 3.5, serve notice in writing to the other Party, copied to the Regulator, if it considers that a Discontinuation Event has occurred (a "Potential Discontinuation Event Notice").			
		3.4	The Parties shall meet as soon as reasonably practicable to discuss the Potential Discontinuation Event Notice and seek to agree whether a Discontinuation Event has occurred.			
		3.5	The SoS shall within [nine (9) months] of agreement by the Parties pursuant to section 3.4, or a determination under the terms of the Discontinuation Agreement, that a Discontinuation Event has occurred:			
			(a)		e the Discontinuation Event is triggered ant to either of section 3.1(b) to section),	
				(i)	serve a Discontinuation Notice on T&SCo, copied to the Regulator; or	
				(ii)	confirm that it has elected not to serve a Discontinuation Notice on T&SCo, and	
			(b)		e the SCA Discontinuation Threshold is red pursuant to section 3.1(a):	
				(i)	serve a Discontinuation Notice on T&SCo, copied to the Regulator;	
				(ii)	confirm that it has elected not to serve a Discontinuation Notice on T&SCo or	
				(iii)	provide an update to T&SCo, in writing, on the progress of its decision on whether or not the SoS will elect to Discontinue (including in such update, a new time period for making its decision and issuing a Discontinuation Notice).	

No.	Heading	Descri	Description		
		3.6	A "Discontinuation Notice" shall specify:		
			(a) the Discontinuation Event that has given rise to the right of the SoS to Discontinue;		
			(b) the Discontinuation Date.		
		3.7	In the event that the SoS elects not to Discontinue pursuant to section 3.5, the SoS waives its right to Discontinue on that basis, provided that such waiver shall be without prejudice to the right of the SoS to Discontinue on the agreement or determination of any future Discontinuation Event.		
		3.8	Where the SoS has served a Discontinuation Notice pursuant to section 3.5 and the SCA is continuing, the SoS shall be entitled to serve a second Potential Discontinuation Event Notice and/or Discontinuation Notice, in accordance with the process in this section 3 (<i>Discontinuation</i>) in the event that a further Discontinuation Event occurs.		
4.	SCA Discontinuation Threshold	4.1	Following service of a Potential Discontinuation Event Notice, T&SCo shall provide to SoS its estimate of and evidence in support of:		
			 (a) aggregate amount of claims already made under the SCA for the Insured Losses arising from any Insured Risk that has already occurred; and 		
			 (b) a forecast of any claims not yet made under the SCA for Insured Losses arising from an Insured Risk that has already occurred, 		
			(c) [<i>other</i>],		
			assured by an independent broker, within [twenty (20)] Business Days.		
		4.2	The SoS shall assess the information and estimate provided by T&SCo pursuant to section 4.1 as soon as reasonably practicable and seek to agree an amount for the purposes of the SCA Discontinuation Threshold.		
		4.3	The SoS may, at its discretion, seek independent advice and request further information or updated forecasts in respect of the information provided pursuant to section 4.1 and T&SCo shall promptly comply with any such further information requests made by the SoS.		
		4.4	If any claims under the SCA, individually or in aggregate with all other previous claims, exceed the then current SCA Discontinuation Threshold and the SoS has not exercised its right to Discontinue pursuant to section 3.5		

No.	Heading	Descri	escription		
			(<i>Discontinuation</i>), then the current SCA Discontinuation Threshold shall be increased by a figure of one hundred million pounds (£100,000,000) (indexed) or above in the first instance, and thereafter by such other figure (indexed) as the SoS may specify in its sole direction.		
5.	RSA Discontinuation Threshold	5.1		njunction with the process for setting the Allowed nue and associated User charges for each Charging	
			(a)	T&SCo will provide the SoS with a forecast of the Allowed Revenue for each of the next three Charging Years and a forecast of the revenues expected to be collected from User charges pursuant to the CCS Network Code, based on Users which have acceded to the CCS Network Code or which have reached [a specified stage of application for accession];	
			(b)	T&SCo shall also provide evidence to the SoS that the Regulator has reviewed and assured or confirmed the forecasts referred to in section 5.1(a) (as applicable), to the extent the Regulator is required to do so under the Licence or the CCS Network Code; and	
			(c)	the SoS will determine whether or not the RSA Discontinuation Threshold has been exceeded using the assured or confirmed forecasts.	
		5.2	and r respe 5.1 a	SoS may, at its discretion, seek independent advice request further information or updated forecasts in ect of the information provided pursuant to section and T&SCo shall promptly comply with any such er information requests made by the SoS.	
		5.3	follow	he purposes of section 3.1(c) (<i>Discontinuation</i>), the ving provisions shall apply in relation to the operation e-operations Difference Payments:	
			(a)	for any Charging Year or Charging Years in respect of which (or part of which) it is forecast that Pre-operations Difference Payments will be payable pursuant to section 8.1(a) of the RSA, such Charging Year(s) will not be treated as a Charging Year or Charging Years for the purposes of section 3.1(c) (<i>Discontinuation</i>); and	
			(b)	subject to section 5.3(a), for any Charging Year(s) in respect of which (or part of which) it is forecast that Pre-operations Difference Payments will be payable pursuant to section [8.1(b)] of the RSA, the Allowed Revenue in such Charging Year(s) will be forecast as zero and, accordingly,	

No.	Heading	Descrip	tion	
				automatically result in the aggregate claims for Difference Payments under the RSA exceeding sixty (60)% of Allowed Revenue in such Charging Year(s).
6.	Consequences of Discontinuation	6.1	Subject to section 6.2 and section 6.3, an election by SoS to Discontinue will result in:	
			(a)	a calculation of compensation payable in respect of debt/equity investors in T&SCo - see sections 10 and 11;
			(b)	the Parties being required to agree and implement the Approved Discontinuation Plan – see section 18;
			(c)	the Regulator being entitled to revoke the Licence; and
			(d)	the SoS being entitled to institute a Transfer Scheme.
		6.2	Where there is a Partial Discontinuation pursuant section 8 (<i>Partial Discontinuation</i>), section 6.1(b) a section 6.1(d) shall not apply.	
			to sec apply payme	e a second Discontinuation Notice is served pursuant ction 3.8 (<i>Discontinuation</i>), section 6.1(a) shall not and T&SCo shall not be entitled to any further ent of compensation as a result of the decision by o Discontinue.
7.	Discontinuation Decision	7.1		ntinuation under these provisions is a right of the nd not an obligation.
8.	Partial Discontinuation	8.1	Discor servin Partie amene of th Disco the Pa	ut prejudice to any rights that the SoS has to ntinue the GSP in its entirety, following either Party ag a Potential Discontinuation Event Notice, the s may agree, as an alternative to Discontinuation, dments to the application of the GSP to take account be potential Discontinuation Event (a "Partial Intinuation"), and as part of any such agreement, arties shall agree the impact of any such Partial ntinuation, including but not limited to:
			(a)	any discontinuation compensation payable in respect of such Partial Discontinuation;
			(b)	any Approved Discontinuation Plan in respect of such Partial Discontinuation; and
			(c)	any other consequences of such Partial Discontinuation (including in respect of any relevant assets or documents).

No.	Heading	Descri	escription			
9.	Total Compensation Amount	9.1	Comp	ntinuation compensation will be capped at the Total ensation Amount and paid following an election by oS to Discontinue in the following order of priority:		
			(a)	first, to senior debt creditors in respect of the Senior Debt Liabilities – see section 10; and		
			(b)	second, to equity investors in T&SCo – see section 11.		
		9.2	If T&SCo is not insolvent at the time SoS serves the Discontinuation Notice, the "Total Compensation Amount" will be a contractually certain amount which (subject to final modelling) is anticipated to be equal to the sum of the RAV and the SRAV assigned by the Regulator at the Discontinuation Date, less any amount payable pursuant to paragraph 1.3(a) of the Insurance Schedule, save to the extent such amount is recovered by SoS pursuant to the loss payee provision referenced in section [3.5] of the SCA, plus:			
			 (a) any accrued amounts as at the Discontinuation Date which would have been assigned by the Regulator to the RAV or the SRAV under the Licence, but for Discontinuation; 			
			(b)	any Breakage Costs; and		
			(c)	the Decommissioning Shortfall Amount in the event that Discontinuation is a result of (i) the RSA Discontinuation Threshold being reached or (ii) conditions for Unavailability of commercial insurance for any of the Insured Risks or Insured Losses being met.		
		9.3	amou discre [Fund the lia it is discha paym any s	e the SoS is liable to pay T&SCo an additional nt pursuant to section 9.2(c) the SoS may, in its sole ation, elect to make such payment directly to the Co / decommissioning fund account] to discharge ability for the Decommissioning Shortfall Amount and agreed that any such payment by the SoS in arging such liability shall (to the extent of such ent) absolve the SoS from any obligation to make uch payments to T&SCo or its [receiver, assignees accessors].		
		9.4	Event	iscontinuation occurs during a period while a Failure has occurred and has not been Remedied or ed to a Remedy Event, then any project spend:		
			(a)	which is incurred during the period that a Failure Event has occurred but has not been remedied or reduced to a Remedy Event; and		
				,,		

No.	Heading	Description				
		(t	 which qualifies for assignment by the Regulator to the RAV under the Licence, 			
		If E ^v	hall be deducted from the Total Compensation Amount. The Failure Event is Remedied or reduced to a Remedy vent prior to a Discontinuation occurring, this deduction hall cease to apply.			
		C w SI U a P T T t t s f b	Subject to section 9.6, if any amount of the Total Compensation Amount payable by the SoS in accordance with this section 9 (<i>Total Compensation Amount</i>) is subject to Tax payable to a Relevant Authority in the United Kingdom, then the SoS shall pay to T&SCo such additional amount as will put T&SCo in the same after Tax position as it would have been had the payment of the Total Compensation Amount not been subject to Tax, taking account of any relief, allowances, deduction, setting off or credit in respect of Tax (whether available by choice or not) which may be available to T&SCo to reduce the Tax to which the payment is subject.			
		ar di ar th A S sr m	/here the SoS is liable to pay T&SCo an additional mount pursuant to section 9.5, the SoS may, in its sole iscretion, elect to make such payment directly to the pplicable Relevant Authority to discharge the liability for ne Tax payable in relation to the Total Compensation mount and it is agreed that any such payment by the oS in discharging such liability shall (to the extent of uch payment) absolve the SoS from any obligation to nake any such payments to T&SCo or its receiver, ssignees or successors.			
10.	Senior Debt Compensation	D D "!	³ T&SCo is not insolvent at the time SoS serves the iscontinuation Notice, following the SoS serving the iscontinuation Notice on T&SCo, the SoS will deposit the Senior Debt Compensation " into a specified account, eing the lower of the Total Compensation Amount and enior Debt Liabilities.			
		10.2 "s	Senior Debt Liabilities" shall mean an amount equal			
		(a	a) outstanding principal, interests, costs, expenses and liabilities in respect of senior debt financing entered into in accordance with the Agreed Financing Principles as at the Discontinuation Date, but excluding:			
			 hedging breakage costs payable by any Relevant Obligor under any Hedging Transactions; and 			
			(ii) early redemption/prepayment fees or premia payable by a Relevant Obligor under			

No.	Heading	Descri	ption	
				any senior debt that contains an embedded hedge of an inflation-linked or interest-rate exposure,
				save to the extent that they constitute Breakage Costs; less
			(b)	any amounts receivable by any Relevant Obligor under any DA Approved Hedging on or after the Discontinuation Date as a result of early termination of such DA Approved Hedging and any other amounts received by senior creditors as the result of any other enforcement action as at the Discontinuation Date.
		10.3	Comp debt	toS may (at its option) elect to pay the Senior Debt ensation in instalments sized by reference to the repayment profile or as may be otherwise agreed een the SoS and the security trustee.
		10.4	from t as a Comp	oS shall continue to pay debt service costs of T&SCo the date of any revocation of T&SCo's Licence arising result of Discontinuation until the Senior Debt ensation is paid in accordance with section 15 ng and mechanics of payment of Compensation).
11.	Equity Compensation	11.1	Disco Disco "Equit sum e	SCo is not insolvent at the time SoS serves the ntinuation Notice, following the SoS serving the ntinuation Notice on T&SCo, the SoS will deposit the cy Compensation" into a specified account, being a equal to the Total Compensation Amount less Senior Compensation (if positive).
		11.2	conne comp a ma	e and to the extent any breakage costs arise in ection with a hedging strategy that is not in liance with the Approved Hedging Policy these will be tter for equity to meet from within the Equity ensation.
12.	Compensation on insolvency	12.1	Disco Amou Comp value shall	ensation will be determined by an independent r in accordance with section 12.2, and such amount not result in greater compensation than would have the case in the event of a Discontinuation outside of
		12.2	indep be in and s	basis of the valuation of T&SCo carried out by an endent valuer pursuant to section 12.1 above, shall accordance with section 14 of the Liaison Agreement hall assume a transfer of the T&SCo's undertaking going concern and in turn apply the valuation

No.	Heading	Descri	escription		
			assumptions in section 14.2(b)(i) of the Liaison Agreement.		
13.	Set off against Equity Compensation	13.1	The SoS will have the right to set off any outstanding claims, subject to sections 9.5 and 9.6 (<i>Total Compensation Amount</i>), in relation to:		
			(a) tax liabilities of T&SCo to HMRC;		
			(b) liabilities of T&SCo to SoS under or pursuant to the GSP,		
			against T&SCo against the payment of the Equity Compensation.		
		13.2	Following agreement or final determination of any liabilities which are subject to the right of set-off in section 13.1, if the amount set-off in respect of any outstanding claim exceeds the amount agreed or determined in respect of that claim, then the SoS shall be obliged to return any such excess to T&SCo within [twenty (20)] Business Days following agreement or determination of the excess.		
14.	Reduction/forfeiture of Equity Compensation	14.1	Equity Compensation may be reduced or forfeited (as proportionate to the relevant circumstances) where there is Wilful Misconduct or Gross Negligence and such Wilful Misconduct or Gross Negligence caused or contributed to the Discontinuation.		
		14.2	"Wilful Misconduct or Gross Negligence" means any act or failure to act by T&SCo or its personnel that was:		
			(a) an intentional breach of the Licence, Support Documents or Storage Permit which results in an increased risk of the occurrence of any Insured Risk or Insured Loss; or		
			(b) intended to cause, or was in reckless disregard of, or wanton indifference to an increased risk of the occurrence of any Insured Risk or Insured Loss, but shall not include an ordinary error of judgement or mistake made by personnel of T&SCo in the exercise in good faith of any function, authority or discretion conferred on such personnel that would constitute mere ordinary negligence.		
15.	Timing and mechanics of payment of Compensation	15.1	Senior Debt Compensation and Equity Compensation will be paid within [twenty (20)] Business Days of agreement or determination of the completion of applicable calculation procedures.		

No.	Heading	Description				
		15.2	Disco	ntinuat	scontinuation, calculation of compensation on ion shall be in accordance with the procedure is section 15.2:	
			(a)	Disco Regul	o shall within [thirty (30)] Business Days of ntinuation provide a written report to the ator in accordance with condition B20 Ilatory Reporting) of the Licence, setting out:	
				(i)	the value of any SRAV as at the Discontinuation Date;	
				(ii)	the value of any RAV as at the Discontinuation Date; and	
				(iii)	evidence of the methodology and calculations, including the underlying data, used by T&SCo to determine the values under sections 15.2(a)(i) and 15.2(a)(ii);	
			(b)	report shall a by T& T&SCo wheth SRAV	ving receipt by the Regulator of the written t pursuant to section 15.2(a), the Regulator assure or confirm such values and calculations &SCo and confirm to each of the SoS and o within [sixty (60) Business Days] of receipt her it agrees with such values assigned to any or RAV by T&SCo or otherwise propose native values for consideration;	
			(c)	Days	o shall within [one hundred (100)] Business of Discontinuation provide a written report to oS, setting out:	
				(i)	the Total Compensation Amount pursuant to section 9(<i>Total Compensation Amount</i>);	
				(ii)	the Senior Debt Liabilities and the Senior Debt Compensation as certified on behalf of the secured creditors pursuant to section 10(Senior Debt Compensation);	
				(iii)	the Equity Compensation payable pursuant to section 11 (<i>Equity compensation</i>); and	
				(iv)	[others];	
			(d)	of the T&SC notice Total	oS shall, within [twenty (20)] Business Days e receipt of the written report provided by o pursuant to section 15.2(c), confirm on e to T&SCo whether or not it agrees with the Compensation Amount, the Senior Debt ities, the Senior Debt Compensation and the	

No.	Heading	Description
		Equity Compensation (and if not set out the reasons for why it disagrees with such amounts);
		(e) within [ten (10)] Business Days of receipt of the notice under section (d), T&SCo shall confirm whether or not it agrees with the amounts notified by the SoS under section (d) and, if T&SCo does not agree, either Party may refer the Relevant Dispute for determination in accordance with section 21 (<i>Dispute Resolution Procedure</i>); and
		(f) where the SoS exercises its right to Discontinue while T&SCo is insolvent or in Special Administration, section 12 (<i>Compensation on</i> <i>insolvency</i>) shall apply to determine the Total Compensation Amount, Senior Debt Compensation and Equity Compensation.
16.	Fee for Compensation	16.1 There will be no charge to T&SCo or the equity recipients and providers of senior debt for the provision of compensation on discontinuation.
17.	Reimbursement of Discontinuation Payment	17.1 The SoS will have a right of reimbursement from T&SCo for amounts paid out under the Discontinuation Agreement on the basis that such right:
		 (a) will rank behind senior debt but ahead of subordinated debt/equity (other than amounts recovered as part of the discontinuation payment itself); and
		 (b) will be backed by security over those of T&SCo's assets which have not transferred to a replacement T&SCo (if there is one).
		17.2 If following payment of both the Senior Debt Compensation and the Equity Compensation there is still a secured senior debt shortfall, the SoS's security over any amounts recoverable by T&SCo is subordinated to the secured senior funder recovery of that shortfall.
		17.3 Without prejudice to the ranking in paragraph 17.1(a), all amounts representing the payment of Senior Debt Compensation and Equity Compensation and any other amount paid out under the Discontinuation Agreement by the SoS shall to the extent of any such payment, be fully and automatically subrogated to the fullest extent of applicable law to all of the rights of shareholders or creditors to payment of any amounts payable by a Relevant Obligor in respect of a Discontinuation Notice .
18.	Discontinuation Plan	 18.1 In the event of service of a Potential Discontinuation Event Notice, T&SCo shall promptly and in any event within [ten (10)] Business Days submit to the Liaison Committee [and

No.	Heading	Description		
			the Inter-Regulator Forum] a draft of the Approved Discontinuation Plan, in the form of the Template Discontinuation Plan, which shall take into account, amongst other things, the impact of the relevant Discontinuation Event and make-safe activities arising therefrom (the "Draft Discontinuation Plan").	
		18.2	Following submission by T&SCo of the Draft Discontinuation Plan pursuant to section 18.1, the Liaison Committee shall meet to consider the plan and agree it by the date of agreement by the Parties or a determination that a Discontinuation Event has occurred (the " Approved Discontinuation Plan ").	
		18.3	The Parties shall seek to agree the Approved Discontinuation Plan by the Discontinuation Date and any failure to agree the Discontinuation Plan shall be subject to section 21 (<i>Dispute Resolution Procedure</i>).	
		18.4	Immediately following the Discontinuation Date, T&SCo shall be required to carry out any immediate make-safe activities to the T&S Network implemented by T&SCo pursuant to the Approved Discontinuation Plan.	
		18.5	The Approved Discontinuation Plan shall apply from the Discontinuation Date until the date on which all activities under the Approved Discontinuation Plan have been completed.	
		18.6	Either Party shall be entitled to propose an amendment to the Approved Discontinuation Plan, on notice to the Liaison Committee.	
		18.7	The Liaison Committee shall meet within [five (5)] Business Days to discuss the proposed amendment to the Approved Discontinuation Plan and either agree, propose other amendments or reject the proposed amendment.	
		18.8	No amendment or variation to the Approved Discontinuation Plan shall be effective unless it is in writing and approved unanimously by the Liaison Committee.	
		18.9	T&SCo shall report to the Liaison Committee and the Inter-Regulator Forum on progress as against the Approved Discontinuation Plan on a monthly basis (or as otherwise agreed in writing by the Parties to the Discontinuation Agreement).	
19.	Approved Discontinuation Plan Obligations and Equity	19.1	Following Discontinuation and until the completion of the Approved Discontinuation Plan, the SoS shall be entitled to withhold from the Equity Compensation to be paid in accordance with section 11 (<i>Equity Compensation</i>), an amount not exceeding fifteen million pounds	

No.	Heading	Description		
	Compensation Retention		(£15,000,000) (indexed), such portion of Equity Compensation retained to be released in accordance with the staged release detailed in the Approved Discontinuation Plan.	
20.	Representatives and personnel	20.1	Roles of the representatives of the Parties to the Discontinuation Agreement are to be developed, including delegated authority.	
		20.2	T&SCo representative shall:	
			 (a) manage and oversee compliance by T&SCo with its duties and obligations under the Discontinuation Agreement; and 	
			(b) exercise any right and perform any obligation exercisable by or to be performed by the T&SCo representative under the Discontinuation Agreement.	
		20.3	SoS representative shall:	
			 (a) manage and oversee compliance by SoS with its duties and obligations under the Discontinuation Agreement; and 	
			(b) exercise any right and perform any obligation exercisable by or to be performed by the SoS representative under the Discontinuation Agreement.	
21.	Dispute Resolution Procedure	21.1	Any dispute arising out of or in connection with the Discontinuation Agreement, including in relation to any amounts payable under or pursuant to the Discontinuation Agreement (a "Relevant Dispute"), shall be resolved in accordance with the Dispute Resolution Procedure set out in this section 21.	
		21.2	In the event of a Relevant Dispute, the matter shall first be referred to the Liaison Committee for resolution by the Parties.	
		21.3	If the Relevant Dispute is not resolved by the Liaison Committee after a period of [ten (10)] Business Days on and from such referral (or such other period as agreed between the Parties in writing), either Party shall be entitled to refer the Relevant Dispute to identified senior representatives for resolution.	
		21.4	If the Relevant Dispute is not resolved by the senior representatives after a further period of [ten (10)] Business Days on and from such referral to identified	

No.	Heading	Description		
		senior representatives (or such other period as between the Parties in writing), then:	agreed	
		(a) the Parties may agree to refer the Relevant for expert determination; or	: Dispute	
		(b) either Party may refer the Relevant Disput courts of England and Wales.	te to the	
22.	Indexation	2.1 Any amounts marked as "indexed" und Discontinuation Agreement shall be subject to in by CPIH, unless otherwise stated.		
23.	Amendments	3.1 No amendment shall be made to the Discont Agreement without the agreement of both Pa writing.		
		3.2 Either Party may, at any time prior to servi Discontinuation Notice by the SoS pursuant to sec refer to the Liaison Committee a proposal for ame to the SCA Discontinuation Threshold or t Discontinuation Threshold and upon such refer Liaison Committee shall consider and discu proposal(s) as soon as reasonably practicable an event within [ten (10)] Business Days.	tion 3.5, ndments he RSA rral, the ss such	
24.	Assignment and Sub-Contracting	4.1 The Discontinuation Agreement shall benefit and parties, their permitted assignees and re successors and any reference in the Discont Agreement to any party shall be construed accord	spective inuation	
		4.2 The Discontinuation Agreement will include restrict the assignment and/or transfer by T&SCo of an rights or obligations under the Discontinuation Ag (in whole or in part except) without the prior consent of the SoS (such consent not to be unreal withheld or delayed), save that T&SCo will be p to create a security assignment in favour of its creditors (and SoS will assist in facilitating this, that all costs and expenses properly incurred by giving effect to such assignment are paid by T&S	ny of its reement written asonably ermitted secured provided y SoS in	
		4.3 The SoS may, subject to the conditions to be set the Discontinuation Agreement, transfer its r novate its obligations under the Discont Agreement to any Minister of the Crown or ar directly wholly-owned by a Minister of the Crown	ights or tinuation ly entity	
25.	Confidentiality	5.1 The Discontinuation Agreement will set out confic provisions which will apply to all Parties in respe Project. There will be customary exceptions.	-	
		5.2 The Discontinuation Agreement sets out that the and that T&SCo may become subject to, the requi		

No.	Heading	Description		
		of the FOIA and the Environmental Information Regulations and each Party shall facilitate compliance by the other Party with its information disclosure requirements pursuant to the FOIA and the Environmental Information Regulations.		
26.	Survivorship	 26.1 Following the expiry of the Discontinuation Agreement each Party shall cease to have rights or obligations under the Discontinuation Agreement, save that the rights under section 9 (<i>Total Compensation Amount</i>), section 10 (<i>Senior Debt Compensation</i>), section 11 (<i>Equity Compensation</i>), section 12 (<i>Compensation on insolvency</i>), section 13 (<i>Set off against Equity Compensation</i>), section 14 (<i>Reduction/forfeiture of Equity Compensation</i>), section 15 (<i>Timing and mechanics of payment of Compensation</i>), section 17 (<i>Reimbursement of Discontinuation Payment</i>), section 18 (<i>Discontinuation Plan</i>), section 19 (<i>Approved Discontinuation Plan Obligations and Equity Compensation Retention</i>), section 21 (<i>Dispute Resolution Procedure</i>), section 22 (<i>Indexation</i>), section 25 (<i>Confidentiality</i>) and section 28 (<i>Governing Law</i>) will continue in full force and effect. 		
27.	Rights of Third Parties	27.1 A person who is not a Party to the Discontinuation Agreement shall have no rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any of its terms.		
28.	Governing Law	28.1 The Discontinuation Agreement shall be governed by and construed in accordance with English law and the courts of England and Wales shall have exclusive jurisdiction.		
29.	Appendix	29.1 Appendix 1 will form a schedule to the Discontinuation Agreement.		
30.	Definitions	In this section D (<i>Discontinuation Agreement</i>), terms defined in the Licence heads of terms shall have the same meaning and:		
		"Affiliate" means in relation to T&SCo, any Parent Undertaking of T&SCo, any Subsidiary Undertaking of T&SCo, or any Subsidiary Undertaking of a Parent Undertaking of T&SCo		
		"Agreed Financing Principles" means the approved financing principles set out in Appendix 1 of this section D (<i>Discontinuation Agreement</i>);		
		"Allowed Revenue" has the meaning given to it in the RSA;		
		"Approved Discontinuation Plan" means the plan agreed or determined pursuant to section 18 (as amended or supplemented from time to time), covering immediate make-safe activities following Discontinuation, to the extent not covered by the		

No.	Heading	Description
		decommissioning plan, and based on the Template Discontinuation Plan set out at Appendix 2 to this section D;
		"Approved Hedging Policy" means the approved hedging policy set out in Appendix 1 of this section D (<i>Discontinuation Agreement</i>);
		"Breakage Costs" means the costs of early termination of any DA Approved Hedging payable by a Relevant Obligor, including early redemption/prepayment fees or premia payable under Embedded Hedging which constitutes DA Approved Hedging;
		"CCS Network Code" means the document designated as such and required to be maintained by the licensee under the Licence;
		"Charging Year" has the meaning given to it in the RSA;
		["Decommissioning Shortfall Amount" means [an amount calculated by reference to the planned accrual of the decommissioning fund pursuant to the Licence, as may be reassessed at the Discontinuation Date].
		Note: definition to be updated to reflect a process for updating of the decommissioning fund as at the Discontinuation Date, to align with the process for updating the decommissioning fund at each Periodic Review.
		"Difference Payments" has the meaning given to it in the RSA heads of terms;
		" Discontinuation " means the exercise by the SoS of its right to issue a Discontinuation Notice in accordance this section D (<i>Discontinuation Agreement</i>) and " Discontinue " shall be construed accordingly;
		"Dispute Resolution Procedure" means the procedure for the resolution of disputes set out in section 21 (<i>Dispute Resolution Procedure</i>) of this section D (<i>Discontinuation Agreement</i>);
		"Discontinuation Date" the date specified as such in, and falling on or after the date of, the relevant Discontinuation Notice;
		"Discontinuation Event" has the meaning given to it in section 3.1;
		"Discontinuation Notice" has the meaning given to it in section 3.6;
		"Draft Discontinuation Plan" has the meaning given to it in section 18;
		"Embedded Hedging" has the meaning given to it in paragraph 12 of Part 1, Appendix 1 (<i>Approved Hedging and Agreed Financing</i>) in this section D (<i>Discontinuation Agreement</i>);

No.	Heading	Description		
		"Environmental Information Regulations" means the Environmental Information Regulations 2004 (as amended);		
		"Equity Compensation" has the meaning given to it in section 11 (<i>Equity Compensation</i>) of this section D (<i>Discontinuation Agreement</i>);		
		"Finance Documents" has the meaning given to it in section E (<i>Liaison Agreement</i>);		
		"FOIA" means the Freedom of Information Act 2000 (as amended);		
		"GSP" has the meaning given to it in the Liaison Agreement heads of terms;		
		"Insured Losses" has the meaning given in section C (<i>Supplemental Compensation Agreement</i>);		
		"Insured Risks" has the meaning given in section C (Supplemental Compensation Agreement);		
		"Inter-Regulator Forum" has the meaning given to it in the section E (<i>Liaison Agreement</i>);		
		"Liaison Committee" has the meaning given to it in the section E (Liaison Agreement);		
		"Licence" means the economic licence to be granted under HM Government's CCUS programme;		
		"Notional Amount" means at any time the notional amount of the senior debt to which that Hedging Transaction relates or for any Embedded Hedging the notional amount of the senior debt to which such Embedded Hedging relates;		
		"Minister of the Crown" shall have the meaning given to it in the Minister of the Crown Act 1975;		
		"Parent Undertaking" means a parent undertaking within the meaning of section 1162 of the Companies Act 2006;		
		"Partial Discontinuation" has the meaning given to it in section 8.1 (<i>Partial Discontinuation</i>) of this section D (<i>Discontinuation Agreement</i>);		
		"Potential Discontinuation Event Notice" has the meaning given to it in section 3.3 (<i>Discontinuation</i>);		
		"Pre-operations Difference Payments" has the meaning given to it in the RSA;		
		"RAV" has the meaning given to in the Licence, as it relates to the Approved T&S Network;		

No.	Heading	Description
		" Relevant Obligor " has the meaning given to it in paragraph 1 of Part 1, Appendix 1 (<i>Approved Hedging and Agreed Financing</i>) in this section D (<i>Discontinuation Agreement</i>);
		"Relevant Regulator" means the NSTA, OPRED, HSE, or Ofgem;
		"Remedied" has the meaning given to it in section E (<i>Liaison Agreement</i>);
		"Renewal Date" has the meaning given in section C (Supplemental Compensation Agreement);
		"Revenue Support Agreement" or "RSA" means the revenue support agreement;
		"Relevant Dispute" has the meaning given to it in section 21.1 (<i>Dispute Resolution Procedure</i>);
		"RSA Discontinuation Threshold" has the meaning given to it in section 3 (<i>Discontinuation</i>) of this section D (<i>Discontinuation</i> <i>Agreement</i>);
		"SCA" means the supplemental compensation agreement;
		"SCA Discontinuation Threshold" has the meaning given to it in section 3 (<i>Discontinuation</i>) of this section D (<i>Discontinuation</i> <i>Agreement</i>);
		"Senior Debt Compensation" has the meaning given to it in section 10 (Senior Debt Compensation) of this section D (Discontinuation Agreement);
		"Senior Debt Liabilities" has the meaning given to it in section 10 (Senior Debt Compensation) of this section D (Discontinuation Agreement);
		"SRAV" has the meaning given to in the Licence, as it relates to the Approved T&S Network;
		"Subsidiary Undertaking" means a subsidiary undertaking within the meaning of section 1162 of the Companies Act 2006;
		"Template Discontinuation Plan" means the template plan as set out in Appendix 2 of this section D (<i>Discontinuation</i> Agreement) that shall be agreed at Licence Award and shall form the basis of any Approved Discontinuation Plan;
		"Total Compensation Amount" has the meaning given to it in section 9 (<i>Total Compensation Amount</i>) of this section D (<i>Discontinuation Agreement</i>);
		"Unavailable" and "Unavailability" have the meaning given to them in section C (Supplemental Compensation Agreement); and

No.	Heading	Description
		"Wilful Misconduct or Gross Negligence" has the meaning given to it in section 14.2 (<i>Reduction/forfeiture of Equity</i> <i>Compensation</i>) of this section D (<i>Discontinuation Agreement</i>).

Appendix 1 to section D (*Discontinuation Agreement*): Approved Hedging Policy and Agreed Financing Principles

Part 1: Approved Hedging Policy

Part A

DA Approved Hedging and Exposure Limits

Note: this Approved Hedging Policy is designed only for the determination of DA Approved Hedging for the purposes of calculation of the Total Compensation Amount under the Discontinuation Agreement. Separate consideration needs to be given to an appropriate finance plan (and any associated hedging) that meets wider value for money requirements.

Note: position on power hedging and parameters to be confirmed following further input from T&SCos and the extent to which power hedging is linked to the finance structure.

General

- Each of T&SCo and FinCo (the "Relevant Obligors") shall only enter into hedging agreements ("Hedging Agreements") and hedging transactions ("Hedging Transactions") that:
 - (a) comply with the hedging policy as set out in the finance documents as at Licence Award (**"Finance Hedging Policy"**); and
 - (b) in relation to Hedging Transactions that will have the benefit of coverage of hedging liabilities as Breakage Costs under and as defined in the Discontinuation Agreement, have been entered into (or amended in accordance with paragraph 12 and 13) in compliance with each of the conditions specified in paragraphs 3 to 7 unless otherwise agreed by the SoS ("DA Approved Hedging").

The Parties acknowledge and agree that only the Relevant Obligors may enter into Hedging Agreements and Hedging Transactions in connection with the Project.

Note: a FinCo would be included in the structure if listed bonds are issued (given that the issuing entity is likely to be a SPV incorporated as a public company), in which case the Hedging Principles should apply to both T&SCo and FinCo.

Approved Hedging Policy Conditions

2. The Relevant Obligors must at all times satisfy and comply with the Approved Hedging Policy Conditions as set out in Part B (*GSP Hedging Compliance Conditions*).

New Hedging Transactions

- 3. Prior to any Relevant Obligor entering into any Hedging Agreement or Hedging Transaction that is intended to constitute DA Approved Hedging (the **"Contemplated Hedging Transaction"**), T&SCo shall calculate:
 - (a) the Notional Aggregate Hedging Exposure using the Notional Exposure Calculation Methodology and taking account of the Contemplated Hedging Transaction; and

- (b) the Actual Aggregate Hedging Exposure as at that Test Date using the Actual Exposure Calculation Methodology.
- 3A T&SCo shall notify the SoS no later than [five] Business Days after a Relevant Obligor enters into a Contemplated Hedging Transaction. Such notification shall include the following information:
 - (a) whether such Contemplated Hedging Transaction is a (i) gilt-lock or forward start pre-Hedging entered into in respect of an anticipated issuance or advance of private placement notes or loans, (ii) interest rate Hedging, (iii) inflation-linked Hedging or (iv) cross-currency Hedging;
 - (b) the aggregate notional exposure covered by such Contemplated Hedging Transaction;
 - (c) the underlying exposure which is being hedged pursuant to such Contemplated Hedging Transaction; and
 - (d) the tenor of such Contemplated Hedging Transaction.
- 3B On each Test Date, T&SCo shall provide a summary of all of the DA Approved Hedging transactions entered into by the Relevant Obligors since the previous Test Date, setting out all of the information set out in paragraphs (a) to (d) above. The T&SCo shall keep a copy of all such reports and make such reports available to the SoS for review during any Discontinuation.

Inception Exposure Limit Conditions

- 4. Subject to paragraph 7, the Relevant Obligors may only enter into the Contemplated Hedging Transaction that will constitute DA Approved Hedging if:
 - (a) the Notional Aggregate Hedging Exposure calculated using the Notional Exposure Calculation Methodology as at that Test Date, taking account of the Contemplated Hedging Transaction, is less than the Notional Aggregate Hedging Exposure Limit; and
 - (b) the Actual Aggregate Hedging Exposure calculated using the Actual Exposure Calculation Methodology as at that Test Date is less than the Actual Aggregate Hedging Exposure Limit.

Note: thresholds for the Notional Aggregate Hedging Exposure Limit and the Actual Aggregate Hedging Exposure Limit in respect of the First Financing Period are subject to further consideration.

Note: further details of the Notional/Actual Exposure Calculation Methodology to be provided in due course.

- 5. No later than [●] months before the start of each Financing Period, T&SCo shall deliver to the SoS for consideration a hedging plan in respect of the Relevant Obligors for such Financing Period, which shall set out:
 - (a) the proposed Notional Aggregate Hedging Exposure Limit and the proposed Actual Aggregate Hedging Exposure Limit to apply during such Financing Period;
 - (b) any proposed amendments to the Severe Sensitivity Assumptions; and/or
 - (c) any update to the proposed limit on total Notional Amount of all Hedging Transactions (as set out in paragraph 17(d) of Part B),

for such Financing Period. The T&SCo shall provide the SoS with such further information and/or justification as the SoS may request. The SoS shall (acting reasonably) notify T&SCo of the Notional Aggregate Hedging Exposure Limit, the Actual Aggregate Hedging Exposure Limit, and any amended Severe Sensitivity Assumptions or limit on total Notional Amount of all Hedging Transactions (as set out in paragraph 17(d) of Part B) which it approves in respect of such Financing Period (and where the SoS does not approve of the proposals made by T&SCo the SoS shall provide reasons for this).

- 6. Without prejudice to the requirements of paragraphs 3, 4 and 5, no Relevant Obligor shall enter into a Contemplated Hedging Transaction, and no Contemplated Hedging Transaction will constitute DA Approved Hedging, unless the Input Parameters (as defined in paragraph 17 of Part B) have been satisfied or waived by the SoS.
- 7. If, in respect of a Contemplated Hedging Transaction, the conditions in paragraphs 4 and 5 or the Input Conditions have not been met, the Relevant Obligors shall only be entitled to enter into a Contemplated Hedging Transaction which will constitute DA Approved Hedging if they have demonstrated to the satisfaction of the SoS that the purpose and direct effect of such Contemplated Hedging Transaction will be to reduce the Actual Aggregate Hedging Exposure and the Notional Aggregate Hedging Exposure below the Consultation and Remediation Threshold (as defined below).

Periodic Calculation

- 8. On each Test Date:
 - (a) T&SCo shall calculate the Actual Aggregate Hedging Exposure using the Actual Exposure Calculation Methodology and the Notional Aggregate Hedging Exposure using the Notional Exposure Calculation Methodology; and
 - (b) provide the information at paragraph 3B above

Actual Exposure - Consultation and Remediation Threshold

- 9. If, as at the relevant Test Date (the **"Consultation and Remediation Threshold"**), the Actual Aggregate Hedging Exposure is greater than the applicable limit for the relevant Financing Period , the Relevant Obligor(s) shall (i) within three months of such Test Date (or such period as may otherwise be agreed with the SoS), consult with the SoS and demonstrate compliance with prudent treasury management policy in accordance with the principles set out in paragraph 1 of Part B and (ii) prepare and implement a remediation plan to be agreed by the SoS in accordance with paragraph 10 below (the **"Agreed Remediation Plan"**), the purpose of which shall be management of hedging in accordance with prudent treasury management principles (taking account of historic and projected movements in interest, inflation and currency markets and rates, including from the date the relevant Hedging Transactions were entered into) with the objectives of balancing:
 - (a) the interests the Relevant Obligor and the Project, including maintaining prudent levels of hedging;
 - (b) maintaining or otherwise managing the Actual Aggregate Hedging Exposure at a level in respect of future Test Dates which complies with the applicable Notional Aggregate Hedging Exposure Limit and Actual Aggregate Hedging Exposure Limit.
- 10. In reviewing a remediation plan proposed by the Relevant Obligor(s) in accordance with paragraph 9 above and agreeing the Agreed Remediation Plan, the SoS will have regard (without limitation) to the following factors and circumstances:

- (a) the scope and extent of the DA Approved Hedging then outstanding (including the number of transactions entered into by the Relevant Obligors, the exposures hedged and the tenors of such transactions);
- (b) the impact on the Relevant Obligor(s) of any early termination or other breakage costs that would be payable upon early termination or repayment of any DA Approved Hedging and the proportionality of the impact on the Relevant Obligors of taking such mitigating steps;
- (c) the requirements of the Finance Hedging Policy to maintain an effective hedging policy to mitigate the Relevant Obligor(s)' exposure to interest rate, currency and inflation markets and rates;
- (d) the projected movements in interest, inflation and currency markets and rates against the Relevant Obligor(s)' exposures against such risks and its ability to mitigate those risks under a portfolio of Hedging Transactions then outstanding and/or to be entered into by the Relevant Obligor(s); and
- (e) such time for the Relevant Obligors as is reasonable in the circumstances to reduce or otherwise manage the hedging exposure taking into account the factors and circumstances specified in (a) to [(d)] above.
- 11. A Remedy Event will not occur if at any Test Date the Actual Aggregate Hedging Exposure is greater than the applicable limit for the relevant Financing Period, unless and until there is a subsequent breach of paragraph 9 by the Relevant Obligors (whether a failure to consult or agree (including by reference to dispute resolution) or to implement an Agreed Remediation Plan).

Notional Exposure - Consultation Threshold

12. If, as at the relevant Test Date, the Notional Aggregate Hedging Exposure is greater than the applicable limit for the relevant Financing Period, the Relevant Obligor(s) and the SoS shall, within two months of such Test Date (or such period as may otherwise be agreed with the SoS), enter into consultation in relation to the circumstances which have given rise to such Notional Aggregate Hedging Exposure and any matters relating to such exposure that are considered relevant at that time.

Early redemption/prepayment fees for fixed rate and [inflation] linked senior debt

- 13. Early redemption/prepayment fees or premia payable under any senior debt that (a) satisfies the Agreed Financing Principles and (b) contains an embedded hedge of an inflation-linked or interest-rate exposures (such as 'Spens' or make-whole payments) ("Embedded Hedging") will only constitute "DA Approved Hedging" and be included as Breakage Costs if:
 - (a) prior to entering into such senior debt containing Embedded Hedging (or in respect of any amendment to the terms of such senior debt in accordance with paragraphs 15 and 16), the conditions specified, mutatis mutandis, in paragraphs 3 to 7 have been satisfied in respect of such Embedded Hedging and provided that, for the purposes of calculating the Notional Aggregate Hedging Exposure and the Actual Aggregate Hedging Exposure, such calculations shall apply the Breakage Costs payable to the relevant lender or investors in respect of such senior debt upon early redemption/prepayment; and
 - (b) any such early redemption/prepayment fee or premia is on customary market terms from time to time for senior debt containing such Embedded Hedging.

- 14. References in paragraphs 3 to 7 to a Hedging Agreement or Hedging Transaction shall, for these purposes, be deemed to refer to the terms and amounts payable in respect of the early redemption/prepayment fee or premia pursuant to the relevant provisions of such senior debt.
- 15. Any calculation of the Notional Aggregate Hedging Exposure or the Actual Aggregate Hedging Exposure pursuant to this policy (including periodic calculations) shall include the exposures representing any such DA Approved Hedging in respect of senior debt containing Embedded Hedging taking account of the Breakage Costs payable upon early redemption/prepayment.

Amendments to DA Approved Hedging

- 16. No amendment may be made to any agreement, document or transaction in respect of DA Approved Hedging if the effect of such amendment would be to change the economic or other terms and conditions that were relevant for the purposes of originally determining compliance with the conditions specified in paragraphs 3 to 7 (a **"Relevant Amendment"**), other than in accordance with paragraph 17.
- 17. Prior to agreeing or implementing a Relevant Amendment, the relevant hedging or early redemption fee must (taking account of the Relevant Amendment) comply with the conditions specified in paragraphs 3 to 7 of this schedule as if the proposed amended terms of the relevant hedging or early redemption fee were the terms of such hedging or early redemption fee with effect from the date that that applicable amendment will be effective for the purposes of determining compliance with paragraphs 3 to 7.
- 18. If any Relevant Amendment of DA Approved Hedging does not comply with paragraph 17, the relevant hedging or early redemption fee to which the Relevant Amendment relates shall cease to constitute DA Approved Hedging.

Definitions

"Actual Aggregate Hedging Exposure" means the actual aggregate Net Exposure in respect of DA Approved Hedging plus any actual Net Exposure associated with any Contemplated Hedging Transaction that is intended to constitute DA Approved Hedging;

"Actual Aggregate Hedging Exposure Limit" means $\pounds[\bullet]$ or such other figure that is accepted by the SoS pursuant to paragraph 5;

"Actual Exposure Calculation Methodology" means the methodology for calculating the actual Net Exposure as at that Test Date, in accordance with (i) (in the case of Hedging Transactions) the terms of the relevant Hedging Agreements and market standard calculation conventions applied in accordance with the relevant ISDA Definitions or (ii) (in the case of Embedded Hedging) the terms of the relevant senior debt documentation and reasonably agreed procedures and methodologies for calculating 'Spens' or make-whole payments set out in writing prior to the Licence Award or start of the relevant price control period under the Licence in each case agreed by the SoS, provided that no Hedging which is not DA Approved Hedging shall be netted against any Hedging which is DA Approved Hedging for the purposes of such calculation methodology.

"First Financing Period" means the period from Licence Award to the date of the occurrence of the first Review Event thereafter;

"Financing Period" shall mean the First Financing Period and each successive period beginning on a Review Event Date and ending on the next succeeding Review Event Date;

"FinCo" means a limited liability company incorporated as an Affiliate of T&SCo for the purposes of the issuance of any listed bonds or notes in connection with the Project and the on-lending of the proceeds thereof to the T&SCo;

"ISDA Definitions" means:

- (a) in the case of interest rate Hedging Transactions, the 2006 ISDA Definitions or 2021 Interest Rate Derivatives Definitions (or any successor or replacement thereof or supplement thereto) (each, the "Interest Rate Definitions");
- (b) in the case of cross-currency Hedging Transactions, the 1998 FX and Currency Option Definitions either alone or together with the Interest Rate Definitions (or, in each case, any successor or replacement thereof or supplement thereto); and
- (c) in the case of inflation-linked Hedging Transactions, the ISDA Inflation Derivatives Definitions either alone or together with the Interest Rate Definitions (or, in each case, any successor or replacement thereof or supplement thereto);

"**Net Exposure**" means the total net marked to market exposure or other termination liabilities (whether or not crystallised) of the Relevant Obligors to all Hedge Counterparties;

"Notional Aggregate Hedging Exposure" means the projected notional aggregate Net Exposure in respect of DA Approved Hedging plus any notional Net Exposure associated with any Contemplated Hedging Transaction that is intended to constitute DA Approved Hedging;

"Notional Aggregate Hedging Exposure Limit" means $\pounds[\bullet]$ or such other figure that is accepted by the SoS pursuant to paragraph 5;

"Notional Exposure Calculation Methodology" means the methodology for calculating the projected notional Net Exposure as at that Test Date using the Severe Sensitivity Assumptions in accordance with (i) (in the case of Hedging Transactions) the terms of the relevant Hedging Agreements and market standard calculation conventions applied in accordance with the relevant ISDA Definitions or (ii) (in the case of Embedded Hedging) the terms of the relevant senior debt documentation and reasonably agreed procedures and methodologies for calculating 'Spens' or make-whole payments set out in writing prior to the Licence Award or start of the relevant price control period under the Licence in each case agreed by the SoS, provided that no Hedging which is not DA Approved Hedging shall be netted against any Hedging which is DA Approved Hedging for the purposes of such calculation methodology;

"Review Event" shall mean either of:

- (a) the expiry of a Periodic Review under the Licence; or
- (b) the occurrence of a Variation (as defined in section 5 of the Liaison Agreement);

"Review Event Date" means, in respect of a Review Event, the date on such event takes effect or is effective;

"Severe Sensitivity Assumptions" means the interest, inflation and foreign exchange rate sensitivities set out in Annex 1 (*Severe Sensitivity Assumptions*);

"SONIA OIS swap curve" means the yield curve for sterling overnight index swap rates published, as at Licence Award, by Bloomberg as yield curve "GBP SONIA 141" (or any successor or replacement thereof) or any other screen page or index for such yield curve as may be agreed by the SoS and the Relevant Obligors in writing;

"Test Date" means:

- (a) the date of entry into a Contemplated Hedging Transaction that is intended to constitute DA Approved Hedging; and
- (b) each date falling [31 March and 30 September] in each year.

Note: frequency of periodic testing remains under review.

Part B

GSP Hedging Compliance Conditions

- 1. Hedging must be implemented and managed in accordance with a prudent treasury management policy, where for these purposes a **"prudent treasury management policy"** shall be deemed to include implementing and managing the Actual Aggregate Hedging Exposure and the Notional Aggregate Hedging Exposure at levels which comply with the applicable limits for the relevant Financing Period, as determined in accordance with paragraph 4 of Part A.
- 2. Without prejudice to paragraph 1, hedging will be generally consistent with creditor requirements to enable the Relevant Obligors to raise senior debt complying with the Agreed Financing Principles.
- 3. The terms of the Hedging Agreements and Hedging Transactions entered into by the Relevant Obligors will be transacted and managed (without selling optionality) with the objective of (i) minimising the potential marked to market hedging liabilities and termination payments and (ii) managing the maturity concentration, refinancing and solvency risks in relation to the Relevant Obligors.
- 4. The Finance Hedging Policy set out in the finance documents and this Approved Hedging Policy is in agreed form at financial close and no Relevant Obligor shall agree to any amendment, consent or waiver unless approved by the SoS.
- 5. The Relevant Obligors shall only:
 - (a) enter into inflation-linked Hedging Transactions (other than gilt-lock or forward start swaps or other pre-Hedging in respect of the anticipated raising of senior debt to which paragraph (c) shall apply) to manage any remaining inflation exposure arising as a result of having fixed or floating rate senior debt against the indexation of its Revenue Calculations, to the extent such exposure is not effectively managed through the issuance of index-linked debt, in accordance with prudent treasury management policy;
 - (b) enter into cross-currency Hedging Transactions (other than forward start swaps or other pre-Hedging in respect of the anticipated raising of senior debt to which paragraph (c) shall apply) to manage foreign exchange exposures on non-sterling denominated debt if (i) the Relevant Obligor has been unable to issue, (ii) prudent treasury management policy would be inconsistent with issuing, or (iii) the all in rate of such non-sterling denominated debt plus the applicable cross-currency Hedging Transaction would be less than the rate that can be secured in the domestic or international markets for, sterling denominated debt; and
 - (c) enter into any forward start swaps or other pre-Hedging in respect of anticipated raising of senior debt ("Pre-Hedging") if (A) such Pre-Hedging has a mandatory early termination date of no more than [36] months after the date entered into and is or will be cash settled, reset to zero or otherwise terminated in full on the earlier of (i) a date on or prior to raising the relevant senior debt and (ii) the expiry of such [36] month period and (B) the amount of any Breakage Costs (under and as defined in the Discontinuation Agreement) in respect of the Pre-Hedging if calculated immediately following such settlement, reset or termination would be zero,

provided that, in each case, the Relevant Obligor shall only enter into any such Hedging Transactions if they are on commercially reasonable terms that a reasonably prudent operator of a UK-based regulated asset would enter into taking into account the Relevant Obligor's liabilities.

- 6. Inflation-linked hedging will only be transacted in the form of either (i) accreting swaps which (subject to pay-as-you-go features) have the commercial effect of substantially replicating an inflation-linked bond in the context of managing inflation exposures or (ii) revenue style swaps which hedge cash-flow (i.e. without accretions).
- 7. Offsetting Hedging Transactions (including, to the extent not otherwise reflected in such calculation but without double-counting, by adding amounts representing annuities or similar embedded debt) shall be included within the [Senior RAR] and [Net Debt/Adjusted RAV financial ratio calculation] (in each case, as defined in the finance documents) [and shall not be included in the calculation of Actual Aggregate Hedging Exposure or Notional Aggregate Hedging Exposure].
- Accretions by indexation on inflation-linked hedging shall be included within the [Senior RAR] and [Net Debt/Adjusted RAV financial ratio calculation] (in each case, as defined in the finance documents) [and shall not be included in the calculation of Actual Aggregate Hedging Exposure or Notional Aggregate Hedging Exposure].
- 9. If the SoS elects to pay the Senior Debt Compensation in instalments pursuant to section 10.3, the Breakage Costs (under and as defined in the Discontinuation Agreement) are only payable by the SoS, in accordance with the terms of the Discontinuation Agreement, if the SoS elects to cease paying the Senior Debt Compensation in instalments pursuant to section 10.3.
- 10. In the event of Discontinuation, the SoS has the right to:
 - (a) prohibit the Relevant Obligor and the Hedge Counterparty from exercising termination rights in relation to each Hedging Transaction (other than termination in accordance with the finance documents in relation to the occurrence of a Permitted Hedge Termination Event) if the SoS elects to pay the Senior Debt Compensation in instalments pursuant to section 10.3 (and pending the decision by the SoS as to whether or not to so elect); or
 - (b) require termination of the Hedging Transactions if the SoS does not elect to pay the Senior Debt Compensation in instalments pursuant to section 10.3 or elects to cease paying the Senior Debt Compensation in instalments pursuant to section 10.3.
- 11. The SoS must have the right after Discontinuation to appoint a swap manager on behalf of the SoS to manage the winding up of hedging.
- 12. [The Actual Aggregate Hedging Exposure and the Notional Aggregate Hedging Exposure calculated pursuant to paragraph 8 of Part A must be provided to the SoS within 10 Business Days of the relevant Test Date.]
- 13. Relevant Obligors must not terminate any Hedging Transaction in anticipation of Discontinuation without the approval of the SoS. Any amount paid as a Distribution in connection with any such termination prohibited by this paragraph 13 will be deemed to have been paid by the SoS for the purposes of any Equity Compensation payable under the Discontinuation Agreement and to that extent to discharge the SoS's obligations to pay that Equity Compensation.
- 14. Relevant Obligors must ensure that no Trigger Event occurs relating to accretions by indexation to the original notional amounts of super-senior Hedging Agreements which are linked to [inflation] exceeding the greater of [●] per cent. ([●%]) of RAV or £[●] (Indexed) as set out in the finance documents.

Note: thresholds subject to further consideration.

- 15. T&SCo shall procure compliance by FinCo with the terms of this Approved Hedging Policy and ensure that the terms of any issuer/borrower loan agreement(s) or back-to-back hedging agreements entered into by it are consistent with the terms and principles of this Approved Hedging Policy. The terms of this Approved Hedging Policy and all related provisions of the Discontinuation Agreement shall be deemed to operate without double-counting in respect of the hedging arrangements of the Relevant Obligors.
- 16. The Relevant Obligor must procure that Hedge Counterparties provide quarterly valuation statement of their marked to market exposure that are prepared on the same basis as is used for accounting purposes and shall provide such quarterly valuation statements to the SoS upon request.

Input Parameters

- 17. Without prejudice to the requirements of paragraphs 3, 4 and 5 of Part A, no Relevant Obligor shall enter into a Contemplated Hedging Transaction, and no Contemplated Hedging Transaction will constitute DA Approved Hedging:
 - (a) unless such Contemplated Hedging Transaction is (i) gilt-lock and forward start pre-Hedging in respect of the anticipated issuance or advance of private placement notes or loans, (ii) interest rate Hedging, (iii) inflation-linked Hedging or (iv) crosscurrency Hedging;
 - (b) if the tenor of that Hedging Transaction would exceed (and disregarding for these purposes any optional or mandatory breaks):
 - in the case of pre-Hedging of anticipated issuance or advance of private placement notes or loans, [20] years including the pre-Hedge period but provided that such pre-hedging complies with paragraph 17(c) of this Part B;
 - (ii) in the case of interest rate Hedging, [12] years (and subject to paragraph (iv) in this Part B in relation to pre-Hedging of anticipated issuance or advance of private placement notes or loans, such period to include any pre-hedging period);
 - (iii) in the case of inflation-linked Hedging, [15] years (and subject to paragraph 17(b)(iv) in this Part B in relation to pre-Hedging of anticipated issuance or advance of private placement notes or loans, such period to include any pre-hedging period); or
 - (iv) in the case of cross-currency Hedging, [15] years (and subject to paragraph 17(b)(iv) in this Part B in relation to pre-Hedging of anticipated issuance or advance of private placement notes or loans, such period to include any pre-hedging period);
 - (c) in relation to cross-currency Hedging, unless that Contemplated Hedging Transaction is of Sterling and either Euro or US dollar denominated notional amounts; or
 - (d) if the Notional Amount of that Contemplated Hedging Transaction when aggregated with the Notional Amount of all other Hedging Transactions would exceed $\pounds[\bullet]$.

Note: thresholds subject to further consideration with the expectation that the aggregate Notional Amount threshold would equate to an amount determined by reference to (i) the maximum amount of senior debt expected to raise during any given price control period and (ii) the upper limit for an effective fixed rate of debt

required by the hedging policy under the senior finance documents (e.g. 105% of the notional amount of senior debt to hedged onto an effective fixed rate).

Note: parameters in clause 17(b) subject to further consideration.

Note: DA requirements subject to further consideration to take into account the potential future requirements of the Regulator. Consideration to form part of wider discussions on financing plan.

Annex 1

Severe Sensitivity Assumptions

Product	Severe sensitivities
Interest rate Pre-hedging	Any new GBP interest rate swap entered into to pre-hedge expected raising of senior debt will be run through a market scenario using a reduction of each value in the SONIA OIS swap curve to $[\bullet]$ % of that value for the Relevant Period.
Interest rate hedging	Any new GBP interest rate swap entered into will be run through a market scenario using a reduction of each value in the SONIA OIS swap curve to $[\bullet]$ % of that value for the Relevant Period.
Inflation hedging	Any new inflation swap transaction will be run through a market scenario showing the year on year inflation rate rising at the higher of (i) the rate of inflation by reference to then current prevailing [inflation index] plus [\bullet]% or (ii) [\bullet]%, in each case for the Relevant Period.
Currency hedging	Any new cross currency hedging transaction will be run through a market scenario showing highest forecast of each of (i) $[\bullet]$ % change in the Relevant Exchange Rate and (ii) $[\bullet]$ % change in interest rate spread between GBP and Relevant Currency for the Relevant Period.

For the purposes of these Severe Sensitivity Assumptions, there shall be no assumed offsetting correlation between these sensitivities and the relevant calculation shall be run on a disaggregated basis.

For the purposes of any calculation made in connection with this Approved Hedging Policy these sensitivities will be applied to both the existing hedging portfolio and any Contemplated Hedging Transaction.

Definitions

"Relevant Currency" means the underlying currency in respect of the senior debt.

"Relevant Exchange Rate" means the foreign exchange rate between GBP and the Relevant Currency.

"Relevant Period" means the time remaining to the maturity in respect of the hedging transaction.

Part 2: Agreed Financing Principles

The financing of T&SCo in connection with the Project shall be in accordance with the following principles:

- 1. The total sources of finance for the Project will comprise senior debt from external funders, investors and/or lenders, together with financing directly or indirectly representing investment by shareholders.
- 2. Financing of the Project (including security and intercreditor arrangements) will be substantially consistent with the principles and market practice from time to time for the financing of UK-based [major] infrastructure assets (including those regulated by statutory regimes equivalent to or comparable with *[insert title of Act of Parliament under which T&SCo will be licenced]*) other than the extent to which the financing structure at Licence Award modifies such principles and market practice to reflect the Project and the GSP (together, the "Agreed Financing Principles").
- 3. Senior debt will be raised (i) on the basis of prudent treasury management, including as to pricing, tenors and management of refinancing risk, (ii) consistent with requirements of the Licence, (iii) and on arm's length and prevailing market terms, in each case consistent with the Agreed Financing Principles taking into account the performance of the Project. Where senior debt has been raised on a fixed rate or [inflation]-linked basis, it may contain early redemption/prepayment fees or premia (such as 'Spens' or make-whole payments) and such payments will constitute Breakage Costs for the purposes of the Discontinuation Agreement provided that the requirements of paragraphs 12 to 14 of Part A are satisfied.
- 4. The Relevant Obligors will finance the senior debt component of the capital structure through a combination of one or more sources of debt including:
 - (a) revolving credit facilities;
 - (b) term loan facilities;
 - (c) liquidity facilities;
 - (d) private placement loans/notes; and
 - (e) listed bonds,

in each case consistent with the terms of paragraphs 2 and 3.

- 5. Unless otherwise managed through the issuance of senior debt, potential interest, inflation and currency exposures of the Relevant Companies (as defined below) in relation to financial indebtedness and/or revenues of the Project will be managed by hedging transactions entered into by the Relevant Obligors subject to the terms of the Finance Hedging Policy and the Approved Hedging Policy.
- 6. The financing of the Relevant Companies shall not include:
 - (a) raising debt (other than Equity Support (as defined in the finance documents)) which benefits from third party credit enhancement;
 - (b) raising any debt (other than shareholder debt) by any Relevant Companies which is structurally or contractually junior or subordinated to senior debt of any Relevant Obligor and which is not included in the [Senior RAR] and [Net Debt/Adjusted RAV financial ratio calculation] (in each case, as defined in the finance documents); or

- (c) secured creditor rights of enforcement which, save as originally contemplated by the terms of the finance documents, are inconsistent with the Agreed Financing Principles; or
- (d) the senior leverage/gearing default ratio of the Relevant Companies exceeding [80]% of RAV.
 For these purposes "Relevant Companies" means [●] and each of its subsidiaries.

Note: to be confirmed depending on agreed financing structure.

- 7. The Relevant Obligors shall not propose or agree an amendment (an **"Amendment"**) to the terms of the finance documents if the effect of such Amendment would be inconsistent with these Agreed Financing Principles.
- 8. T&SCo shall procure that FinCo shall only enter into (and shall not propose or agree to any amendment which would conflict with) financing arrangements pursuant to [the FinCo Transaction Documents (as defined in the Master Definitions Agreement)] that are consistent with (i) the Agreed Financing Principles and (ii) the other terms of the Discontinuation Agreement, including this schedule, as such terms relate to the financing arrangements and terms applicable to T&SCo and the other Relevant Companies.
- 9. Relevant Obligors must not make any amendments to its finance documents, enter into any new finance documents or exercise any rights [of acceleration] under its finance documents in anticipation of Discontinuation without the approval of the SoS.

Note: this provision is designed to deal with the period leading up to an anticipated Discontinuation where the SoS will be sensitive to changes in the financing arrangements that would adversely impact its liability for discontinuation compensation. This remains subject to review of the T&SCos finance plan.

Appendix 2 to section D (Discontinuation Agreement): Template Discontinuation Plan

Note: Template Discontinuation Plan remains under development.

The Template Discontinuation Plan agreed at, and to be developed by the Parties following, Licence Award shall include the following items as a minimum:

Item	Requi	irements
Discontinuation	details around the event(s) leading to Discontinuation;	
Make-safe	(a)	details of what immediate make-safe activities the T&SCo intends to undertake prior to application of the Approved Offshore Decommissioning Programme and the Onshore Decommissioning Plan;
	(b)	details of the impact of make safe activities pursuant to (a) above on the T&S Network;
Access	(a)	access right requirements to allow T&SCo to carry out the make safe activities;
		a timetable for make safe activities which includes key milestones and schedule payments of any portion of compensation withheld in accordance with the discontinuation agreement;
Cost / funding	(a)	a cost estimate for the make safe activities;
	(b)	how make safe activities will be funded;
Staged release of Equity Compensation	(a)	as specified by the SoS, the details of any staged release of Equity Compensation pursuant to section 19 (<i>Approved Discontinuation Plan</i> <i>Obligations and Equity Compensation Retention</i>) of the Discontinuation Agreement;
Other	(a)	[other]

E. LIAISON AGREEMENT

No	Heading	Description		
1.	Parties	1.1	The Parties to the Liaison Agreement will be:	
			(a) SoS; and	
			(b) T&SCo	
2.	Conditions Precedent	2.1	Detailed conditions precedent to the effectiveness of the Liaison Agreement will be specified in the full form documentation – see an outline list at Appendix 1 to this section E (<i>Liaison Agreement</i>). Such conditions precedent are expected to track the conditions precedent to the award of the Licence by the Regulator on and by Licence Award.	
		2.2	T&SCo shall notify the SoS when it is satisfied that the conditions precedent have been satisfied in a form and substance satisfactory to it.	
		2.3	Following receipt of T&SCo's notice pursuant to section 2.2, if the SoS is of the view that the conditions precedent have been satisfied (acting reasonably), it shall give notice to T&SCo, that the conditions precedent have been satisfied.	
3.	Term	3.1	Subject to section 2 (<i>Conditions Precedent</i>), the Liaison Agreement will commence on Licence Award and continue until the expiry of the Discontinuation Agreement (the "Term").	
		3.2	In the event of a transfer of T&SCo pursuant to the terms of the Licence, legislation or otherwise in accordance with Law, the Liaison Agreement will provide that all of the [Project Documents and] Support Documents will be transferred at the same time, in order to ensure that the integrity of the contractual structure agreed at Licence Award is maintained.	
4.	Changes to Project Documents	4.1	T&SCo shall not without the prior written consent of the SoS:	
			 (a) enter into, vary, amend, replace, supplement or terminate all or any part of any Project Document which would be reasonably likely to have a Material Adverse Effect; or 	
			(b) notwithstanding section 4.1(a), enter into a Project Document where the proposed counterparty does not have the legal capacity, financial resources or technical competence to	

No	Heading	Description			
			perform the rights of the counterparty under the replacement Project Document.		
5.	Variations to the Approved T&S Network	5.1	T&SCo must obtain the prior written consent of the SoS, where a Change in Scope results in:		
			 (a) an extension of the asset life of the Approved T&S Network that results in an extension of the period of application of Allowed Revenue; 		
			(b) an expansion or upgrade to the Approved T&S Network that would result in an increase in the RAV or SRAV and associated increase in		
			Allowed Revenue by $[\bullet]$; or		
			(c) the development of a new Storage Complex that does not form part of the Approved T&S Network and/or increase in permitted flow rate/capacity for an existing Storage Complex that forms part of the Approved T&S Network,		
			("Variation").		
			Note: the definition of Variation remains subject to development in conjunction with the APDP and as part of ESNZ's management of change review.		
6.	Liaison Committee	6.1	The Parties shall establish and maintain a liaison committee (the "Liaison Committee"), consisting of the representatives appointed by each Party pursuant to section 7 (<i>Representatives and personnel</i>).		
		6.2	T&SCo shall report on a quarterly basis to the Liaison Committee as set out at section 8 (<i>Reporting and</i> <i>information requirements – Project Update Report</i>).		
		6.3	The Liaison Committee shall meet to discuss the Project Update Report on a quarterly basis and no earlier than [five (5) Business Days] following T&SCo sharing the Project Update Report with the SoS.		
		6.4	Each Party shall have the right (but not an obligation) to invite any Relevant Regulator and/or the RSA Counterparty to attend any meeting of the Liaison Committee, provided that not less than [five (5) Business Days'] written notice is given to the other Party prior to such meeting(s). Such Relevant Regulator(s) and/or the RSA Counterparty shall be entitled to attend such meeting(s) in an observer capacity only.		
		6.5	T&SCo shall also notify the Liaison Committee of any enhanced reporting requirements in accordance with		

No	Heading	Description		
			section 10 (<i>Reporting and information requirements – enhanced reporting</i>).	
7.	Representatives and personnel	7.1	Roles of the representatives of the Parties to the Liaison Agreement are to be developed, including delegated authority.	
		7.2	T&SCo representative shall:	
			 (a) manage and oversee compliance by T&SCo with its duties and obligations under the Liaison Agreement; and 	
			(b) exercise any right and perform any obligation exercisable by or to be performed by the T&SCo representative under the Liaison Agreement.	
		7.3	SoS representative shall:	
			 (a) manage and oversee compliance by SoS with its duties and obligations under the Liaison Agreement; and 	
			(b) exercise any right and perform any obligation exercisable by or to be performed by the SoS representative under the Liaison Agreement.	
8.	Reporting and information requirements – Project Update Report	8.1	T&SCo shall prepare and circulate a report on a quarterly basis to the Liaison Committee and the Inter-Regulator Forum on any issues which may have a material impact on the Project, including the requirements set out in Appendix 3 as such requirements may be updated by the SoS from time to time (the "Project Update Report").	
9.	Reporting and information requirements – additional information requests	9.1	The SoS (acting reasonably) may from time to time request T&SCo to include additional information in the Project Update Report. T&SCo will provide an index of such additional information.	
		9.2	The T&SCo representative shall address such information requests and shall:	
			 (a) respond to SoS acknowledging such request promptly and in any event within [five (5)] Business Days; and 	
			 (b) provide the information requested to SoS where possible (acting reasonably) within [ten (10)] Business Days. 	

No	Heading	Description			
10.	Reporting and	10.1	In the	e event that:	
	information requirements – enhanced reporting		(a)	claims, individually or in aggregate with all other previous claims, under the SCA reach a specified threshold (which shall be lower than the SCA Discontinuation Threshold);	
				Note : threshold remains subject to further consideration by ESNZ.	
			(b)	aggregate claims for Difference Payments under the RSA are forecast in the next three Charging Years to exceed a specified threshold (which shall be lower than the RSA Discontinuation Threshold);	
				Note: threshold remains subject to further consideration by ESNZ.	
			(c)	an Insured Risk occurs;	
			(d)	commercial insurance becomes Unavailable;	
			(e)	T&SCo receives a written notice from a Relevant Regulator requiring steps to be taken by T&SCo to remedy or prevent a contravention in relation to the Storage Complex;	
			(f)	a Remedy Event occurs; or	
			(g)	[others],	
			Inter- practi Days prepa the Li updat	o shall notify the Liaison Committee and the Regulator Forum as soon as reasonably cable and in any event within [five (5)] Business of the above events occurring and T&SCo shall re and circulate a report on a monthly basis to aison Committee and the Inter-Regulator Forum ting on the status the above issues and the ct on the Project until such issue is resolved.	
11.	Reporting and information requirements – reporting failures	11.1	1.1 In the event that T&SCo fails to comply with any of its reporting obligations set out in sections 8 (Reporting and information requirements – Project Update Report) to 10 (Reporting and information requirements), SoS shall have the right to issue a formal warning notice to T&SCo.		
		11.2	repor	e T&SCo has received a warning notice, the ting failure shall be escalated to the senior gement of T&SCo.	

Heading	Description			
	11.3	The receipt by T&SCo of two warning notices within a given Periodic Review period under the Licence shall constitute a Remedy Event.		
Qualifying Acquisition	Qualify	ifying Acquisition		
	12.1	For the purposes of the Liaison Agreement, "Qualifying Acquisition" means an event in which a person:		
		(a) acquires direct or indirect control of T&SCo or		
		(b) acquires direct or indirect control of assets which fall within the regulatory ringfence under the Licence.		
	12.2	For the purposes of section 12.1(a) and subject to section 12.4, "control" means either		
		(a) where the percentage of shares or voting rights that the person holds in T&SCo increases:		
		(i) from 25% or less to more than 25%;		
		(ii) from 50% or less to more than 50%; or		
		(iii) from less than 75% to 75% or more, or		
		(b) where the acquisition is of voting rights in T&SCo that (whether alone or together with other voting rights held by the person) enable the person to secure or prevent the passage of any class of shareholder resolution governing the affairs of T&SCo with respect to all or substantially all matters relating to the T&S Business.		
	12.3	For the purposes of section 12.1(b) and subject to section 12.4, "control" means the acquisition of a right or interest such that the person is able:		
		(a) to use the asset, or use it to a greater extent than prior to the acquisition; or		
		(b) to direct or control how the asset is used, or direct or control how it is used to a greater extent than prior to the acquisition.		
	12.4	For the purposes of section 12.1, " indirect control " means where the person does not directly hold the relevant right or interest, but holds a majority stake in an entity and that entity:		
		(a) holds the interest or right; or		
		Qualifying Acquisition Qualifying 12.1 12.1 12.2 12.2		

No	Heading	Description		
			(b)	is part of a chain of entities and either (i) each of those entities other than the last has a majority stake in the entity immediately below it in the chain, and the last entity holds the interest or right, or (ii) if the last entity in the chain holds a majority stake in the T&SCo, all but one of the other entities in the chain hold a majority stake in the entity immediately below it in the chain, and the other entity in the chain holds more than 25% of the shares or voting rights in the entity immediately below it.
		12.5		e purposes of section 12.4, A has a " majority " in B if:
			(a)	A holds a majority of voting rights in B,
			(b)	A is a member of B and has the right to appoint or remove a majority of the board of directors of B,
			(c)	A is a member of B and controls alone, pursuant to an agreement with other shareholders or members, a majority of the voting rights in B, or
			(d)	A has the right to exercise, or actually exercises, dominant influence over B for the purposes of section 1162(2)(c) and paragraph 4 of Schedule 7 the Companies Act 2006,
			where than 5	e "majority" means a simple majority of more 50%.
		12.6	In sec	tion 12.5:
			(a)	the reference to the right to appoint or remove a majority of the board of directors of an entity is to the right to appoint or remove directors holding a majority of the voting rights at meetings of the board on all or substantially all matters,
			(b)	the reference to the board of directors, in the case of an entity that does not have such a board, is to be read as a reference to the equivalent management body of that entity.
		<u>Notifica</u>	<u>tion of</u>	a Qualifying Acquisition
		12.7	T&SC	o shall
			(a)	inform the SoS in writing; and

No	Heading	Description		
			 (b) provide the information set out in Appendix 4 to this Section E (Liaison Agreement) of which it is aware or ought reasonably to be aware, having made reasonable enquiries, 	
			not less than [thirty (30)] Business Days in advance of completion of any Qualifying Acquisition of which it is aware or ought reasonably to be aware, including in light of the provisions of section 12.8.	
		12.8	T&SCo shall procure from each company or other person which T&SCo knows or reasonably should know is at any time an Ultimate Controller of T&SCo a legally enforceable undertaking in favour of T&SCo that the Ultimate Controller (the "information covenantor") will give to T&SCo, and will procure that any person (including, without limitation, a corporate body) which is a subsidiary of, or is controlled by, the information covenantor (other than T&SCo and its subsidiaries) will give to T&SCo all such information as may be necessary (and is within its power to give) to enable T&SCo to comply fully with the obligations imposed on it by paragraph 12.7. Such undertaking shall be obtained upon such corporate body or other person in question becoming an Ultimate Controller of T&SCo and shall remain in force for so long as T&SCo has a Licence under the Act and the information covenantor remains an Ultimate Controller of T&SCo.	
13.	Remedy and Failure Event regime	13.1	The Remedy Events are defined to capture T&SCo's failures which take the T&S Network outside the basis for the SoS's offer of the GSP. They are therefore defined by reference to events which, of themselves or subject to having the requisite Material Adverse Effect, represent a shift outside the basis for that offer – see definition of Remedy Events outlined in Appendix 2.	
		13.2	Upon the occurrence of a Remedy Event there will be a restriction on Distributions until the Remedy Event (and associated Failure Event) is Remedied.	
		13.3	Following the occurrence of a Remedy Event, T&SCo will have the opportunity to remedy the Remedy Event or agree a Remediation Plan with the SoS, in each case within a prescribed period. It should be noted in particular that where remediation of a Remedy Event is not reasonably practicable, the Remediation Plan will only require mitigation of the Remedy Event, rather than remediation (see paragraph 3 of Appendix 2).	

No	Heading	Description			
		13.4	If a Failure Event arises, then this has the following consequences:		
			(a) where the Failure Event has arisen as a consequence of breach by T&SCo of any provision of the Supplemental Compensation Agreement or breach of the Storage Permit, escalation of the fee and additional retentions under the Supplemental Compensation Agreement; and/or		
			(b) any project spend:		
			 (i) which is incurred during the period that a Failure Event has occurred but has not been remedied or reduced to a Remedy Event; and 		
			(ii) which qualifies for assignment by the Regulator to the RAV under the Licence,		
			shall be deducted from the Total Compensation Amount by the SoS pursuant to the Discontinuation Agreement (and such deductions shall cease to apply where a Failure Event is remedied or reduced to a Remedy Event).		
14.	Transfer Scheme	14.1	Where the Regulator is entitled to revoke the Licence, T&SCo acknowledges that a Transfer Scheme may be applied.		
		14.2	T&SCo hereby provides its consent to the operation of any such Transfer Scheme on the basis that the amount of the consideration paid to the T&SCo by the transferee, by way of compensation to the T&SCo for the transfer under the Transfer Scheme, is calculated as follows:		
			 (a) if revocation is as a result of Discontinuation, compensation will be provided for as set out under the Discontinuation Agreement; or 		
			(b) if revocation is as a result of any other reason, compensation will be determined by an independent valuer (who must be a qualified insolvency practitioner from an international professional services firm and who may rely on valuation analysis prepared by the international professional services firm and/or an international investment bank) appointed for such purpose by the SoS and such		

No	Heading	Description				
			deterr princi		n shall be based on the	e following
			(i)	has be	the undertaking of T& en transferred as a goin a Transfer Scheme:	
				(A)	entered a special adm process immediately revocation of the Lio	n member SCo would eceived if tion of the d instead hinistration before the cence and hinistration n a transfer
				(B)	the compensation p any) to T&SCo for pay creditors and mem applicable) should re worse outcome in terms for each me creditor as a resul Transfer Scheme thar of the valuation app them in limb (A) above	ment to its abers (as sult in no monetary mber and it of the a the result plicable to
				(C)	in undertaking the referred to in 14.2(b)(i)(A) abov independent valuer sh	
					(aa) assume that th have been a wi out of administration T&S transfer sc	lling buyer special through a
					(bb) assume that transfer scher have included limitation) the GSP, RSA and T and	(without Licence,
					Note: separate will also be inco stapling provise GSP/RSA.	rporating a

No	Heading	Description
		(cc) assist with (on behalf of the Secretary of State) any consultation, concerning matters directly relevant to the provision the Transfer Scheme should make as regards compensation for the T&SCo, as between the Secretary of State with the Regulator (including in particular regarding any assumptions the independent valuer should make concerning the conditions of the Licence relevant to the post- transfer revenue that the may have been made either following a T&S transfer scheme), the Health and Safety Executive, the NSTA, the appropriate devolved authorities (if any), and such other persons as the Secretary of State considers appropriate; and
		(D) when undertaking the valuation referred to in paragraph 14.2(b)(i)(A):
		(aa) the independent valuer shall not include in its calculation the remuneration and disbursements of the special administrator which would only have been incurred as a result of a special administration;
		(bb) any assumed expenses of the assumed special administration that would have fallen within paragraph 99(3) or paragraph 99(4) of Schedule B1 of the Insolvency Act 1986 (as

No	Heading	Descrip	otion		
				financial assistar may have been by the Secretar to the T&SCo	rgy Bill]), without osts of its or any nce which a provided y of State o (or its inistrator) assumed
				in each case wh arise under the Scheme, and arisen, on the no special adn has been appoin	e Transfer have not basis that ninistrator
			(ii)	where the undertaking of T&S transferred as a going concer Transfer Scheme, the ind valuer shall determine com payable by reference to the ou each shareholder and member T&SCo entered a compulsory i immediately prior to revocation Licence;	n under a dependent pensation utcome for er had the liquidation
			(iii)	references to "as a going co paragraph (ii) 14.2(b)(i) at mean that the Transfer Sch made with the objective so section 50(2)(a) of the Ener securing that the activities a by the Licence are continue transferee (or if the transfer is more transferees, by those to in aggregate).	oove shall neme was et out in rgy Bill of authorised ed by the s to two or
		14.3	entitled to is T&SCo may property, rig concerning a complying w (and subject	owledges that where the Ressue a T&SCo of Last Resort be the intended transferee of conts or liabilities under a Transferent nother CCS Licensee. For the print any T&SCo of Last Resort to its terms), T&SCo hereby print to a Transfer Scheme.	Direction, lesignated er Scheme urposes of : Direction
		14.4	valuer under	tions to be applied by the ind clause 14.2 concerning a hy inistration in the event of a	/pothetical

No	Heading	Description		
			Scheme shall not restrict in any way, or fetter the discretion of, the Secretary of State during the course of an actual special administration.	
15.	Dispute Resolution Procedure	15.1	Any dispute arising out of or in connection with the Liaison Agreement, including in relation to any amounts payable under or pursuant to the Liaison Agreement (a "Relevant Dispute"), shall be resolved in accordance with the Dispute Resolution Procedure set out in this section 15.	
		15.2	In the event of a Relevant Dispute, the matter shall first be referred to the Liaison Committee for resolution by the Parties.	
		15.3	If the Relevant Dispute is not resolved by the Liaison Committee after a period of [ten (10)] Business Days on and from such referral (or such other period as agreed between the Parties in writing), either Party shall be entitled to refer the Relevant Dispute to identified senior representatives for resolution. If the Relevant Dispute is not resolved by the senior representatives after a further period of [ten (10)] Business Days on and from such referral to identified senior representatives (or such other period as agreed between the Parties in writing), then:	
			 (a) the Parties may agree to refer the Relevant Dispute for expert determination; or 	
			(b) either Party may refer the Relevant Dispute to the courts of England and Wales.	
16.	Limits of Liability	16.1	There will be provisions relating to:	
			(a) exclusion of liability for consequential/indirect losses; and	
			(b) remedies for breach under the Liaison Agreement or Support Documents.	
17.	Assignment and Sub- Contracting	17.1	The Liaison Agreement will include restrictions on assignment, transfers and sub-contracting the benefit or burden of any right or obligation under the Support Documents in whole or in part except with the prior written consent of the other relevant parties (such consent not to be unreasonably withheld or delayed).	
		17.2	The SoS may, subject to the conditions to be set out in the Liaison Agreement, transfer its rights or novate its obligations under the Liaison Agreement to any	

No	Heading	Description		
		Minister of the Crown or any entity directly wholly- owned by a Minister of the Crown.		
18.	Confidentiality	18.1 The Liaison Agreement will set out confidentiality provisions which will apply to all Parties in respect of the Project. There will be customary exceptions.		
		18.2 The Liaison Agreement sets out that the SoS is, and that T&SCo may become subject to, the requirements of the FOIA and the Environmental Information Regulations and each Party shall facilitate compliance by the other Party with its information disclosure requirements pursuant to the FOIA and the Environmental Information Regulations.		
19.	Survivorship	19.1 Following the expiry of the Liaison Agreement each Party shall cease to have rights or obligations under the Liaison Agreement, save that the rights under section 13 (<i>Remedy and Failure Event regime</i>), section 14 (<i>Transfer Scheme</i>), section 15 (<i>Dispute</i> <i>Resolution Procedure</i>), section 16 (<i>Limits of Liability</i>), section 18 (<i>Confidentiality</i>) and section 21 (<i>Governing</i> <i>Law</i>) will continue in full force and effect.		
20.	Rights of Third Parties	20.1 A person who is not a Party to the Liaison Agreement shall have no rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any of its terms.		
21.	Governing Law	21.1 The Liaison Agreement shall be governed by and construed in accordance with English law and the courts of England and Wales shall have exclusive jurisdiction.		
22.	Definitions	In this section E (<i>Liaison Agreement</i>), terms defined in the Licence heads of terms shall have the same meaning and:		
		"Approved Discontinuation Plan" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);		
		"Approved T&S Network" has the meaning given to it in the Licence heads of terms, as at Licence Award;		
		"Allowed Revenue" has the meaning given to in the Licence and as it relates to the Approved T&S Network;		
		"CCS Network Code" means the document designated as such and required to be maintained by the licensee under the Licence;		
		"Commercial Insurances" has the meaning given to it in section C (<i>Supplemental Compensation Agreement</i>);		

No	Heading	Description
		"Difference Payments" has the meaning given to it in the RSA heads of terms;
		"Discontinuation" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);
		"Discontinuation Agreement" means the agreement between SoS and T&SCo setting out the discontinuation events, the consequences of discontinuation and how discontinuation compensation is to be calculated;
		"Dispute Resolution Procedure" means the procedure for the resolution of disputes set out in section 15 (<i>Dispute</i> <i>Resolution Procedure</i>) of this section E (<i>Liaison</i> <i>Agreement</i>);"Environmental Information Regulations" means the Environmental Information Regulations 2004 (as amended);
		"Failure Event" has the meaning given to it in Appendix 2 of this section E (Liaison Agreement);
		"Finance Documents" means the agreements in relation to the financing of T&SCo and any subsequent agreements or amendments to such agreements entered into by T&SCo in relation to its indebtedness and any other document designated by the SoS (acting reasonably) as a Finance Document from time to time;
		"FOIA" means the Freedom of Information Act 2000 (as amended);
		"GSP" means the Government Support Package, comprising the Supplemental Compensation Agreement, the Discontinuation Agreement and the Liaison Agreement;
		"FOIA" means the Freedom of Information Act 2000 (as amended);
		"Inter-Regulator Forum" means the industry forum made up of T&SCo, the SoS and the following:
		(a) [the Regulator;
		(b) NSTA;
		(c) OPRED;
		(d) HSE;
		(e) RSA Counterparty; and
		(f) [<i>others</i>]];

No	Heading	Description
		Note : the establishment, scope and role of an Inter-Regulator Forum remains subject to further consideration by ESNZ.
		"Law" means any enactment or subordinate legislation, rule, regulation, order, directive or other provision which has force of law in the United Kingdom;
		"Liaison Committee" means the liaison committee established pursuant to section 6 (<i>Liaison Committee</i>) of this section E (<i>Liaison Agreement</i>);
		"Licence" means the economic licence to be granted under HM Government's CCUS programme pursuant to section 7 of the Act;
		"Material Adverse Effect" has the meaning given to it in Appendix 2 of this section E (<i>Liaison Agreement</i>);
		"Minister of the Crown" shall have the meaning given to it in the Minister of the Crown Act 1975;
		"Project Documents" means any of the following agreements entered into, or to be entered into, by T&SCo in respect of the T&S Network:
		(a) [operating contracts;
		(b) crown estate leases;
		(c) permits and consents;
		(d) insurances[, whether required to be procured by T&SCo by Law, the Licence, the GSP, or otherwise];
		(e) [<i>others</i>]],
		Note : list of Project Documents remains under review by ESNZ.
		"Project Update Report" has the meaning given to it in section 8 of this section E (<i>Liaison Agreement</i>);
		"Qualifying Acquisition" has the meaning given to it in section 12 (<i>Qualifying Acquisition</i>) of this section E (<i>Liaison Agreement</i>);
		"RAV" has the meaning given to in the Licence, as it relates to the Approved T&S Network;
		"Relevant Dispute" has the meaning given to it in section 15.1;
		"Relevant Regulator" means the NSTA, OPRED, HSE, or Ofgem;

No	Heading	Description
		"Remediation Plan" has the meaning given to it in Appendix 2 of this section E (<i>Liaison Agreement</i>);
		"Remedy Event" has the meaning given to it in Appendix 2 of this section E (<i>Liaison Agreement</i>);
		"Remedy Event Notice" means a notice of the Remedy Event served by SoS on T&SCo
		"Revenue Support Agreement" or "RSA" means the revenue support agreement;
		"RSA Discontinuation Threshold" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);
		"SCA Discontinuation Threshold" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);
		"SRAV" has the meaning given to in the Licence, as it relates to the Approved T&S Network;
		"Storage Complex" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Complex is identified as part of the Approved T&S Network;
		"Storage Permit" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Permit relates to the Approved T&S Network;
		"Storage Site" has the meaning given to that term in the Carbon Dioxide Storage Regulations and as such Storage Site is identified as part of the Approved T&S Network;
		"Support Documents" means the GSP and the RSA;
		"Template Discontinuation Plan" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);
		"Term" has the meaning given to it in section 3 (<i>Term</i>) of this Section E (<i>Liaison Agreement</i>);
		"Total Compensation Amount" has the meaning given to it in section D (<i>Discontinuation Agreement</i>);
		"Unavailability" has the meaning given to it in section C (Supplemental Compensation Agreement); and
		"Variation" has the meaning given to it in section 5 (<i>Variations to the Approved T&S Network</i>) of this Section E (<i>Liaison Agreement</i>).

Appendix 1: Outline Conditions precedent to the GSP

Note: this list of conditions precedent is a high-level list only, offering examples of the categories of conditions precedent that may be required. This list is expected to develop as the business model (and User business models) progress.

Α.	T&S network	Examp	les
1.	Legislation	1.1	New legislation required for implementation of the CCUS programme
2.	Licence	2.1	Appointment of Regulator
		2.2	Grant of the Licence
3.	CCS Network Code	3.1	Establishment and maintenance of the CCS Network Code
		3.2	T&SCo becoming legally bound by the CCS Network Code by signing or acceding to the Code Agreement
4.	Revenue Support	4.1	Appointment of RSA Counterparty
		4.2	Award of RSA
5.	Property and planning	5.1	Planning approval and conditions
		5.2	Land agreements
		5.3	Asset protection agreements
6.	Licences and consents	6.1	Storage Permit
		6.2	Grant of rights from the Crown Estate and/or Crown Estate Scotland
		6.3	UK ETS permit
		6.4	Other
7.	Mandated requirements	7.1	[Any specific mandated requirements have been met]
8.	Equity	8.1	Evidence that T&SCo has been established
		8.2	T&SCo governance arrangements (including shareholders agreement) in satisfactory form

9.	Finance	9.1	Finance Documents including intercreditor arrangements in satisfactory form
10.	Model/tax	10.1	Model/tax audit of base case financial model, including sensitivity checks
11.	Sub-contractors	11.1	Appointment of works contractor(s) and professional team
		11.2	Appointment of operator contractor(s)
		11.3	Appointment of other key sub-contractor(s)
		11.4	Execution of sub-contracts
12.	Technical	12.1	Satisfactory conclusion of technical assessment of costs/returns for Licence
		12.2	Satisfactory assurance of store performance, including (but not limited to) a "Competent Person's Report" (CPR) delivered by an independent auditor, which shall be subject to SoS approval of:
			(a) the independent auditor selected by T&SCo to conduct the CPR; and
			(b) the final form of CPR furnished to the SoS by (or on behalf of) T&SCo
13.	Insurance	13.1	Placements and commitments in respect of the Commercial Insurances (which excludes insurances that are Unavailable at Licence Award) in accordance with sections 3.3 and 3.4 of the SCA
		13.2	Broker's undertakings
		13.3	Lead Insurer Agreement
14.	Inter-regulator Forum	14.1	Establishment of Inter-regulator Forum
15.	Formalities	15.1	Approvals of contract counterparties
		15.2	Legal opinions in satisfactory form
		15.3	Adviser due diligence reports in satisfactory form
		15.4	Evidence of satisfaction of know your customer checks
	User business models	Indica	ative CPs

16.	Power	16.1	Awarded DPAs, including any associated conditions precedent
17.	Industry	17.1	Awarded ICCCs, including any associated conditions precedent
18.	Hydrogen	18.1	Awarded support contract, including any associated conditions precedent
19.	Bioenergy Carbon Capture System (BECCS)	19.1	Awarded support contract, including any associated conditions precedent
20.	Direct Air Carbon Capture System (DACCS)	20.1	Awarded support contract, including any associated conditions precedent

Appendix 2: Remedy and Failure Event regime

No.	Item	Comm	ercial T	erms
1.	Remedy Event	1.1	"Rem	edy Event" means:
			(a)	breach of an obligation of T&SCo under the Liaison Agreement which has or is reasonably likely to have a Material Adverse Effect;
			(b)	breach of an obligation of T&SCo under the Support Documents which has or is reasonably likely to have a Material Adverse Effect;
			(c)	breach of a material obligation of T&SCo under its Storage Permit which results in a written notice from NSTA to T&SCo
			(d)	failure to maintain Commercial Insurance or to comply in all material respects with the renewal process where there is no Unavailability pursuant to the Supplemental Compensation Agreement;
			(e)	failure to (i) market test the insurance market or (ii) comply in material respects with the insurance market testing process, pursuant to the terms of the Supplemental Compensation Agreement;
			(f)	failure to issue a Qualifying Acquisition Notice to the SoS pursuant to the terms of the Liaison Agreement;
			(g)	breach by T&SCo of any restriction on making Distributions;
			(h)	receipt of two warning notices within a given Periodic Review period under the Licence, validly issued pursuant to section 11 (<i>Reporting and</i> <i>information requirements – reporting failures</i>) of the Liaison Agreement;
			(i)	breach by T&SCo of the Agreed Financing Principles and/or the Approved Hedging Policy;
			(j)	failure to maintain tax residency in the United Kingdom; or
			(k)	[other].
			Note : ESNZ.	list of Remedy Events remains under review by
2.	Material Adverse Effect	2.1	"Mate effect	erial Adverse Effect" means a material adverse on:

No.	Item	Commercial Terms		
			(a) any right or liability (whether actual, potential or contingent) of the SoS; or	
			(b) the ability of the SoS to perform its obligations,	
			in each case under or in relation to the Support Documents or at Law; or	
			(c) the effect of materially increasing the likelihood of a call on any element of the Support Documents.	
3.	Failure Event	3.1	A "Failure Event" only arises where SoS has served a Remedy Event Notice on T&SCo and T&SCo has not remedied a Remedy Event and T&SCo fails:	
			 (a) to put forward or to agree a Remediation Plan (in accordance with section 4), in each case within a prescribed period; or 	
			(b) to comply in any material respect with the agreed Remediation Plan.	
4.	Remediation Plan	4.1	"Remediation Plan" means a programme for remedying or mitigating the Remedy Event (as applicable) within a prescribed period which is acceptable to the SoS, acting reasonably.	
		4.2	For the purposes of agreeing the Remediation Plan, the SoS and T&SCo shall have due regard to the following considerations:	
			 (a) the extent to which the event or circumstance giving rise to the Remedy Event is reasonably capable of being remedied; 	
			(b) the extent to which the event or circumstance giving rise to the Remedy Event is reasonably capable of being mitigated; and	
			(c) any alternative steps which may be appropriate having regard to the risks of the SoS under the GSP.	
		4.3	There will be a process for submission by the T&SCo of a draft Remediation Plan which will require the SoS, within a prescribed time period, to respond notifying that it accepts the draft Remediation Plan or the reasons why (acting reasonably) that it does not accept the draft Remediation Plan.	
		4.4	The agreed Remediation Plan will be considered at Liaison Committee meetings to ensure that it remains up-to-date and also to record progress against the plan. Where the Parties at a Liaison Committee meeting agree that the	

No.	Item	Commercial Terms			
		agreed Remediation Plan is not capable of being completed for reasons outside of the control of the Parties, the Parties will act in good faith to agree to amend the Remediation Plan (with such amendment(s) to be effected in accordance with section 4.5 below).			
		4.5	Any amendments proposed to the Remediation Plan shall be subject to agreement in writing by the Parties.		
5.	Remedied	5.1	A Remedy Event (and the associated Failure Event, if applicable) shall be treated as " Remedied " if:		
			(a) T&SCo remedies the Remedy Event;		
			(b) T&SCo completes the relevant Remediation Plan; or		
			(c) the SoS waives the Remedy Event.		
6.	Reduction of Failure Event to	6.1	If a Failure Event has arisen but:		
	Remedy Event		(a) where a Remediation Plan has not previously been agreed, the SoS and T&SCo have subsequently agreed a Remediation Plan and the SoS is satisfied (acting reasonably) that T&SCo is making substantial progress in carrying out the Remediation Plan; or		
			(b) where a Remediation Plan has previously been agreed but T&SCo has failed to comply in any material respect with the agreed Remediation Plan, the SoS is subsequently satisfied (acting reasonably) that T&SCo has resumed making substantial progress in carrying out the Remediation Plan,		
			the Failure Event will be reduced to a Remedy Event (noting that the restriction on Distributions will then continue until the Remedy Event is Remedied).		

Appendix 3: Project Update Report

Project specification	(a)	Any new significant risks or material changes to the remaining significant risks (as updated) identified in the Project Risk Register.
	(b)	Details of plans for new users to connect to the T&S Network or existing Users to disconnect from the T&S Network.
	(c)	Any proposed Variations.
	(d)	Details of any non-regulated activity in respect of the T&S Network.
Performance	(a)	Reasonable details of the performance of each of [the Project Documents and] the Support Documents, including any material breaches of [the Project Documents or] the Support Documents which have been made by any party thereto since the last report.
	(b)	Details of any Remedy Event and/or Failure Event.
	(c)	Details of any specified regulatory reports relevant to the containment of CO_2 and the monitoring of CO_2 including for the purposes of UK ETS compliance.
Finance	(a)	An update on the financing plan, including any proposed or contemplated equity or debt issuance (with the dates, amounts and principal terms of such issuances) and hedging arrangements.
	(b)	Details of any event of default or potential event of default or breach of cover ratios under the T&SCo's financing arrangements.
Insurance	(a)	Any change in the terms or availability of the Commercial Insurances (or any change that T&SCo foresees is likely to occur in respect of the terms or availability of the Commercial Insurances).
	(b)	Any detail relating to the renewal of insurances, carried out by T&SCo pursuant to section 11 (<i>Renewal of Insurances</i>) of the SCA.
	(c)	Details of any claims made under the Commercial Insurances (and in respect of claims made (but not settled) over a certain de minimis threshold, the status of those claims).
HMG Contingent Liabilities	(a)	Any relevant information relating to the management by HM Government of its contingent liabilities as part of its responsibilities for managing public finances.
SCA Discontinuation Threshold	(a)	T&SCo forecasts of the likelihood of triggering the SCA Discontinuation Threshold pursuant to the Discontinuation Agreement.
RSA Discontinuation Threshold	(a)	T&SCo forecasts of the likelihood of triggering the RSA Discontinuation Threshold pursuant to the Discontinuation Agreement.

The Project Update Report shall include the following:

Discontinuation Plan	(a)	Any updates required to the Approved Discontinuation Plan.
Qualifying Acquisition	(a)	Details in relation to any Qualifying Acquisition or proposed Qualifying Acquisition.
Information	(a)	Any information required as a result of the enhanced reporting requirements at section 10 (<i>Reporting and information requirements - Enhanced Reporting</i>) of section E (<i>Liaison Agreement</i>).
Other	(a)	Any additional information requested by the SoS pursuant to section 10 (<i>Reporting and information requirements – enhanced reporting</i>) of section E (<i>Liaison Agreement</i>).

Appendix 4: Qualifying Acquisition Notifications

Where T&SCo is required to notify the SoS of any Qualifying Acquisition pursuant to section 12 of the Liaison Agreement, the T&SCo shall provide in such notification the following information:

No.	Item	Detail
1	Acquirer contact details and related notifications	 a) number of all persons involved in the Qualifying Acquisition b) name of person(s) involved c) contact details of the person(s) d) details on any notification to any UK or overseas investment screening regimes that person (or persons) has submitted within the last 12 months
2	Acquisition details	 a) the percentage of the shares or voting rights that the person(s) holds in the entity b) expected date of the Qualifying Acquisition c) approvals required by UK regulators d) rationale for the Qualifying Acquisition
3	Entity details	 a) name and address of the entity b) contact details of the authorised individual in entity c) companies house registration number d) standard industrial classification SIC code. e) details on all licences held to operate within the UK. f) details on entity relationship with the UK Government h) pre-Qualifying Acquisition structure chart of the entity. i) details on any non-UK government which have a direct or indirect role in the operation or decision making of the entity.
4	Acquiring person(s) details	 a) name, address, and business registration details b) details on any non-UK government, or representative of any non-UK government, have share ownership or voting rights in the person(s). c) details on any non-UK government which have a direct or indirect role in the operation or decision making of the acquirer d) when the Qualifying Acquisition completes, details on if there will be any contractual arrangements in place regarding share ownership or voting rights between the person(s) and any other party. e) structure chart of person(s) which includes the % ownership and country of incorporation
5	Additional information	Other relevant documentation and information on the Qualifying Acquisition.

Annex C: Revenue Support Agreement

CCUS: Revenue Support Agreement for T&SCo

Introduction and Indicative Heads of Terms

Note: these indicative heads of terms are subject to the "Disclaimer" section at the front of the explanatory note document to which they are annexed.

A. INTRODUCTION

1.1 Background

- (a) The indicative heads of terms in section B (*Revenue Support Agreement*) of this document set out the basis upon which a revenue support agreement ("RSA") could be provided to a company licensed to provide transport and storage services ("T&SCO") as part of HM Government's ("HMG") CCUS Programme (the "Programme").
- (b) The contents of the indicative heads of terms in section B (*Revenue Support Agreement*) of this document are indicative only and do not constitute an offer by the RSA Counterparty and do not create a basis for any form of expectation or reliance. Any RSA arrangements that are developed in the future will be subject to approval by Ministers, in consultation with the economic regulator ("**Regulator**"), and the development and completion of necessary contractual documentation.
- (c) The introduction set out in this section A does not form part of the indicative heads of terms in section B (*Revenue Support Agreement*) of this document and is intended only to provide an overview of the rationale and assumptions for the provision of the RSA.

1.2 Rationale

- (a) In the development of the revenue model for the transport and storage ("T&S") network, HMG has identified that there may be certain demand-related revenue risks to T&SCo, where T&SCo's actual revenues may fall short of T&SCo's Allowed Revenue set by the Regulator under the economic regulatory regime ("ERR") or where the first User of the T&S Network joins later than expected. HMG has considered how these demand-related revenue risks will be mitigated and identified a number of risk mitigation mechanisms ("RMMs"). If the initial proposals for these RMMs are not sufficient, HMG has identified the RSA as a mechanism which will enable the recovery by T&SCo of:
 - (i) its Allowed Revenue in the event of a shortfall; and/or
 - (ii) unavoidable operating expenditure and [allowed cost of debt and equity and depreciation]¹ where the first User is delayed in joining the T&S Network.
- (b) The detailed terms of coverage of the RSA remain under consideration, but the indicative heads of terms in section B (*Revenue Support Agreement*) of this document set out current HMG thinking on the support which could be provided by the RSA. Final provisions will depend upon the final terms of the ERR and the target risk profile.

1.3 **Description**

Note: see section 8 below.

- (a) This document comprises the following:
 - (i) section A: an introduction to the RSA; and
 - (ii) section B: heads of terms for the RSA.
- (b) The scope of the areas of specific protection set out in section B (*Revenue Support Agreement*) of this document remains subject to review, in particular in the context of the evolving detail of the ERR.

1.4 **Initial assumptions**

- (a) This document is based on the following assumptions in respect of the T&S Network:
 - (i) ownership of the T&S Network: the "onshore" (i.e. Onshore Transportation Systems) and "offshore" (i.e. Offshore Transportation and Storage Systems) elements of the T&S Network will be owned by T&SCo. In particular, T&SCo will hold any Crown Estate (including Crown Estate Scotland) lease and licence/permit issued by the NSTA (or licensing authority of a devolved administration);
 - development and operation of the T&S Network: T&SCo will be responsible for the development, construction, operation and maintenance of the T&S Network, including obtaining all necessary permits and approvals;
 - (iii) expansion of the T&S Network: in the future T&SCo may need to expand the T&S Network under the terms of the ERR and will provide access to new Users of the T&S Network in accordance with the terms of its Licence, the CCS Network Code and all relevant laws and regulations;²
 - (iv) decommissioning: T&SCo will be responsible for the decommissioning and monitoring of:
 - (A) the Offshore Transportation and Storage System in accordance with the Petroleum Act 1998 as applied under the Energy Act 2008 (as may be amended or supplemented) and with regards to post closure and monitoring obligations, T&SCo will be responsible as set out in its storage permit and licence granted by the NSTA; and
 - (B) the Onshore Transportation System in accordance with the relevant decommissioning conditions of any development/planning consent;
 - (v) T&S charges: Users of the T&S Network will pay T&SCo charges for provision of T&S services which will be regulated under the ERR;
 - (vi) CO₂ ownership: title and risk in CO₂ delivered to the T&S Network will pass to T&SCo at the Delivery Point; and
 - (vii) government support package ("GSP"): T&SCo will benefit from a GSP provided by HMG to cover certain high impact, but low probability, risks beyond those which are manageable by operation of the ERR and the RSA, which the investors and/or supply chain, including insurers, of T&SCo cannot

²

Consideration is being given to changes required to the existing third party access rules as a result of the new regulatory regime.

take, or cannot price at an efficient level which is good value for money for UK taxpayers, consumers or Users.

1.5 **Defined terms**

Capitalised terms not otherwise defined in this document, including in section 19 (*Definitions*), have the meaning given to them in the Licence heads of terms.

B. REVENUE SUPPORT AGREEMENT

No.	Item	Commercial Terms		
1.	Parties	1.1. The Parties to the RSA will be:		
		(a) T&SCo and		
		(b) [Low Carbon Contracts Company Ltd] ("RSA Counterparty"),		
		each a "Party" together the "Parties".		
		Note: the minded to position is that the Low Carbon Contracts Company Ltd (LCCC) will be the RSA Counterparty. This remains subject to parliamentary approval of the Energy Bill and implementation of the RSA Counterparty designation process.		
2.	Rationale	2.1. The Revenue Support Agreement is intended to provide mitigation against demand-related risk of shortfall in actual revenues earned by T&SCo compared with the Allowed Revenue set by the Regulator under the ERR.		
3.	Term	3.1. The term of the Revenue Support Agreement shall commence on and from Licence Award and shall expire on the first to occur of:		
		(a) $[\bullet]$ (or such other date as the Parties may agree);		
		(b) Discontinuation; and		
		(c) revocation of the Licence.		
4.	Conditions precedent	4.1. T&SCo will be required to satisfy certain initial conditions precedent specified in the Revenue Support Agreement including, amongst other things:		
		 (a) evidence that it holds a Licence pursuant to which T&SCo is entitled to receive an Allowed Revenue; 		
		(b) a capacity legal opinion from its legal advisers;		
		 (c) evidence of satisfaction of know your customer checks; and 		
		(d) other conditions precedent, equivalent to those set out in Appendix 1 of the Liaison Agreement heads of terms.		
5.	Interim Difference Payment	5.1. The Interim Difference Payment shall be calculated as the amount by which the Adjusted Allowed Revenue is greater than the Adjusted Forecast Market Revenue.		
		Calculation and evidence		

No.	Item	Commercial Terms	
		 5.2. T&SCo shall submit to the RSA Counterparty its evidence of and calculations for the Interim Difference Payment for each Charging Yeart no later than four months prior to commencement of Charging Yeart. 	
		5.3. The evidence and calculations provided by T&SCo pursuant to section 5.2 shall include, as a minimum, the following in respect of each Interim Difference Payment:	
		(a) the methodology and input data used by T&SCo to calculate the relevant Interim Difference Payment;	
		(b) evidence of review, assurance or confirmation by the Regulator in relation to amounts required for the Interim Difference Payment in accordance with section 10 (<i>Regulator review, assurance or confirmation</i>); and	
		 (c) a certificate signed by a director or authorised signatory of T&SCo, certifying that the information provided to the RSA Counterparty pursuant to section 5.2 is true, complete and accurate. 	
		Assurance by the RSA Counterparty	
		5.4. Following receipt by the RSA Counterparty of the evidence of and calculations for the Interim Difference Payment in accordance with sections 5.2 and 5.3, the RSA Counterparty shall review such evidence and calculations and notify T&SCo in writing whether or not the RSA Counterparty is satisfied with the Interim Difference Payment provided by T&SCo, prior to the commencement of Charging Yeart.	
		5.5. If the RSA Counterparty notifies T&SCo in writing pursuant to section 5.4 that it is satisfied with the evidence of and calculations for the Interim Difference Payment submitted by T&SCo, then the pro-rata calculation shall be carried out by the RSA Counterparty pursuant to section 5.8.	
		5.6. If the RSA Counterparty notifies T&SCo in writing pursuant to section 5.4 that it is not satisfied with the evidence of and calculations for any Interim Difference Payment submitted by T&SCo, then section 5.7 shall apply.	
		5.7. T&SCo shall co-operate with the RSA Counterparty to provide any other information that the RSA Counterparty requires for assurance of T&SCo's calculations of the Interim Difference Payment pursuant to section 5.4 and the provisions of sections 5.4 to 5.7 shall continue to apply until the RSA Counterparty (acting reasonably) notifies T&SCo in writing that it is satisfied pursuant to section 5.5.	

No.	Item	Commercial Terms		
		Pro-rating and payment		
		5.8. Following notice of satisfaction by the RSA Counterparty pursuant to section 5.5, the RSA Counterparty shall:		
		 (a) calculate the monthly Interim Difference Payment by applying the Interim Difference Payment on a pro-rata basis in respect of the months in Charging Yeart ("Monthly IDP Amount"); and 		
		(b) within 5 Business Days, provide T&SCo with a provisional payment schedule specifying each Monthly IDP Amount in Charging Year _t .		
		5.9. T&SCo shall be entitled to payment of the Monthly IDP Amount by the RSA Counterparty (as adjusted by any Monthly QRP Adjustment), in accordance with section 7 (<i>Monthly billing and payment</i>).		
6.	Quarterly Reconciliation Payment	6.1. After the end of each quarter of Charging Yeart, a reconciliation of the Adjusted Forecast Market Revenue for that quarter against the Adjusted Market Revenue for that quarter shall be performed by T&SCo for the purposes of determining the "Quarterly Reconciliation Payment" .		
		Note: the Quarterly Reconciliation Payment applies on a quarterly rolling basis in each Charging Year, including in respect of Q4. The reconciliation of the Adjusted Forecast Market Revenue against the Adjusted Market Revenue of T&SCo can therefore be addressed solely through the Quarterly Reconciliation Payment on an ongoing basis without the need for the Annual Difference Payment mechanism, which has been deleted.		
		Calculation and evidence		
		6.2. The Quarterly Reconciliation Payment shall be calculated in accordance with this section 6 (<i>Quarterly Reconciliation Payment</i>) as the amount by which the Adjusted Forecast Market Revenue for the relevant quarter is greater than the Adjusted Market Revenue for the relevant quarter.		
		 6.3. T&SCo shall submit to the RSA Counterparty its evidence of and calculations for the Quarterly Reconciliation Payment, within 10 Business Days following the Invoice Due Date for the final month in Quartert. 		
		6.4. The evidence and calculations provided by T&SCo pursuant to section 6.3 shall include, as a minimum, the following in respect of each Quarterly Reconciliation Payment:		

No.	Item	Commercial Terms	
		(a)	the methodology and input data used by T&SCo to calculate the relevant Quarterly Reconciliation Payment;
		(b)	evidence of review, assurance or confirmation by the Regulator in relation to amounts required for the Quarterly Reconciliation Payment in accordance with section 10 (<i>Regulator review, assurance or confirmation</i>); and
		(c)	a certificate signed by a director or authorised signatory of T&SCo, certifying that the information provided to the RSA Counterparty pursuant to section 6.3 is true, complete and accurate.
		E	Assurance by the RSA Counterparty
) 	Following receipt by the RSA Counterparty of the evidence of and calculations for the Quarterly Reconciliation Payment in accordance with sections 6.3 and 6.4, the RSA Counterparty shall review such evidence and calculations and notify T&SCo in writing whether or not the RSA Counterparty is satisfied with the Quarterly Reconciliation Payment provided by T&SCo, within 20 Business Days of receipt of the evidence and calculations pursuant to sections 6.3 and 6.4.
		t c c	f the RSA Counterparty notifies T&SCo in writing pursuant to section 6.5 that it is satisfied with the evidence of and calculations for the Quarterly Reconciliation Payment submitted by T&SCo, then the pro-rata calculation shall be carried out by the RSA Counterparty pursuant to section 5.9.
		t	f the RSA Counterparty notifies T&SCo in writing pursuant to section 6.5 that it is not satisfied with the evidence of and calculations for any Quarterly Reconciliation Payment submitted by T&SCo, then section 6.8 shall apply.
		ן ר נ נ נ נ נ נ נ נ נ נ נ נ נ נ נ נ נ נ	T&SCo shall co-operate with the RSA Counterparty to provide any other information that the RSA Counterparty requires for assurance of T&SCo's calculations of the Quarterly Reconciliation Payment pursuant to section 6.5 and the provisions of sections 6.5 to 6.8 shall continue to apply until the RSA Counterparty (acting reasonably) notifies T&SCo in writing that it is satisfied pursuant to section 6.6.
		<u> </u>	Pro-rating and payment
			Following notice of satisfaction by the RSA Counterparty bursuant to section 6.6, the RSA Counterparty shall:

No.	Item	Commercial Terms		
		 (a) pro-rate the Quarterly Reconciliation Payment for Quarter_t over the three months of Quarter_{t+2} (the "Monthly QRP Adjustment"); and (b) within 5 Business Days, provide T&SCo with an updated provisional payment schedule specifying each Monthly QRP Adjustment to be applied in Quarter_{t+2}. 		
		6.10. The Monthly QRP Adjustment (positive or negative) shall be applied as an adjustment to the Monthly IDP Amount in accordance with section 7 (<i>Monthly billing and payment</i>).		
7.	Monthly billing and payment	7.1. Within 5 Business Days of the end of Month _t , the RSA Counterparty shall deliver a billing statement to T&SCo for Month _t (in accordance with the RSA Counterparty's billing statement requirements), setting out:		
		(a) the Monthly IDP Amount calculated in accordance with section 5 (<i>Interim Difference</i> Payment); and		
		(b) any positive or negative Monthly QRP Adjustment calculated in accordance with section 6 (<i>Quarterly Reconciliation Payment</i>).		
		7.2. The RSA Counterparty shall pay the net amount of the Monthly IDP Amount and any positive or negative Monthly QRP Adjustment within 5 Business Days of delivering the billing statement pursuant to section 7.1.		
		7.3. Where in Month _t the Monthly IDP Amount is less than any negative Monthly QRP Adjustment applied (i.e. the payment in respect of Month _t pursuant to section 7.2 is net negative), the difference between these amounts shall be addressed through reference to the annual iteration process administered by the Regulator pursuant to condition H16.1(b) of the Licence.		
8.	Revenue support prior to operations	8.1. Where there is delay to COD beyond the Scheduled COD, (excluding any extension arising from condition G2.7 of the Licence) and in the context of such delay T&SCo has provided to the RSA Counterparty a copy of a confirmation from the Regulator under condition F5.2(a) of the Licence that an Event of First User Delay has occurred (or has previously occurred and is continuing) revenue support will be provided by the RSA Counterparty to T&SCo:		
		 (a) in respect of each month (or part month as applicable) during the period from the later of (A) the date by which T&SCo provided all evidence required by the Regulator pursuant to condition F4 of the Licence; and (B) the date of Scheduled COD, until the earlier of (aa) twelve (12) months; and (bb) expiry of the period of delay arising from the Event of First User Delay, for amounts accrued 		

No.	Item	Comme	rcial Terms	
			based on	o in respect of Unavoidable Opex which shall be actual operating costs evidenced by T&SCo to Counterparty;
		(b)	during th month pe only while First Use	t of each month (or part month as applicable) he period from the expiry of the twelve (12) eriod referred to in paragraph (a) above, but st the period of delay arising from the Event of er Delay has not yet expired, for amounts by T&SCo in respect of:
			(i)	Unavoidable Opex evidenced by T&SCo to the RSA Counterparty;
			(ii)	allowed cost of debt and equity which shall be based on the assumed debt and equity elements of the WACC; and
			(iii)	allowed cost of depreciation which shall be based on the depreciation building block,
		(each a "Pre	e-operations Difference Payment").
			Note: this re financing pro	emains under review in the context of T&SCo's oposals.
		0	Construction	liaise with the first User pursuant to the Agreement and shall use reasonable to mitigate First User Delay.
		<u>(</u>	Calculation a	nd evidence
		۲ ج ن	provide evid any Pre-opei	ent of First User Delay occurs, T&SCo shall ence of entitlement to, and calculations for rations Difference Payment to the RSA v within 10 Business Days following the end of
		t	o section 8.	e and calculations provided by T&SCo pursuant 3 shall include, as a minimum, the following in e Pre-operations Difference Payment:
		(a)	condition	of a confirmation from the Regulator under F5.2(a) of the Licence that an Event of First ay has occurred (or has previously occurred and ning);
		(b)		nodology and input data used by T&SCo to the Pre-operations Difference Payment; and
		(c)	Regulato	of review, assurance or confirmation by the r in relation to amounts required for the Pre- is Difference Payment in accordance with

No.	Item	Commercial Terms
		section 10 (<i>Regulator review, assurance or confirmation</i>); and
		(d) a certificate signed by a director or authorised signatory of T&SCo, certifying that the information provided to the RSA Counterparty is true, complete and accurate.
		Assurance by the RSA Counterparty
		8.5. Following receipt by the RSA Counterparty of the evidence of and calculations for the Pre-operations Difference Payment pursuant to sections 8.3 and 8.4, the RSA Counterparty shall review such evidence and calculations and notify T&SCo in writing whether or not the RSA Counterparty is satisfied with the Pre-operations Difference Payment provided by T&SCo within 20 Business Days following receipt of the evidence and calculations pursuant to sections 8.3 and 8.4.
		8.6. If the RSA Counterparty notifies T&SCo in writing pursuant to section 8.5 that it is satisfied with the evidence of and calculations for the Pre-operations Difference Payment, then the RSA Counterparty shall pay any Pre-operations Difference Payment that it is obliged to pay to T&SCo in accordance with sections 8.9 and 8.10.
		8.7. If the RSA Counterparty notifies T&SCo in writing pursuant to section 8.5 that it is not satisfied with any of the evidence of and calculations for any Pre-operations Difference Payment, then section 8.8 shall apply.
		8.8. T&SCo shall co-operate with the RSA Counterparty to provide any other information that the RSA Counterparty requires for assurance of T&SCo's calculations of the Pre-operations Difference Payment pursuant to section 8.5 and the provisions of sections 8.5 to 8.8 shall continue to apply until the RSA Counterparty (acting reasonably) notifies T&SCo in writing that it is satisfied pursuant to section 8.6.
		<u>Payment</u>
		8.9. Within 20 Business Days following the notice of satisfaction by the RSA Counterparty pursuant to section 8.6, the RSA Counterparty shall deliver a billing statement to T&SCo (in accordance with the RSA Counterparty's billing statement requirements) in respect of any Preoperations Difference Payment for Montht.
		8.10. Within 5 Business Days following the delivery of the billing statement by the RSA Counterparty pursuant to section

No.	Item	Commercial Terms		
		8.9, the RSA Counterparty shall pay any Pre-operations Difference Payment to T&SCo.		
9.	RSA expiry reconciliation	9.1. T&SCo shall notify the RSA Counterparty immediately upon:		
		 (a) any revocation notice being issued by the Regulator under the Licence; or 		
		(b) any Discontinuation Notice being issued by the SoS under the Discontinuation Agreement.		
		Calculation and evidence		
		9.2. Within 10 Business Days following expiry of the RSA pursuant to section 3 (<i>Term</i>), T&SCo shall provide the RSA Counterparty with evidence and calculations, in accordance with this section 9 (<i>RSA expiry reconciliation</i>), of any pro-rata Difference Payments that are due and payable under the RSA to be accounted for up to and including the date of expiry of the RSA.		
		9.3. The evidence and calculations provided by T&SCo pursuant to section 9.2 shall include, as a minimum, the following in respect of each pro-rata Difference Payment:		
		(a) the methodology and input data used by T&SCo to calculate the relevant pro-rata Difference Payment;		
		(b) evidence of review, assurance or confirmation by the Regulator in relation to amounts required for each pro- rata Difference Payment in accordance with section 10 (<i>Regulator review, assurance or confirmation</i>); and		
		(c) a certificate signed by a director or authorised signatory of T&SCo, certifying that the information provided to the RSA Counterparty pursuant to section 9.2 is true, complete and accurate.		
		Assurance by the RSA Counterparty		
		9.4. Following receipt by the RSA Counterparty of the evidence of and calculations for any pro-rata Difference Payments in accordance with sections 9.2 and 9.3, the RSA Counterparty shall review such evidence and calculations and notify T&SCo in writing whether or not the RSA Counterparty is satisfied with the pro-rata Difference Payments provided by T&SCo within 20 Business Days following receipt of the evidence and calculations pursuant to sections 9.2 and 9.3.		
		9.5. If the RSA Counterparty notifies T&SCo in writing pursuant to section 9.4 that it is satisfied with the evidence of and		

No.	Item	Commercial Terms	
			calculations for any pro-rata Difference Payments, then the RSA Counterparty or T&SCo (as the case may be) shall pay any such pro-rata Difference Payments that it is obliged to pay in accordance with sections 9.8 to 9.10.
		9.6.	If the RSA Counterparty notifies T&SCo in writing pursuant to section 9.4 that it is not satisfied with any of the evidence of and calculations for any pro-rata Difference Payments pursuant to section 9.4, then section 9.7 shall apply.
		9.7.	T&SCo shall co-operate with the RSA Counterparty to provide any other information that the RSA Counterparty requires for assurance by the RSA Counterparty of T&SCo's calculations of any pro-rata Difference Payments pursuant to section 9.4 and the provisions of sections 9.4 to 9.7 shall continue to apply until the RSA Counterparty (acting reasonably) notifies T&SCo in writing that it is satisfied pursuant to section 9.5.
			Payment
		9.8.	Within 20 Business Days following the notice of satisfaction by the RSA Counterparty pursuant to section 9.5, the RSA Counterparty shall deliver a billing statement to T&SCO (in accordance with the RSA Counterparty's billing statement requirements) in respect of any applicable pro-rata Difference Payments due and payable.
		9.9.	Where any applicable pro-rata Difference Payment calculated under this section 9 (<i>RSA expiry reconciliation</i>) is:
		(a)	positive (i.e. there is a revenue collection shortfall by T&SCo), the RSA Counterparty shall pay the difference to T&SCo within 5 Business Days following the delivery of the billing statement by the RSA Counterparty pursuant to section 9.8; or
		(b)	negative (i.e. there is a revenue collection surplus by T&SCo), T&SCo shall pay the difference to the RSA Counterparty within 5 Business Days following the delivery of the billing statement by the RSA Counterparty pursuant to section 9.8 and in accordance with section 9.10.
		9.10.	Any payments that are required to be paid by T&SCo to the RSA Counterparty pursuant to this section 9 (<i>RSA</i> <i>expiry reconciliation</i>) shall be paid by T&SCo to a bank account nominated in writing by the RSA Counterparty to T&SCo from time to time.

No.	Item	Commercial Terms	
10.	Evidence of Regulator review, assurance or confirmation	10.1. Prior to any Difference Payment being paid to T&SCo by the RSA Counterparty, T&SCo shall be required to provide the RSA Counterparty with evidence of Regulator review and assurance or confirmation of such amounts (as applicable) against information that has otherwise been provided by T&SCo to the Regulator pursuant to the Licence or the CCS Network Code:	
		(a) the Allowed Revenue;	
		(b) the Adjusted Allowed Revenue;	
		(c) the Market Revenue;	
		(d) the Adjusted Market Revenue;	
		(e) the Adjusted Forecast Market Revenue; and	
		(f) the Forecast Market Revenue.	
11.	RSA Counterparty payment obligation and undertakings		
12.	RSA Counterparty right to request further information		
		 (a) the satisfaction of any know your customer checks (or equivalent) following the satisfaction of any initial conditions precedent pursuant to section 4 (<i>Conditions</i> precedent); 	
		(b) the discharge of the RSA Counterparty's statutory and/or regulatory obligations; or	
		(c) the discharge of any of the RSA Counterparty's obligations under the Revenue Support Agreement,	

No.	Item	Commercial Terms		
		and T&SCo shall promptly comply with any such information request made by the RSA Counterparty.		
13.	Representations, warranties and undertakings	13.1. Each of T&SCo and the RSA Counterparty shall make standard representations and warranties.		
14.	Limitation of liability	14.1. The RSA Counterparty shall exclude liability for any special, indirect or consequential loss including any such loss which constitutes loss of use, loss of goodwill, loss of profit or loss of revenue, provided that this shall not exclude liability for express payment obligations under the Revenue Support Agreement.		
15.	Payment disruption	15.1. Subject to section 15.2, the RSA Counterparty shall be relieved from liability to pay T&SCo, and deemed not to be in breach of the Revenue Support Agreement, to the extent that, and for so long as, the RSA Counterparty is suffering a material disruption to those payment systems or financial markets which are required to operate in order for payments or transfers of money to be made pursuant to the Revenue Support Agreement which the RSA Counterparty could not reasonably have overcome and which are not due to the RSA Counterparty's fault or negligence (a "Payment Disruption Event").		
		15.2. Where a Payment Disruption Event occurs, the RSA Counterparty shall:		
		 (a) give notice to T&SCo as soon as reasonably practicable of the nature and extent of the Payment Disruption Event; 		
		 (b) use reasonable endeavours to mitigate the effects of the Payment Disruption Event; 		
		(c) use reasonable endeavours to carry out and perform its obligations under the RSA in any way that is reasonably practicable for the RSA Counterparty to do so; and		
		(d) use reasonable endeavours to pay any sums to T&SCo that are due and payable under the RSA in accordance with its terms immediately upon cessation of the Payment Disruption Event.		
16.	Transfers	16.1. The Revenue Support Agreement shall benefit and bind the Parties, their permitted assignees and respective successors and any reference in the Revenue Support Agreement to any party shall be construed accordingly.		
		16.2. The Revenue Support Agreement will include restrictions on the assignment and/or transfer by T&SCo of any of its		

No.	Item	Commercial Terms
		rights or obligations under the Revenue Support Agreement (in whole or in part) without the prior written consent of the RSA Counterparty (such consent not to be unreasonably withheld or delayed), save that T&SCo will be permitted to create a security assignment in favour of its secured creditors (and the RSA Counterparty will assist in facilitating this, provided that all costs and expenses properly incurred by the RSA Counterparty in giving effect to such assignment are paid by T&SCo).
17.	Survivorship	 17.1. Following the expiry of the Revenue Support Agreement, each Party shall cease to have rights or obligations under the Revenue Support Agreement, save that in relation to (i) any legacy Difference Payments and/or (ii) pro-rata Difference Payments pursuant to section 9 (<i>RSA expiry reconciliation</i>), the provisions of sections 5 (<i>Interim Difference Payment</i>), 6 (<i>Quarterly Reconciliation Payment</i>), 7 (<i>Monthly billing and payment</i>), 8 (<i>Revenue support prior to operations</i>), 9 (<i>RSA expiry reconciliation</i>), 10 (<i>Regulator review, assurance or confirmation</i>), 11 (<i>RSA Counterparty payment obligation and undertakings</i>), 14 (<i>Limitation of liability</i>), 15 (<i>Payment disruption</i>), 16 (<i>Transfers</i>) and 19 (<i>Definitions</i>) will continue in full force and effect.
18.	Boilerplate	18.1. The Revenue Support Agreement will include "boilerplate provisions" (including, but not limited to, provisions in relation to interpretation, notices, confidentiality, intellectual property, dispute resolution, entire agreement, severability, set-off, third party rights, amendments, counterparts and governing law), which will be broadly based on the equivalent provisions set out in the GSP.
19.	Definitions	In this section B (<i>Revenue Support Agreement</i>), terms defined in the Licence heads of terms shall have the same meaning, and:
		"Adjusted Allowed Revenue" means the Allowed Revenue in the relevant Charging Year, other than by way of Flow Charges as calculated under the Charging Methodology in Section H of the CCS Network Code;
		"Adjusted Market Revenue" means Market Revenue earned and received by T&SCo in the relevant Charging Year, other than by way of Flow Charges;
		"Adjusted Forecast Market Revenue" means Market Revenue forecast to be earned and received by T&SCo in the relevant Charging Year, other than by way of Flow Charges;
		"Allowed Revenue" means the allowed revenue as it relates to the Approved T&S Network which T&SCo is entitled to charge in

No.	Item	Commercial Terms
		the relevant Charging Year determined by the Regulator pursuant to and in accordance with the Licence from time to time;
		"Approved T&S Network" has the meaning given to it in the Licence heads of terms, as at Licence Award;
		"CCS Network Code" means the document designated as such and required to be maintained by the Licensee under the Licence;
		"Charges" has the meaning given in the CCS Network Code, as such Charges relate to the Approved T&S Network;
		"Charging Year" means a calendar year commencing on 1 April of one calendar year and ending on 31 March of the immediately following calendar year, save that:
		 (a) where COD falls part way through a Charging Year, the relevant number of calendar months shall be the number of calendar months (or days) during the Charging Year which fall within the period from COD to 31 March in the relevant year; and
		(b) where the Revenue Support Agreement expires part way through a Charging Year, the relevant number of calendar months shall be the number of calendar months (or days) during the Charging Year which fall within the period from 1 April to the expiry date in the relevant year;
		"Charging Yeart" means the Charging Year to which the calculation relates;
		"COD" has the meaning given to it in the Licence heads of terms;
		"COD Readiness Activities" has the meaning given to it in the Licence heads of terms;
		"Construction Agreement" has the meaning given to it in the CCS Network Code;
		"Difference Payment" means any of the following:
		(a) Interim Difference Payment;
		(b) Quarterly Reconciliation Payment;
		(c) Pre-operations Difference Payment; and
		(d) pro-rata payment pursuant to section 9 (<i>RSA expiry reconciliation</i>);

No.	Item	Commercial Terms
		"Discontinuation" has the meaning given to it in the Discontinuation Agreement heads of terms;
		"Discontinuation Notice" has the meaning given to it in the Discontinuation Agreement heads of terms;
		"Event of First User Delay" has the meaning given to it in the Licence heads of terms;
		"First User Delay" means any delay to COD which is caused by, or shall be caused by, the first User;
		"Flow Charges" has the meaning given to it in the CCS Network Code;
		"Forecast Market Revenue" means Market Revenue forecast to be earned and received by T&SCo over a Charging Year;
		"Interim Difference Payment" means the amount determined pursuant to section 5 (Interim Difference Payment);
		"Invoice Due Date" has the meaning given to that term in paragraph 10.2 of Section H of the CCS Network Code;
		"Licence" means the economic licence to be granted under HM Government's CCUS programme;
		"Market Revenue" means any revenue that relates to the Approved T&S Network, earned and received by T&SCo in respect of the relevant Charging Year from:
		(a) Charges; and
		(b) any other sources that relate to the Approved T&S Network in accordance with the terms of the Licence,
		and shall not include any revenue that relates to the Approved T&S Network and that is due and payable to T&SCo in respect of the relevant Charging Year but which has not been received by T&SCo
		Note: the Market Revenue will include all revenue of T&SCo, including non-regulated revenue, for example.
		Note: provisions reflecting the following principles are to be incorporated:
		(i) T&SCo to be under an obligation to procure asset damage insurance and associated BI insurance to the extent available;
		<i>(ii)</i> Code to provide for suspension of T&S charges to the extent covered by the BI insurance procured pursuant to the above obligation; and

No.	Item	Commercial Terms
		<i>(iii)</i> RSA to treat amounts recoverable under the BI insurance procured pursuant to the above obligation as Market Revenues.
		Detailed drafting to be developed, including scoping of procurement obligation and definition of amounts recoverable.
		"Month $_t$ " means the month to which the relevant Difference Payment calculations relate;
		"Monthly IDP Amount" has the meaning given to in section 5.8;
		"Monthly QRP Adjustment" has the meaning given to it in section 6.9;
		"Pre-operations Difference Payment" means the amount determined pursuant to section 8 (<i>Revenue support prior to operations</i>);
		$^{\prime\prime}Quarter_{t}^{\prime\prime}$ means the quarter to which the relevant Difference Payment calculations relate;
		"Quarter $_{t+2}$ " means the second quarter following Quarter $_t$;
		"Quarterly Reconciliation Payment" has the meaning given to it in section 6 (Quarterly Reconciliation Payment);
		"Scheduled Commercial Operations Date" has the meaning given to it in the Licence; and
		"Unavoidable Opex" means Opex which cannot reasonably be mitigated by T&SCo, taking into account the need to maintain an economic, efficient and effective approach to Opex costs not only following an Event of First User Delay but also in reaching COD, including Opex that needs to be incurred in order to maintain the T&S Assets in a state and condition that meets the requirement for COD Readiness whilst an Event of First User Delay is continuing.

This publication is available from: www.gov.uk/desnz

If you need a version of this document in a more accessible format, please email: <u>alt.formats@energysecurity.gov.uk</u>.

Please tell us what format you need. It will help us if you say what assistive technology you use.